

# Brownian continuum random tree conditioned to be large

Romain Abraham, Jean-Franç Ois Delmas, Hui He

## ▶ To cite this version:

Romain Abraham, Jean-Franç Ois Delmas, Hui He. Brownian continuum random tree conditioned to be large. 2022. hal-03581326v1

## HAL Id: hal-03581326 https://hal.science/hal-03581326v1

Preprint submitted on 19 Feb 2022 (v1), last revised 2 Jul 2024 (v2)

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

### BROWNIAN CONTINUUM RANDOM TREE CONDITIONED TO BE LARGE

ROMAIN ABRAHAM, JEAN-FRANÇOIS DELMAS, AND HUI HE

ABSTRACT. We consider a Feller diffusion  $(Z_s, s \ge 0)$  (with diffusion coefficient  $\sqrt{2\beta}$  and drift  $\theta \in \mathbb{R}$ ) that we condition on  $\{Z_t = a_t\}$ , where  $a_t$  is a deterministic function, and we study the limit in distribution of the conditioned process and of its genealogical tree as  $t \to +\infty$ . When  $a_t$  does not increase too rapidly, we recover the standard size-biased process (and the associated genealogical tree given by the Kesten's tree). When  $a_t$  behaves as  $\alpha\beta^2t^2$  when  $\theta = 0$  or as  $\alpha e^{2\beta|\theta|t}$  when  $\theta \neq 0$ , we obtain a new process whose distribution is described by a Girsanov transformation and equivalently by a SDE with a Poissonian immigration. Its associated genealogical tree is described by an infinite discrete skeleton (which does not satisfy the branching property) decorated with Brownian continuum random trees given by a Poisson point measure.

As a by-product of this study, we introduce several sets of trees endowed with a Gromovtype distance which are of independent interest and which allow here to define in a formal and measurable way the decoration of a backbone with a family of continuum random trees.

#### 1. INTRODUCTION

In [1], for the geometric reproduction law, and in [5], for general super-critical reproduction laws with finite mean and some special sub-critical reproduction laws, the authors consider the limit of a Galton-Watson (GW) process  $(Z_n, n \in \mathbb{N})$  started at  $Z_0 = 1$  conditionally on  $Z_n = a_n$ as n goes to infinity, provided the event  $\{Z_n = a_n\}$  has positive probability. They also consider more generally the local limit of the GW tree, which in particular allows to study condensation phenomenon (on this latter subject, see [26, 25, 4]). According to the different growth rate of  $a_n$  as n goes to infinity, they observe different regimes for the limiting random tree: if  $a_n = 0$ for n large, the limiting tree corresponds to the GW tree conditioned on the extinction event; if  $a_n$  is strictly positive but grows slowly (including the case  $a_n$  bounded), then the limit is the so called Kesten tree, which consists in an infinite spine decorated with independent GW trees with the initial reproduction law; if  $a_n$  grows at a moderate speed (given in the super-critical case of finite variance by  $a_n \sim \alpha m^n$  with  $\alpha > 0$  and m the mean of the reproduction law), then the limit is a skeleton given by an immigration process decorated again with independent GW trees with the initial reproduction law; if  $a_n$  grows faster than  $m^n$  (that is  $\lim_{n\to\infty} m^{-n}a_n = \infty$ ) then results are known only for the geometric reproduction law (the limit exhibits a condensation at the root, that is, the root has an infinite number of children, and then those children generate independent trees) and for bounded reproduction laws (the limit is the regular b-ary tree, with b the possible maximum number of children).

This work is a first step to extend those results to random real trees called Lévy trees introduced by Duquesne and Le Gall in [18, 19] which are scaling limits of (sub)critical GW trees and can be seen as genealogical trees for (sub)critical continuous state branching processes (CSBP); see also [3, 20] for the extension of this latter representation to the super-critical case. We shall only

*Date*: February 19, 2022.

<sup>2010</sup> Mathematics Subject Classification. 60J80, 54E50.

Key words and phrases. Feller diffusion, continuum random trees, conditioning, local limits.

consider Feller diffusions, which correspond to CSBPs with quadratic branching mechanism and whose genealogy can be described using the Brownian continuum random tree introduced by Aldous [9]. Our results belong also to the family of works dedicated to the description of limits of conditioned random real trees, in this direction, see [29, 28, 17, 2].

We now present informally our result. We denote by  $\mathcal{T}$  the rooted Brownian continuum tree which describes the genealogy of a population started at time 0, and by  $Z_t$  the "size" of the population at time  $t \ge 0$ . The process  $Z = (Z_t, t \ge 0)$  is a quadratic CSBP solution to the following stochastic differential equation (SDE):

$$\mathrm{d}Z_t = \sqrt{2\beta Z_t} \,\mathrm{d}B_t - 2\beta\theta Z_t \mathrm{d}t, \quad \text{for } t \ge 0,$$

where  $(B_t, t \ge 0)$  is some standard Brownian motion,  $\beta > 0$  and  $\theta \in \mathbb{R}$ . The CSBP is associated with the branching mechanism

$$\psi_{\theta}(\lambda) = \beta \lambda^2 + 2\beta \theta \lambda.$$

The CSBP is sub-critical (resp. super-critical) if  $\theta > 0$  (resp.  $\theta < 0$ ). We denote by  $\mathbb{N}^{\theta}$  the  $\sigma$ -finite excursion measure associate with the Brownian tree  $\mathcal{T}$  and the process Z, stressing the dependence in the parameter  $\theta$ , whereas the time scaling parameter  $\beta$  is fixed. Intuitively, under  $\mathbb{N}^{\theta}$ , the population starts from an infinitesimal individual which is the root of  $\mathcal{T}$ , and the non-zero process Z starts from an infinitesimal mass. In the introduction, we simply denote by  $\mathbf{t}_t$  the real tree  $\mathbf{t}$  truncated at level t. We denote by  $\mathcal{G}_t$  the  $\sigma$ -field generated by  $\mathcal{T}_t$  for  $t \geq 0$ ; in particular the process Z is adapted to the filtration ( $\mathcal{G}_t, t \geq 0$ ). Let  $a = (a_t, t \geq 0)$  be a non-negative function. We shall consider for s > 0, the possible limiting distribution of  $\mathcal{T}_s$ , conditionally on  $\{Z_t = a_t\}$  as t goes to infinity. Let F be any bounded continuous function defined on the set of trees (see Section 5 for the topology on the set of rooted locally compact tree).

• Extinction case:  $a_t = 0$  for t large. We have:

$$\lim_{t \to \infty} \mathbb{N}^{\theta} \big[ F(\mathcal{T}_s) \mid Z_t = a_t \big] = \mathbb{N}^{|\theta|} \big[ F(\mathcal{T}_s) \big].$$

We recall that the measure of the non extinction event is given by  $\mathbb{N}^{\theta}[Z_t > 0$  for all  $t > 0] = 2 \max(0, -\theta)$ . So the conditioning on the extinction event does not change anything in the (sub)critical case. In the super-critical case, this result is a direct consequence of the Girsanov transformation used in [3] to define the super-critical Lévy tree, see also (78). This result holds in general for any compact Lévy trees (but for the above value of the measure of the non extinction event).

• Low regime: a is positive and  $a_t = o(t^2)$  if  $\theta = 0$  or  $a_t = o(e^{2\beta|\theta|t})$  if  $\theta \neq 0$ . We have (see Theorem 6.13):

$$\lim_{t \to \infty} \mathbb{N}^{\theta} \left[ F(\mathcal{T}_s) \mid Z_t = a_t \right] = \mathbb{E} \left[ F\left(\mathcal{T}_s^{0, |\theta|}\right) \right],$$

where for  $q \geq 0$ ,  $\mathcal{T}^{0,q}$  is a Kesten tree with parameter  $(\beta, q)$ , which is informally obtained by grafting the trees  $(\mathcal{T}_i, i \in I)$  respectively at levels  $(h_i, i \in I)$  on an infinite spine, where the point measure  $\sum_{i \in I} \delta_{h_i, \mathcal{T}_i}$  is a Poisson point measure with intensity rate  $2\beta \mathbf{1}_{\{h>0\}} dh \mathbb{N}^q [d\mathcal{T}]$ . See Section 6.2 for a more formal definition of the Kesten tree. When conditioning instead by  $\{Z_t > 0\}$ , this result appears already in [2] for general compact Lévy trees.

• Moderate regime:  $a_t \sim \alpha \beta^2 t^2$  if  $\theta = 0$  or  $a_t \sim \alpha (2\theta)^{-2} e^{2\beta |\theta| t}$  if  $\theta \neq 0$ , where  $\alpha \in (0, +\infty)$ . We have (see Theorem 6.12):

$$\lim_{t \to \infty} \mathbb{N}^{\theta} \left[ F(\mathcal{T}_s) \mid Z_t = a_t \right] = \mathbb{E} \left[ F\left(\mathcal{T}_s^{\alpha, |\theta|}\right) \right],$$

 $\mathbf{2}$ 

where for  $q \geq 0$ ,  $\mathcal{T}^{\alpha,q}$  is a random tree informally obtained by grafting the tree  $\mathcal{T}_i$  at vertex  $x_i$  of a backbone tree  $\mathfrak{T}^{\alpha,q}$ , with the point measure  $\sum_{i\in I} \delta_{x_i,\mathcal{T}_i}$  being, conditionally given  $\mathfrak{T}^{\alpha,q}$ , a Poisson point measure with intensity rate  $2\beta \mathscr{L}(dx) \mathbb{N}^q[d\mathcal{T}]$ , and  $\mathscr{L}(dx)$ being the length measure on  $\mathfrak{T}^{\alpha,q}$ . The backbone tree  $\mathfrak{T}^{\alpha,q}$  represents in some sense the genealogy associated to an inhomogeneous Poissonian immigration with finite rate, see Section 6.4 for a more precise description. Let us stress that the backbone tree does not enjoy the branching property, as already observed by [1, 5] in a discrete setting.

• High regime:  $\lim_{t\to\infty} t^{-2}a_t = +\infty$  if  $\theta = 0$  or  $\lim_{t\to\infty} e^{-2\beta|\theta|t}a_t = +\infty$  if  $\theta \neq 0$ . The description of the possible limit in this regime is still an open question. As in the discrete setting studied in [1], one expects to have a condensation phenomenon at the root. However, such limiting tree would not be locally compact (at least at the root), and the study of such trees would require a non trivial extension of the current topology developed for locally compact trees.

The present result on the convergence in distribution of the conditioned Brownian tree in the low and moderate regime is given in Section 6.4. It relies on:

- (i) The extremal time-space harmonic functions for the process Z given by Overbeck [31]. Those harmonic functions appear in the limit when conditioning the process Z by the events  $\{Z_t = a_t\}$  in the moderate regime, see Section 3.
- (ii) A nice representation of the Doob *h*-transform using those extremal harmonic functions, based on Feller diffusion with an increasing immigration rate given by an inhomogeneous Poisson process. Roughly speaking, for  $\theta \ge 0$  and  $\alpha > 0$ , the local time process  $Z^{\alpha} = (Z_t^{\alpha}, t \ge 0)$  of the tree  $\mathcal{T}^{\alpha, \theta}$  is a Markov process which satisfies the following SDE:

$$dZ_t^{\alpha} = \sqrt{2\beta Z_t^{\alpha}} dB_t - 2\beta \theta Z_t^{\alpha} dt + 2\beta \left(S_t^{\alpha,\theta} + 1\right) dt \quad \text{for } t \ge 0, \text{ and } Z_0^{\alpha} = 0,$$

where  $B = (B_t, t \ge 0)$  is a standard Brownian motion and  $S^{\alpha,\theta} = (S_t^{\alpha,\theta}, t \ge 0)$  is a Poisson process with intensity  $\alpha\beta e^{2\beta\theta t} dt$  independent of B. See the main result of Section 4 in Proposition 4.1 for more details. Its proof, given in Section 4.2, uses a result from Rogers and Pitman [33] for a transformation of a Markov process (which in our case amount to not be able to distinguish the immigration  $S^{\alpha,\theta}$ ) to still be a Markov process.

- (iii) An extension of a result from Duquesne and Le Gall [19, Theorem 4.5] on the description of the Brownian tree  $\mathcal{T}$  as a decorated sub-tree spanned by *n* vertices chosen randomly at level *t*, see Theorem 6.8 in Section 6.3. We believe that this theorem is of independent interest and complete the description of [20] where one chooses theses vertices at random without condition on their level.
- (iv) A transcription of the genealogy for immigration process from point (ii) above into a backbone tree, see Section 6.1 and the combinatorial Lemma 6.1 on the distribution of a tree built sequentially by grafting uniformly branches at random levels.

Let us mention here that there have been several works on skeletal/backbone decompositions for (spatial) branching processes and their corresponding genealogical trees, for example see [2, 10, 12, 20, 23, 24, 27] and the references therein. In particular, in [23], coupled systems of SDEs were established to represent the skeletal decompositions for continuous-state branching processes (conditioned on survival), where the skeletons are determined by continuous-time Galton-Watson processes. And we refer to [20], for reconstruction of a Lévy tree from a backbone tree, which could be formed by leaves taking at random in a Poissonnian way from the Lévy tree according to the so-called mass measure; see Remark 5.4 there and [18]. For representations of branching processes (with immigrations) via SDEs, we also refer to [15] and references therein. To complete the outline of the paper, let us mention that Section 2 is devoted to some notations and elementary facts for quadratic CSBP and Section 6.2 to known results on the Brownian tree. Eventually the large Section 5 is devoted to various topological results on the spaces of trees. The main objective of this section is to define the grafting, the splitting, as well as the decorating of trees in a measurable way on the set of equivalent classes of locally compact rooted real trees. An index of all the (numerous) relevant notations of this section is provided at the end of the document.

#### 2. General quadratic CSBP

2.1. Notations. We set  $\mathbb{R}_+ = [0, +\infty)$ ,  $\mathbb{R}_+^* = (0, +\infty)$ ,  $\mathbb{N} = \{0, 1, 2, \cdots\}$  and  $\mathbb{N}^* = \{1, 2, \cdots\}$ . For  $x \in \mathbb{R}$ , we set  $x_+ = \max(0, x)$  and  $x_- = \max(0, -x)$ , so that  $x = x_+ - x_-$ . We write  $\delta_x$ 

for the Dirac mass at x.

Let (E, d) be a metric space. We denote by  $\mathcal{M}_+(E)$  the space of non-negative measures on E endowed with the vague topology. For  $\mu \in \mathcal{M}_+(E)$  and A a Borel subset of E, we denote by  $\mu_{|A}(dx)$  the measure  $\mathbf{1}_A(x)\mu(dx)$ . We write  $\mu(f) = (f, \mu) = \int f \, d\mu = \langle f, \mu \rangle$  for the integral of the measurable real-valued function f with respect to the measure  $\mu$ , whenever it is meaningful.

We say that a function from a measurable space to a measurable space is bi-measurable if it is measurable and the image of any measurable set is a measurable set (when the function is one-to-one this is equivalent to the function and its inverse being measurable).

2.2. Quadratic CSBP. Let  $\beta > 0$  be fixed. Let  $\theta \in \mathbb{R}$ . We consider the quadratic branching mechanism  $\psi_{\theta}$  given for  $\lambda \in \mathbb{R}$  by:

(1) 
$$\psi_{\theta}(\lambda) = \beta \lambda^2 + 2\beta \theta \lambda.$$

The corresponding CSBP  $Z = (Z_t, t \ge 0)$  is the unique strong solution to the following stochastic differential equation (SDE):

(2) 
$$dZ_t = \sqrt{2\beta Z_t} dB_t - 2\beta \theta Z_t dt \quad \text{for } t \ge 0,$$

where  $B = (B_t, t \ge 0)$  is a standard Brownian motion and  $Z_0 = x \ge 0$ . For  $t \ge 0$ , let  $\mathcal{F}_t$  be the  $\sigma$ -field generated by  $(Z_s, t \in [0, t])$ . We write  $\mathbb{P}_x^{\theta}$  to stress the value of the parameter  $\theta$ , and the initial value of the process  $Z, Z_0 = x$ . We denote by  $\mathbb{N}^{\theta}$  the canonical measure of the process Z, normalized in such a way that for  $\lambda \ge 0$ :

$$\mathbb{N}^{\theta} \left[ 1 - \mathrm{e}^{-\lambda \sigma} \right] = \psi_{\theta}^{-1}(\lambda),$$

where  $\sigma = \int_0^\infty Z_t \, dt$  is the total size of the population under the canonical measure  $\mathbb{N}^{\theta}$  and  $\psi_{\theta}^{-1}(\lambda)$  is the only root t to  $\psi_{\theta}(t) = \lambda$  such that  $t \ge 2\theta_-$ . In particular, the process  $(Z_t, t \ge 0)$  under  $\mathbb{P}^{\theta}_x$  is distributed as the process  $\left(\sum_{i \in I} Z_t^{(i)}, t \ge 0\right)$  where  $\sum_{i \in I} \delta_{Z^{(i)}}$  is a Poisson point measure with intensity  $x\mathbb{N}^{\theta}(dZ)$ . We refer to [18] for  $\theta \ge 0$  (critical and sub-critical case) and to [16, 3, 20] for  $\theta < 0$  (super-critical case) for a detailed presentation of the CSBP process Z and the corresponding continuum Brownian random tree  $\mathcal{T}$ .

In order to recall the Laplace transform of  $Z_t$ , we introduce the following positive functions  $c^{\theta}$  and  $\tilde{c}^{\theta}$  defined for  $t \in (0, +\infty)$  by:

(3) 
$$c_t^{\theta} = \frac{2\theta}{\mathrm{e}^{2\beta\theta t} - 1}$$
 and  $\tilde{c}_t^{\theta} = \frac{2\theta}{1 - \mathrm{e}^{-2\beta\theta t}},$ 

4

with the convention  $c_t^0 = \tilde{c}_t^0 = 1/\beta t$ . The functions  $c^{\theta}$  and  $\tilde{c}^{\theta}$  are decreasing with  $\lim_{t\to 0+} c_t^{\theta} = \lim_{t\to 0+} \tilde{c}_t^{\theta} = +\infty$ ,  $\lim_{t\to +\infty} c_t^{\theta} = 2\theta_-$  and  $\lim_{t\to +\infty} \tilde{c}_t^{\theta} = 2\theta_+$ . We also have for t > 0:

(4) 
$$\tilde{c}_t^{\theta} = c_t^{-\theta} = c_t^{\theta} + 2\theta.$$

Remark 2.1 (Scaling property of Z). In this remark, we write  $Z^{[\beta,\theta]}$  for Z under  $\mathbb{N}^{\theta}$  or  $\mathbb{P}^{\theta}_{x}$  in order to stress the dependence in  $\beta > 0$  and  $\theta \in \mathbb{R}$ . Let  $Y = (Y_s, s \ge 0)$  be a Feller diffusion, *i.e.*, a CSBP with parameters  $\beta = 1$  and  $\theta = 0$ ; given as the unique strong solution to the SDE:

(5) 
$$\mathrm{d}Y_s = \sqrt{2Y_s} \,\mathrm{d}B_s, \quad \text{for } s \ge 0$$

Let  $\beta > 0$  and  $\theta \in \mathbb{R}$ , and define the process  $Y' = (Y'_s, s \in [0, 1/(2\theta_-)))$  by:

(6) 
$$Y'_{s} = e^{2\beta\theta t} Z_{t}^{[\beta,\theta]} \quad \text{with} \quad s = \frac{1}{c_{t}^{\theta}} \cdot$$

Then, we get that Y' is distributed as  $(Y_s, s \in [0, 1/(2\theta_-)))$  under  $\mathbb{N}^{\theta}$  or  $\mathbb{P}^{\theta}_x$  provided  $Y'_0 = Y_0 = x \ge 0$ .

We define for t > 0 and  $\lambda > -\tilde{c}_t^{\theta}$ :

(7) 
$$u^{\theta}(\lambda,t) = \frac{\lambda c_t^{\theta}}{\tilde{c}_t^{\theta} + \lambda} = c_t^{\theta} - \frac{c_t^{\theta} \tilde{c}_t^{\theta}}{\tilde{c}_t^{\theta} + \lambda},$$

and set  $u^{\theta}(\lambda, 0) = \lambda$  for t = 0. This gives that for t > 0 and  $\lambda > -\tilde{c}_t^{\theta}$ :

$$u^{\theta}(\lambda, t) = \begin{cases} \frac{2\theta\lambda}{(2\theta + \lambda) e^{2\beta\theta t} - \lambda}, & \text{if } \theta \neq 0, \\ \lambda/(1 + \lambda\beta t), & \text{if } \theta = 0. \end{cases}$$

For r > 0 and  $t \ge 0$ , we have that:

$$u^{\theta}(c_r^{\theta},t) = c_{t+r}^{\theta}$$

We recall from the above mentioned references ([18, 16, 3, 20]) for  $\lambda \ge 0$  and by analytic prolongation for  $\lambda < 0$ , that for t > 0 and  $x \ge 0$ :

(8) 
$$\mathbb{N}^{\theta} \left[ 1 - e^{-\lambda Z_t} \right] = u^{\theta}(\lambda, t) \text{ and } \mathbb{E}_x^{\theta} \left[ e^{-\lambda Z_t} \right] = e^{-xu^{\theta}(\lambda, t)} \text{ for all } \lambda > -\tilde{c}_t^{\theta}.$$

We denote by  $\zeta = \inf\{t > 0; Z_t = 0\}$  the lifetime of the process Z. We recall that for all t > 0:

$$\mathbb{N}^{\theta}[\zeta > t] = \lim_{\lambda \to \infty} u^{\theta}(\lambda, t) = c^{\theta}(t).$$

By considering the series in  $\lambda$  in (7) and (8), we deduce that for all t > 0 and  $n \in \mathbb{N}^*$ :

(9) 
$$\mathbb{N}^{\theta} \left[ \left( \tilde{c}_t^{\theta} Z_t \right)^n \right] = n! c_t^{\theta}.$$

We now give a martingale related to the CSBP Z. Recall that  $\tilde{c}_t^{\theta}$  is decreasing in t, and thus  $-\tilde{c}_{t+r}^{\theta} > -\tilde{c}_r^{\theta}$ . The next lemma is an easy consequence of (8) and the following elementary equality:

$$u(-\tilde{c}^{\theta}_{t+r},t) = -\tilde{c}^{\theta}_r$$
 for all  $t \ge 0$  and  $r > 0$ .

**Lemma 2.2.** Let  $\theta \in \mathbb{R}$ ,  $x \in \mathbb{R}_+$ , r > 0 and the quadratic CSBP  $(Z_t, t \ge 0)$  solution of (2). The process  $\left(e^{\tilde{c}_{t+r}^{\theta}Z_t}, t \in I\right)$  is a martingale under  $\mathbb{N}^{\theta}$  with  $I = (0, +\infty)$  and under  $\mathbb{P}_x^{\theta}$  with  $I = \mathbb{R}_+$  with respect to the filtration  $(\mathcal{F}_t, t \ge 0)$ .

We end this section with the computation of the densities of the entrance law  $q_t^{\theta}(dx)$  and the transition kernel  $q_t^{\theta}(x, dy)$  of the CSBP Z under its excursion measure, where for t, s > 0, x > 0 and  $y \ge 0$ :

$$q_t^{\theta}(\mathrm{d}x) = \mathrm{d}\mathbb{N}^{\theta}[Z_t = x, \zeta > t]$$
 and  $q_{t,s}^{\theta}(x, \mathrm{d}y) = \mathrm{d}\mathbb{N}^{\theta}[Z_{t+s} = y | Z_s = x].$ 

**Lemma 2.3** (Entrance law and transition densities of Z). Let  $\theta \in \mathbb{R}$ . Let t, s > 0, x > 0 and  $y \ge 0$ . We have:

$$q_t^{\theta}(\mathrm{d}x) = q_t^{\theta}(x) \,\mathrm{d}x \quad and \quad q_{t,s}^{\theta}(x,\mathrm{d}y) = \mathrm{e}^{-xc_t^{\theta}} \,\delta_0(\mathrm{d}y) + q_t^{\theta}(x,y) \,\mathrm{d}y,$$

where:

(10) 
$$q_t^{\theta}(x) = c_t^{\theta} \tilde{c}_t^{\theta} e^{-\tilde{c}_t^{\theta} x}$$

(11) 
$$q_t^{\theta}(x,y) = x c_t^{\theta} \tilde{c}_t^{\theta} e^{-(x+y)c_t^{\theta} - 2\theta y} \sum_{k \in \mathbb{N}} \frac{(xyc_t^{\theta} \tilde{c}_t^{\theta})^k}{k!(k+1)!}$$

*Proof.* We omit the parameter  $\theta$  in the proofs. On one hand, from the definition of  $q_t(dx)$ , we get that for  $\lambda \ge 0$ :

$$\int_{0}^{+\infty} e^{-\lambda x} q_t(\mathrm{d}x) = \mathbb{N}\left[e^{-\lambda Z_t} \mathbf{1}_{\{\zeta > t\}}\right] = -\mathbb{N}\left[1 - e^{-\lambda Z_t}\right] + \mathbb{N}\left[\zeta > t\right] = c(t) - u(\lambda, t).$$

On the other hand, using (7), we get:

$$\int_0^\infty c_t \tilde{c}_t \,\mathrm{e}^{-(\tilde{c}_t+\lambda)x} \,\mathrm{d}x = c(t) - u(\lambda, t).$$

Then use that finite positive measures on  $\mathbb{R}_+$  are characterized by their Laplace transform to obtain that  $q_t(dx) = q_t(x) dx$  with  $q_t$  given by (10).

From the definition of  $q_t(x, dy)$ , we get that for  $\lambda \ge 0$ :

$$\int_0^{+\infty} e^{-\lambda y} q_t(x, \mathrm{d}y) = \mathbb{N} \left[ e^{-\lambda Z_{t+s}} \mid Z_s = x \right] = e^{-xu(\lambda, t)} = e^{-\frac{a}{b} + \frac{a}{b+\lambda}},$$

where, thanks to (7),  $a = xc_t \tilde{c}_t$  and  $b = \tilde{c}_t$ . Notice that:

$$e^{\frac{a}{b+\lambda}} = 1 + \sum_{k \in \mathbb{N}} \frac{1}{(k+1)!} \left(\frac{a}{b+\lambda}\right)^{k+1} = 1 + a \sum_{k \in \mathbb{N}} \int_0^{+\infty} \frac{(ay)^k}{k!(k+1)!} e^{-by-\lambda y} \, \mathrm{d}y.$$

Using (3), we deduce that  $q_t(x, dy) = e^{-xc_t} \delta_0(dy) + q_t(x, y) dy$ , with  $q_t(x, y)$  given by (11).  $\Box$ 

#### 3. Local limits for the process Z

3.1. Some martingales. We present in this section two martingales which will naturally appear in the local limits for the Brownian continuum random tree (CRT). Recall  $\theta \in \mathbb{R}$ . Let  $\alpha \in \mathbb{R}$ . We define the non-negative process  $M^{\alpha,\theta} = (M_t^{\alpha,\theta}, t > 0)$  by:

(12) 
$$M_t^{\alpha,\theta} = Z_t \operatorname{e}^{-\alpha/c_t^{\theta}} \sum_{i \in \mathbb{N}} \frac{(\alpha Z_t)^i}{i!(i+1)!} \operatorname{e}^{(i+1)2\beta\theta t}$$

Remark 3.1. Using the Bessel function  $I_1(x) = \sum_{i \in \mathbb{N}} (x/2)^{2i+1} / i! (i+1)!$ , we can rewrite  $M_t^{\alpha, \theta}$  as:

$$M_t^{\alpha,\theta} = \sqrt{\frac{\mathrm{e}^{2\beta\theta t} Z_t}{\alpha}} \,\mathrm{e}^{-\alpha/c_t^{\theta}} I_1\left(2\sqrt{\alpha \,\mathrm{e}^{2\beta\theta t} Z_t}\right).$$

Using the process Y' defined in (6), which is a time change of the process Z (and which is distributed as a CSBP with parameter  $\beta = 1$  and  $\theta = 0$ ), we have for t > 0:

(13) 
$$M_t^{\alpha,\theta} = \sqrt{\frac{Y'_s}{\alpha}} e^{-\alpha s} I_1\left(2\sqrt{\alpha Y'_s}\right) = Y'_s e^{-\alpha s} \sum_{i \in \mathbb{N}} \frac{(\alpha Y'_s)^i}{i!(i+1)!} \cdot$$

Using Theorem 3 of [31] and Remark 3.1, we immediately get the following result.

**Proposition 3.2.** Let  $\alpha, \theta \in \mathbb{R}$ ,  $x \in \mathbb{R}_+$ . The process  $\left(M_t^{\alpha, \theta}, t \in I\right)$  is a martingale under  $\mathbb{N}^{\theta}$  with  $I = (0, +\infty)$  and under  $\mathbb{P}_x^{\theta}$  with  $I = \mathbb{R}_+$ .

Moreover, for all t > 0, we have, using (9) and  $\tilde{c}_t^{\alpha}/c_t^{\theta} = e^{2\beta\theta t}$ , that:

$$\mathbb{N}^{\theta}\left[M_t^{\alpha,\theta}\right] = 1.$$

Since  $u^{\theta}(-2\theta, t) = -2\theta$ , see (4) and (7), and  $-2\theta > -\tilde{c}_t^{\theta}$ , we deduce that  $(e^{2\theta Z_t}, t > 0)$  is a martingale under  $\mathbb{N}^{\theta}$ . According to [3, Section 4], we have that for  $\theta \in \mathbb{R}$ , t > 0:

(14) 
$$\mathbb{N}^{-\theta}[\mathrm{d}Z]_{|\mathcal{F}_t} = \mathrm{e}^{2\theta Z_t} \, \mathbb{N}^{\theta}[\mathrm{d}Z]_{|\mathcal{F}_t}.$$

For  $\alpha, \theta \in \mathbb{R}$ , we set  $\tilde{M}^{\alpha, \theta} = (\tilde{M}^{\alpha, \theta}_t, t > 0)$  with:

(15) 
$$\tilde{M}_t^{\alpha,\theta} = e^{2\theta Z_t} M_t^{\alpha,-\theta}$$

Using (12), we get:

(16) 
$$\tilde{M}_t^{\alpha,\theta} = Z_t e^{2\theta Z_t} e^{-\alpha/\tilde{c}_t^{\theta}} \sum_{i \in \mathbb{N}} \frac{(\alpha Z_t)^i}{i!(i+1)!} e^{-(i+1)2\beta\theta t}.$$

We then deduce from Proposition 3.2 the following corollary.

**Corollary 3.3.** Let  $\theta, \alpha \in \mathbb{R}$ . The process  $\tilde{M}^{\alpha,\theta}$  is a martingale under  $\mathbb{N}^{\theta}$ , and for t > 0 and any non-negative  $\mathcal{F}_t$ -measurable random variable  $H_t$ , we have:

(17) 
$$\mathbb{N}^{\theta}[H_t \, \tilde{M}_t^{\alpha, \theta}] = \mathbb{N}^{-\theta} \left[ H_t \, M_t^{\alpha, -\theta} \right]$$

Remark 3.4 (The case  $\theta = 0$  and  $\alpha = 0$ ). Let t > 0. For  $\theta = 0$ , we have:

$$\tilde{M}_t^{\alpha,0} = M_t^{\alpha,0} = Z_t e^{-\alpha\beta t} \sum_{i \in \mathbb{N}} \frac{(\alpha Z_t)^i}{i!(i+1)!}$$

For  $\alpha = 0$ , we have:

(18) 
$$M_t^{0,\theta} = Z_t e^{2\beta\theta t} \quad \text{and} \quad \tilde{M}_t^{0,\theta} = Z_t e^{2\theta(Z_t - \beta t)}.$$

Then for  $\alpha = \theta = 0$ , we have:

$$\tilde{M}_t^{0,0} = M_t^{0,0} = Z_t.$$

3.2. Local limit. We first consider the Poisson regime, whose name is inherited from the representation given in Proposition 4.1 based on a Poisson immigration. Let  $a = (a_t, t > 0)$  be a positive function.

**Proposition 3.5** (Poisson regime). Let  $\theta \in \mathbb{R}$ , s > 0 and  $H_s$  be a bounded non-negative  $\mathcal{F}_s$ -measurable random variable. Let  $\alpha \in (0, +\infty)$ . Assume the function a is such that as  $t \to \infty$  large:

$$a_t \sim \begin{cases} \alpha \beta^2 t^2, & \text{if } \theta = 0; \\ \alpha (2\theta)^{-2} e^{2\beta |\theta| t}, & \text{if } \theta \neq 0. \end{cases}$$

Then we have:

(19) 
$$\lim_{t \to \infty} \mathbb{N}^{\theta}[H_s | Z_t = a_t] = \mathbb{N}^{|\theta|} \left[ H_s M_s^{\alpha, |\theta|} \right] = \begin{cases} \mathbb{N}^{\theta}[H_s M_s^{\alpha, \theta}], & \text{if } \theta \ge 0; \\ \mathbb{N}^{\theta}[H_s \tilde{M}_s^{\alpha, \theta}], & \text{if } \theta \le 0. \end{cases}$$

Remark 3.6. Contrary to Proposition 3.2, it is not possible to deduce the result for any  $\theta$  from the result for  $\theta = 0$  in Proposition 3.5, since  $a_t$  is not continuous at  $\theta = 0$ .

*Proof.* Let s > 0 and  $H_s$  be fixed. For t > 0, we set:

$$A_t = \mathbb{N}^{\theta}[H_s | Z_{t+s} = a_{t+s}].$$

We omit most of the time the superscript  $\theta$  in this proof. Thanks to (10) and (11), we have:

$$A_t = \frac{\mathbb{N}\left[H_s q_t(Z_s, a_{t+s})\right]}{q_{t+s}(a_{t+s})}$$
$$= \frac{c_t \tilde{c}_t}{c_{t+s} \tilde{c}_{t+s}} e^{-a_{t+s}(c_t+2\theta-\tilde{c}_{t+s})} \mathbb{N}\left[H_s Z_s e^{-Z_s c_t} \sum_{i \in \mathbb{N}} \frac{\left(Z_s a_{t+s} c_t \tilde{c}_t\right)^i}{i!(i+1)!}\right]$$

As t goes to infinity, we have  $\tilde{c}_{t+r}c_{t+r} \sim (2\theta)^2 e^{-2\beta|\theta|t} e^{-2\beta|\theta|r}$  for  $\theta \neq 0$  and  $\tilde{c}_{t+r}c_{t+r} \sim (\beta t)^{-2}$  for  $\theta = 0$ . Thanks to the condition on a, it is tedious but elementary to check that:

0.01.01

$$\lim_{t \to \infty} c_t \tilde{c}_t / c_{t+s} \tilde{c}_{t+s} = e^{2\beta |\theta|s},$$
$$\lim_{t \to \infty} a_{t+s} (c_t + 2\theta - \tilde{c}_{t+s}) = \lim_{t \to \infty} a_{t+s} (\tilde{c}_t - \tilde{c}_{t+s}) = \begin{cases} \alpha/c_s & \text{if } \theta \ge 0, \\ \alpha/\tilde{c}_s & \text{if } \theta \le 0, \end{cases}$$
$$\lim_{t \to \infty} a_{t+s} c_t \tilde{c}_t = \alpha e^{2\beta |\theta|s}.$$

In particular, we get  $\lim_{t\to+\infty} D_t = D_{\infty}$ , where:

$$D_t = \frac{c_t \tilde{c}_t}{c_{t+s} \tilde{c}_{t+s}} Z_s e^{-Z_s c_t} \sum_{i \in \mathbb{N}} \frac{(Z_s a_{t+s} c_t \tilde{c}_t)^i}{i!(i+1)!},$$
$$D_{\infty} = Z_s \sum_{i \in \mathbb{N}} \frac{(\alpha Z_s)^i}{i!(i+1)!} \times \begin{cases} e^{(i+1)2\beta\theta s}, & \text{if } \theta \ge 0; \\ e^{2\theta Z_s - (i+1)2\beta\theta s}, & \text{if } \theta \le 0. \end{cases}$$

Using the definitions of  $M^{\alpha,\theta}$  and  $\tilde{M}^{\alpha,\theta}$ , see (12) and (16), we get that:

$$D_{\infty} = \begin{cases} e^{\alpha/c_s} & M_s^{\alpha,\theta}, & \text{if } \theta \ge 0; \\ e^{\alpha/\tilde{c}_s} & \tilde{M}_s^{\alpha,\theta}, & \text{if } \theta \le 0. \end{cases}$$

There exists a finite constant C such that, for t large enough:

$$a_{t+s}c_t \tilde{c}_t \le C e^{2\beta|\theta|s}$$

8

Recall  $-c_t \leq 2\theta$  (as  $c_t + 2\theta = \tilde{c}_t \geq 0$ ). We deduce that there exists a finite constant  $C_0$  such that, for t large enough:

$$D_t \le C_0 \left( M_s^{C,\theta} + \tilde{M}_s^{C,\theta} \right).$$

Since  $M_s^{C,\theta}$  and  $\tilde{M}_s^{C,\theta}$  are integrable, we then deduce that the first equality in (19) holds using the dominated convergence theorem. Then second equality in (19) is a consequence of (17).

We now consider the Kesten regime, whose proof is left to the reader as it is very similar to the proof in the Poisson regime when one takes  $\alpha = 0$  and uses (18).

**Proposition 3.7** (Kesten regime). Let  $\theta \in \mathbb{R}$ , s > 0 and  $H_s$  be a bounded non-negative  $\mathcal{F}_s$ -measurable random variable. Assume the function a is positive  $(a_t > 0)$  and such that  $as t \to \infty$ :

$$a_t = \begin{cases} o(t^2), & \text{if } \theta = 0; \\ o(e^{2\beta|\theta|t}), & \text{if } \theta \neq 0. \end{cases}$$

Then we have:

(20) 
$$\lim_{t \to \infty} \mathbb{N}^{\theta}[H_s | Z_t = a_t] = \mathbb{N}^{|\theta|} \left[ H_s Z_s e^{2\beta |\theta|s} \right] = \begin{cases} \mathbb{N}^{\theta}[H_s M_s^{0,\theta}], & \text{if } \theta \ge 0; \\ \mathbb{N}^{\theta}[H_s \tilde{M}_s^{0,\theta}], & \text{if } \theta \le 0. \end{cases}$$

For completeness, we add the well known extinction case, that is the function  $a_t = 0$  for large t, which is a direct consequence of (14).

**Proposition 3.8** (Extinction regime). Let  $\theta \in \mathbb{R}$ , s > 0 and  $H_s$  be a bounded non-negative  $\mathcal{F}_s$ -measurable random variable. Then we have:

(21) 
$$\lim_{t \to \infty} \mathbb{N}^{\theta}[H_s | Z_t = 0] = \mathbb{N}^{|\theta|}[H_s] = \begin{cases} \mathbb{N}^{\theta}[H_s], & \text{if } \theta \ge 0; \\ \mathbb{N}^{-\theta}[H_s], & \text{if } \theta \le 0. \end{cases}$$

#### 4. h-transform

We give a representation of the distribution of the process Z under the *h*-transform given by the martingale  $M^{\alpha,\theta}$ . The proof will be done for  $\beta = 1$  and  $\theta = 0$ , and then use a time-change, see (6), to get  $\theta \in \mathbb{R}$ .

4.1. **SDE representation.** Let  $\beta > 0$  and  $\theta \in \mathbb{R}$ . Let  $B = (B_t, t \ge 0)$  be a standard Brownian motion. Let  $\alpha > 0$  and  $S^{\alpha,\theta}(dt)$  be a Poisson point measure on  $\mathbb{R}_+$  with intensity  $\alpha\beta e^{2\beta\theta t} dt$  independent of the Brownian motion B. We set  $S_t^{\alpha,\theta} = S^{\alpha,\theta}([0,t])$  for  $t \in \mathbb{R}_+$ . We define the process  $Z^{\alpha} = (Z_t^{\alpha}, t \ge 0)$  under  $\mathbb{P}^{\theta}$  as the unique strong solution (conditionally on S) of the following SDE:

$$dZ_t^{\alpha} = \sqrt{2\beta Z_t^{\alpha}} \, dB_t - 2\beta \theta Z_t^{\alpha} dt + 2\beta \left( S_t^{\alpha,\theta} + 1 \right) dt \quad \text{for } t \ge 0, \quad \text{and } Z_0^{\alpha} = 0.$$

**Proposition 4.1.** Let  $\alpha > 0$ ,  $\theta \in \mathbb{R}$  and  $t_0 > 0$ . The process  $(Z_t, t \in [0, t_0])$  under  $\mathbb{N}^{\theta} \left[ \bullet M_{t_0}^{\alpha, \theta} \right]$ (resp. under  $\mathbb{N}^{\theta} \left[ \bullet \tilde{M}_{t_0}^{\alpha, \theta} \right]$ ) is distributed as the process  $(Z_t^{\alpha}, t \in [0, t_0])$  under  $\mathbb{P}^{\theta}$  (resp.  $\mathbb{P}^{-\theta}$ ).

The proof of this proposition is detailed in the next subsection.

4.2. Proof of Proposition 4.1. Following Remark 2.1, we first use a scaling argument to remove the parameters  $\beta$  and  $\theta$ .

Let  $\alpha > 0$ . Let  $S^{\alpha} = (S_t^{\alpha}, t \ge 0)$  be a Poisson process with parameter  $\alpha$  independent of the Brownian motion *B*. Let  $Y^{\alpha} = (Y_t^{\alpha}, t \ge 0)$  be the unique strong solution (conditionally on *S*) of the following SDE:

(22) 
$$dY_t^{\alpha} = \sqrt{2Y_t^{\alpha}} dB_t + 2 \left(S_t^{\alpha} + 1\right) dt \quad \text{for } t \ge 0, \quad \text{and } Y_0^{\alpha} = 0.$$

Let  $\beta, \alpha > 0$  and  $\theta \in \mathbb{R}$ , and write  $Z^{[\beta,\theta,\alpha]}$  for the process  $Z^{\alpha}$  under  $\mathbb{N}^{\theta}$  or  $\mathbb{P}^{\theta}$  to stress the dependence in  $\beta$  and  $\theta$ . Define the process  $(Y'^{\alpha}, S'^{\alpha}) = ((Y'^{\alpha}, S'^{\alpha}), s \in [0, 1/(2\theta_{-})))$  by:

(23) 
$$Y_s^{\prime\alpha} = e^{2\beta\theta t} Z_t^{[\beta,\theta,\alpha]} \quad \text{and} \quad S_s^{\prime\alpha} = S_t^{\alpha,\theta}, \quad \text{with} \quad s = \frac{1}{c_t^{\theta}}$$

Then, it is elementary that this deterministic time change yields the following result.

**Lemma 4.2.** Let  $\beta, \alpha > 0$  and  $\theta \in \mathbb{R}$ . The process  $(Y'^{\alpha}, S'^{\alpha})$  under  $\mathbb{P}^{\theta}$  (whose law depends on  $(\beta, \theta)$  and  $\alpha$ ) is distributed as  $((Y_s^{\alpha}, S_s^{\alpha}), s \in [0, 1/(2\theta_-)))$ .

Let  $(P_t, t \ge 0)$  be the transition semi-group on  $\mathbb{R}_+ \times \mathbb{N}$  of the Markov process  $(Y^{\alpha}, S^{\alpha})$ .

**Lemma 4.3.** The semi-group  $(P_t, t \ge 0)$  is Feller, that is for all  $t \ge 0$  and all bounded continuous function f defined on  $\mathbb{R}_+ \times \mathbb{N}$ , the function  $P_t(f)$  is also bounded and continuous.

Proof. Let  $((Y_t^{\alpha,(x,s)}, S_t^{\alpha,(x,s)}), t \ge 0)$  denote the solution of the SDE (22) starting from (x, s). Let  $(X_t^x, t \ge 0)$  be a Feller process starting from x (it is distributed as a solution to the SDE (5)) independent of the  $(Y_t^{\alpha,(x,s)}, S_t^{\alpha,(x,s)})_{t\ge 0}$ . We denote by  $Q_t$  the semi-group of the process  $X^x$  and recall that  $Q_t$  is a Feller semi-group. By the branching property, we have the equality in distribution for the processes:

$$\left( (Y_t^{\alpha,(x,s)}, S_t^{\alpha,(x,s)}), t \ge 0 \right) \stackrel{(d)}{=} \left( (Y_t^{\alpha,(0,s)} + X_t^x, S_t^{\alpha,(0,s)}), t \ge 0 \right).$$

Then for every  $t \ge 0$ ,  $x, y \in \mathbb{R}_+$ ,  $s \in \mathbb{N}$  and every bounded continuous function f defined on  $\mathbb{R}_+ \times \mathbb{N}$ , we have:

$$P_{t}f(x,s) - P_{t}f(y,s) = \mathbb{E}\left[f\left(Y_{t}^{\alpha,(x,s)}, S_{t}^{\alpha,(x,s)}\right) - f\left(Y_{t}^{\alpha,(y,s)}, S_{t}^{\alpha,(y,s)}\right)\right]$$
$$= \mathbb{E}\left[f\left(Y_{t}^{\alpha,(0,s)} + X_{t}^{x}, S_{t}^{\alpha,(0,s)}\right) - f\left(Y_{t}^{\alpha,(0,s)} + X_{t}^{y}, S_{t}^{\alpha,(0,s)}\right)\right]$$
$$= \mathbb{E}\left[Q_{t}f_{\left(Y_{t}^{\alpha,(0,s)}, S_{t}^{\alpha,(0,s)}\right)}(x) - Q_{t}f_{\left(Y_{t}^{\alpha,(0,s)}, S_{t}^{\alpha,(0,s)}\right)}(y)\right]$$

where  $f_{(y,s)}$  is the continuous map  $x \mapsto f(y+x,s)$ . By the Feller property of the semi-group  $Q_t$ and the dominated convergence theorem, we deduce that  $\lim_{y\to x} P_t f(x,s) - P_t f(y,s) = 0$ . This gives the Feller property of the kernel  $P_t$ .

We now give the density of  $(Y_t^{\alpha}, S_t^{\alpha})$ . Recall that  $Y_0^{\alpha} = S_0^{\alpha} = 0$ . Let N be the counting measure on N.

**Lemma 4.4.** Let t > 0. The random variable  $(Y_t^{\alpha}, S_t^{\alpha})$  has a density f on  $\mathbb{R}_+ \times \mathbb{N}$  with respect to  $dy \otimes N(dk)$  given by:

(24) 
$$f(y,k) = \frac{1}{t^2} \frac{\alpha^k y^{k+1}}{k!(k+1)!} e^{-(\alpha t + t^{-1}y)}, \quad y \ge 0, \ k \in \mathbb{N}.$$

*Proof.* Conditionally on S, by Definition (22), we can see  $Y^{\alpha}$  as a quadratic CSBP process (with  $\beta = 1$ ) with immigration whose rate is  $2(S_t^{\alpha} + 1)dt$ . This implies that, conditionally on  $S^{\alpha}$ , the process  $Y^{\alpha}$  is distributed as  $\left(\sum_{i \in I} \mathbf{1}_{\{h_i \leq t\}} Z_{t-h_i}^{(i)}, t \geq 0\right)$ , where  $\sum_{i \in I} \delta_{(h_i, Z^{(i)})}(dt, dZ)$  is a Poisson point measure on  $\mathbb{R}_+ \times \mathcal{C}[0, \mathbb{R}_+)$  with intensity  $2(S_t^{\alpha} + 1)dt \mathbb{N}[dZ]$  and  $\mathbb{N}$  is the excursion measure of a CSBP with branching mechanism  $\psi(\lambda) = \lambda^2$ .

We deduce that for  $\lambda, \mu \geq 0$ :

$$\mathbb{E}\left[\mathrm{e}^{-\lambda Y_t^{\alpha}-\mu S_t^{\alpha}}\right] = \mathbb{E}\left[\mathrm{e}^{-\mu S_t^{\alpha}-\int_0^t 2(S_r^{\alpha}+1)\mathbb{N}\left[1-\mathrm{e}^{-\lambda Z_{t-r}}\right]\mathrm{d}r}\right] = \mathbb{E}\left[\mathrm{e}^{-\mu S_t^{\alpha}-2\int_0^t (S_r^{\alpha}+1)\frac{\lambda}{1+(t-r)\lambda}\,\mathrm{d}r}\right],$$

where we used (8) for the last equality (with  $\beta = 1$  and  $\theta = 0$ ). Denote by  $(\xi_i, i \in \mathbb{N}^*)$  the increasing sequence of the jumping times of the Poisson process  $S^{\alpha}$ , and set  $\xi_0 = 0$ . Then, we have on  $\{S_t^{\alpha} = k\}$ :

$$\int_{0}^{t} (S_{r}^{\alpha} + 1) \frac{\lambda}{1 + (t - r)\lambda} dr = \sum_{i=0}^{k} (i + 1) \int_{\xi_{i}}^{\xi_{i+1} \wedge t} \frac{\lambda}{1 + (t - r)\lambda} dr$$
$$= -\sum_{i=0}^{k} (i + 1) \log(1 + (t - r)\lambda) \Big|_{\xi_{i}}^{\xi_{i+1} \wedge t}$$
$$= \sum_{i=0}^{k} \log(1 + (t - \xi_{i})\lambda).$$

Conditionally on  $\{S_t^{\alpha} = k\}$ , the random set  $\{\xi_1, \ldots, \xi_k\}$  is distributed as  $\{tU_1, \ldots, tU_k\}$  (notice the order is unimportant and is not preserved), where  $U_1, \ldots, U_k$  are independent random variables uniformly distributed on [0, 1]. We deduce that:

$$\mathbb{E}\left[e^{-\lambda Y_t^{\alpha}-\mu S_t^{\alpha}}\right] = \sum_{k\in\mathbb{N}} \frac{(\alpha t)^k e^{-\alpha t-\mu k}}{k!} \mathbb{E}\left[\prod_{i=1}^k (1+t(1-U_i)\lambda)^{-2}\right] (1+t\lambda)^{-2}$$
$$= \sum_{k\in\mathbb{N}} \frac{(\alpha t)^k e^{-\alpha t-\mu k}}{k!} (1+t\lambda)^{-k-2}$$
$$= \sum_{k\in\mathbb{N}} \int_{\mathbb{R}_+} \mathrm{d}y \ f(y,k) e^{-\lambda y-\mu k},$$

where for the last equality, we used the definition of f given in (24). This finishes the proof.  $\Box$ 

Let  $q'_t$  be the distribution of  $Y_t^{\alpha}$  for  $t \in \mathbb{R}_+$ . We have  $q'_0 = \delta_0$  the Dirac mass at 0, and for t > 0, we deduce from Lemma 4.4 that  $q'_t(dy)$  has a density, also denoted by  $q'_t$ , on  $\mathbb{R}_+$  with respect to the Lebesgue measure given by:

$$q'_t(y) = t^{-2} e^{-(\alpha t + t^{-1}y)} \sum_{k \in \mathbb{N}} \frac{\alpha^k y^{k+1}}{k!(k+1)!}, \quad t > 0, \ y \ge 0.$$

We now give some properties of the conditional law of  $S_t$  given  $Y_t$ .

**Lemma 4.5.** Let  $y \in \mathbb{R}_+$ . The law of  $S_t^{\alpha}$  conditionally on  $\{Y_t^{\alpha} = y\}$  does not depend on t. More precisely, we get for all  $t \ge 0$ ,  $k \in \mathbb{N}$  and  $y \ge 0$ :

(25) 
$$\mathbb{P}(S_t^{\alpha} = k | Y_t^{\alpha} = y) = C^{-1} \frac{(\alpha y)^k}{k!(k+1)!} \quad with \quad C = \sum_{j \in \mathbb{N}} \frac{(\alpha y)^j}{j!(j+1)!} \in [1, +\infty).$$

*Proof.* Using Lemma 4.4, we directly have (25) for t > 0. Notice that for y = 0, we have C = 1 and

$$C^{-1} \frac{(\alpha y)^k}{k!(k+1)!} = \mathbf{1}_{\{k=0\}}.$$

As  $(Y_0^{\alpha}, S_0^{\alpha}) = (0, 0)$ , we deduce that (25) also holds for t = 0.

We can now prove the Markov property of the process  $Y = (Y_t, t \ge 0)$ .

**Lemma 4.6.** The process  $Y^{\alpha}$  is Markov, and its transition semi-group  $(Q_t, t \in \mathbb{R}_+)$  is the unique Feller semi-group such that  $q'_t = q'_0Q_t$  for  $t \in \mathbb{R}_+$ , with  $q'_t$  the distribution of  $Y^{\alpha}_t$ .

Proof. We say a probability kernel K is continuous if for all continuous and bounded function f, Kf is also continuous (and bounded). We shall check hypothesis from [33, Lemma 1]. With the notation therein  $(X = (Y^{\alpha}, S^{\alpha}) \text{ and } \phi(y, s) = y)$ , the semi-group  $(P_t, t \ge 0)$  is Feller, see Lemma 4.3. The probability kernel  $\Lambda(y; dz, dk) = \mathbb{P}(S_t^{\alpha} = k | Y_t^{\alpha} = y) \, \delta_y(dz) \, \mathrm{N}(dk)$  is clearly continuous and does not depend on t. The probability kernel  $\Phi(y, k; dz) = \delta_y(dz)$  is also clearly continuous. Lemma 4.5 gives exactly condition (i) in [33, Lemma 1]. We now check condition (ii) in [33, Lemma 1], that is the one-dimensional marginal distributions of  $Y^{\alpha}$ ,  $(q'_t, t \in \mathbb{R}_+)$ , are determining, that is if h and g are bounded continuous functions defined on  $\mathbb{R}_+$ , then  $\mathbb{E}[h(Y_t^{\alpha})] = \mathbb{E}[g(Y_t^{\alpha})]$  for all  $t \in \mathbb{R}_+$  implies h = g. To prove this, notice that:

$$t^2 e^{\alpha t} \mathbb{E}[h(Y_t^{\alpha})] = \int_{\mathbb{R}_+} e^{-t^{-1}y} H(y) dy,$$

where  $H(y) = h(y) \sum_{k \in \mathbb{N}} \frac{\alpha^k y^{k+1}}{k!(k+1)!}$ . As the Laplace transform characterizes the bounded continuous function, we deduce that if  $\mathbb{E}[h(Y_t^{\alpha})] = \mathbb{E}[g(Y_t^{\alpha})]$  for all  $t \in \mathbb{R}_+$ , then H = G (with  $G(y) = g(y) \sum_{k \in \mathbb{N}} \frac{\alpha^k y^{k+1}}{k!(k+1)!}$ ) and thus h = g on  $(0, +\infty)$  and by continuity on  $\mathbb{R}_+$ .

As the assumption of [33, Lemma 1] are satisfied, we deduce that  $Y^{\alpha}$  is a Markov process, and that its transition semi-group  $(Q_t, t \in \mathbb{R}_+)$  is the unique Feller semi-group such that  $q'_t = q'_0 Q_t$  for  $t \in \mathbb{R}_+$ , with  $q'_t$  the distribution of  $Y_t^{\alpha}$ .

We now compare the distribution of  $Y^{\alpha}$  and the distribution of the Feller diffusion Y defined in Remark 2.1, which is a CSBP with parameter  $\beta = 1$  and  $\theta = 0$ . Following (13), we set for t > 0:

$$M_t^{\alpha} = \sqrt{\frac{Y_t}{\alpha}} e^{-\alpha t} I_1\left(2\sqrt{\alpha Y_t}\right).$$

Let  $\mathbb{N}$  denote the canonical measure of Y.

**Lemma 4.7.** Let  $\alpha > 0$ . Let  $t_0 > 0$ . The process  $(Y_t^{\alpha}, t \in [0, t_0])$  has the same distribution as the process  $(Y_t, t \in [0, t_0])$  under  $\mathbb{N} \left[ \bullet M_{t_0}^{\alpha} \right]$ .

*Proof.* We first check the two processes have the same one-dimensional marginals. Clearly  $Y_0^{\alpha} = Y_0 = 0$ . Let t > 0. According to Lemma 2.3, the entrance law of  $Y_t$  under  $\mathbb{N}$  has density  $y \mapsto t^{-2} e^{-y/t}$ . We deduce that for  $\lambda \geq 0$ :

$$\mathbb{N}\left[\mathrm{e}^{-\lambda Y_t} M_t^{\alpha}\right] = \int_{\mathbb{R}_+} \mathrm{e}^{-\lambda y} \ y \, \mathrm{e}^{-\alpha t} \sum_{i \in \mathbb{N}} \frac{(\alpha y)^i}{i!(i+1)!} \ t^{-2} \, \mathrm{e}^{-y/t} \ \mathrm{d}y$$
$$= \int_{\mathbb{R}_+} \mathrm{e}^{-\lambda y} \ t^{-2} \, \mathrm{e}^{-(\alpha t+t^{-1}y)} \sum_{i \in \mathbb{N}} \frac{\alpha^i y^{i+1}}{i!(i+1)!} \ \mathrm{d}y$$
$$= \mathbb{E}\left[\mathrm{e}^{-\lambda Y_t^{\alpha}}\right].$$

Since the Laplace transform characterizes the probability distribution on  $\mathbb{R}_+$ , we obtain that  $Y_t^{\alpha}$  has the same distribution as  $Y_t$  under  $\mathbb{N} [\bullet M_t^{\alpha}]$ .

Using Doob's *h*-transform, we get that the process  $(Y_t, t \in [0, t_0])$  under  $\mathbb{N} \left[ \bullet M_{t_0}^{\alpha} \right]$  is Markov. Using that  $M^{\alpha}$  is a martingale under  $\mathbb{N}$  (see Proposition 3.2 and use that Y is distributed as Z when  $\beta = 1, \theta = 0$ ), that  $M_t^{\alpha}$  is a function of  $Y_t$ , and that Y is Feller under  $\mathbb{N}$ , we get that  $(Y_t, t \in [0, t_0])$  under  $\mathbb{N} \left[ \bullet M_{t_0}^{\alpha} \right]$  is also Feller. We deduce from the uniqueness property of Lemma 4.6 and the identification of the one-dimensional marginals from the first step of the proof, that  $(Y_t^{\alpha}, t \in [0, t_0])$  has the same distribution as  $(Y_t, t \in [0, t_0])$  under  $\mathbb{N} \left[ \bullet M_{t_0}^{\alpha} \right]$ .

We can now give the proof of Proposition 4.1. Let  $\beta, \alpha > 0, \theta \in \mathbb{R}$  and  $t_0 > 0$ . Using the time changes given by (6), (13) and (23), we deduce that the process  $(Z_t^{\alpha}, t \in [0, t_0])$  under  $\mathbb{P}^{\theta}$  is distributed as the process  $(Z_t, t \in [0, t_0])$  under  $\mathbb{N}^{\theta} \left[ \bullet M_{t_0}^{\alpha, \theta} \right]$ . Then, using Corollary 3.3, we also deduce that the process  $(Z_t^{\alpha}, t \in [0, t_0])$  under  $\mathbb{P}^{-\theta}$  is distributed as the process  $(Z_t, t \in [0, t_0])$  under  $\mathbb{P}^{-\theta}$  is distributed as the process  $(Z_t, t \in [0, t_0])$  under  $\mathbb{N}^{\theta} \left[ \bullet \tilde{M}_{t_0}^{\alpha, \theta} \right]$ . This finishes the proof of Proposition 4.1.

#### 5. TOPOLOGY ON SET OF TREES

In Section 5.1, we recall the usual basic definitions and notations for rooted real trees. In Section 5.2 (resp. Section 5.3), we consider the Polish space of equivalent classes of compact (resp. locally compact) rooted trees with distinguished vertices endowed with the Gromov-Hausdorff distance. We define various grafting measurable operations (denoted by  $\mathfrak{S}^*_*$ ) of a tree on an another tree in Section 5.4. Motivated by the fact that some random trees are obtained as decorated backbone trees, we introduce in Section 5.5 the space of marked trees, that is of trees with a distinguished sub-tree (or backbone tree). We also establish in this section the measurability of various truncation maps. The short Section 5.6 is devoted to special case of the backbone tree being reduced to an infinite spine (this is the case for the Kesten tree). In Section 5.7, we consider specifically discrete trees which are spanned by n distinguished vertices. and describe them as a set of branches indexed by all the possible subsets of the n distinguished vertices. This description is then used in Section 5.8 to split (with a function  $Split_n$ ) a locally tree with n distinguished vertices as sub-trees supported by the different branches of the discrete tree spanned by the distinguished vertices. Then, we provide in a sense the inverse construction in Section 5.9 where (with a function  $\operatorname{Graft}_n$ ) we decorate the branches of a discrete trees with subtrees. In Section 5.10, we describe a measurable procedure to decorate a branch with a family of sub-trees given by the atoms of a point measure on the set of trees (the function Tree) and a measurable procedure to describe the decoration of a distinguished branch of a tree (the function  $\mathcal{M}$ ) through a point measure on the set of trees.

In a nutshell, the main objective of this section is to define the grafting and splitting functions, as well as the decorating and de-decorating functions in a measurable way on the set of locally compact rooted real trees. An index of all the (numerous) relevant notations of this section is provided at the end of the document.

5.1. Notations and definitions for trees. We use as usual the framework of real trees to encode the genealogy of a continuous state branching process. We refer to [21] for a detailed introduction to real trees.

A real tree (or simply a tree in the rest of the text) is a metric space (T, d) that satisfies the two following properties for every  $u, v \in T$ :

(i) There is a unique isometric map  $f_{u,v}$  from [0, d(u, u)] into T such that

 $f_{u,v}(0) = u$  and  $f_{u,v}(d(u,v)) = v$ .

(ii) If  $\varphi$  is a continuous injective map from [0, 1] into T such that  $\varphi(0) = u$  and  $\varphi(1) = v$ , then the range of  $\varphi$  is also the range of  $f_{u,v}$ .

The range of the map  $f_{u,v}$  is denoted by  $\llbracket u, v \rrbracket$ . It is the unique continuous path that links u to v in the tree. We will write  $\llbracket u, v \rrbracket$  (resp.  $\llbracket u, v \rrbracket$ ,  $\llbracket u, v \rrbracket$ ) for  $\llbracket u, v \rrbracket \setminus \{v\}$  (resp.  $\llbracket u, v \rrbracket \setminus \{u\}, \llbracket u, v \rrbracket \setminus \{u, v\}$ ).

A rooted tree is a tree (T, d) with a distinguished vertex denoted by  $\rho$  and called the root. We always consider rooted trees in this work. For an element x of a rooted tree  $(T, d, \rho)$ , we denote by  $H(x) = d(\rho, x)$  its height, and we set  $H(T) = \sup_{x \in T} H(x)$  the height of the tree T.

The set of leaves Lf(T) of T is  $\{\varrho\}$  if  $T = \{\varrho\}$  is reduced to its root and the set of  $x \in T \setminus \{\varrho\}$ such that  $T \setminus \{x\}$  has only one connected component. The skeleton of the tree is the set  $Sk(T) = T \setminus Lf(T)$ . The set of branching points (or vertices) Br(T) is the set of  $x \in T$  such that  $T \setminus \{x\}$  has at least 3 connected components if  $x \neq \varrho$  or at least 2 components if  $x = \varrho$ .

For a vertex  $x \in T$ , we define the subtree  $T_x$  "above" x as:

$$T_x = \{ y \in T : x \in \llbracket \varrho, y \rrbracket \}.$$

The real tree  $T_x$  is endowed with the distance induced by T and will be rooted at x.

We define a partial order  $\prec$  (called the genealogical order) on a rooted tree  $(T, d, \varrho)$  by:

$$u \prec v \iff u \in \llbracket \varrho, v \rrbracket$$

and we say in this case that u is an ancestor of v. If  $u, v \in T$ , we denote by  $u \wedge v$  the most recent common ancestor of u and v, *i.e.* the unique vertex of T such that:

$$\llbracket \varrho, u \rrbracket \cap \llbracket \varrho, v \rrbracket = \llbracket \varrho, u \wedge v \rrbracket.$$

The trace of the Borel  $\sigma$ -field of T on Sk(T) is generated by the sets [s, s'],  $s, s' \in Sk(T)$  (see [22]). Hence, there exists a  $\sigma$ -finite Borel measure  $\mathscr{L}^T$  on T, such that

$$\mathscr{L}^T(\mathrm{Lf}(T)) = 0 \text{ and } \mathscr{L}^T(\llbracket s, s' \rrbracket) = d(s, s').$$

This measure  $\mathscr{L}^T$  is called the length measure on T. When there is no ambiguity, we simply write  $\mathscr{L}$  for  $\mathscr{L}^T$ .

5.2. Set of (equivalence classes of) *n*-pointed compact trees. Let  $n \in \mathbb{N}$ . A rooted *n*-pointed tree  $(T, d, \mathbf{v})$  is a tree (T, d) with a root  $\rho$  and *n*-distinguished (possibly equal) vertices  $v_1, \ldots, v_n \in T$ , with the notation  $\mathbf{v} = (v_0 = \rho, v_1, \ldots, v_n)$ .

A correspondence  $\mathcal{R}$  between two rooted *n*-pointed trees  $(T, d, \mathbf{v})$  and  $(T', d', \mathbf{v}')$  is a subset of  $T \times T'$  such that for all  $x \in T$  (resp.  $x' \in T'$ ), there exists  $x' \in T'$  (resp.  $x \in T$ ) such that  $(x, x') \in \mathcal{R}$ , and for all  $0 \leq k \leq n$ , we have  $(v_k, v'_k) \in \mathcal{R}$ , where  $\mathbf{v} = (v_0 = \varrho, \ldots, v_n)$  and  $\mathbf{v}' = (v'_0 = \varrho', \ldots, v'_n)$ . The distortion of  $\mathcal{R}$  is defined as:

dist 
$$(\mathcal{R}) = \sup \Big\{ |d(x,y) - d'(x',y')| : (x,x'), (y,y') \in \mathcal{R} \Big\}.$$

For two compact rooted *n*-pointed trees  $T = (T, d, \mathbf{v})$  and  $T' = (T', d', \mathbf{v}')$ , we set:

$$d_{\mathrm{GH}}^{(n)}(T,T') = \inf \frac{1}{2} \operatorname{dist} (\mathcal{R}),$$

where the infimum is taken over all the correspondences  $\mathcal{R}$  between  $(T, d, \mathbf{v})$  and  $(T', d', \mathbf{v}')$ . The function  $d_{\text{GH}}^{(n)}$  is the so-called Gromov-Hausdorff pseudo-distance, see [30]. Furthermore, we have that  $d_{\text{GH}}^{(n)}(T, T') = 0$  if and only if there exists an isometric one-to-one map  $\varphi$  from (T, d) to (T', d')

14

which preserves the root and the distinguished vertices (that is  $\varphi(v_k) = v'_k$  for all  $0 \le k \le n$ ). The relation  $d_{\text{GH}}^{(n)}(T,T') = 0$  defines an equivalence relation between compact rooted *n*-pointed trees. The set  $\mathbb{T}_{\text{K}}^{(n)}$  of equivalence classes of compact rooted *n*-pointed trees endowed with  $d_{\text{GH}}^{(n)}$ is then a metric Polish space, see [30, Proposition 9]. We simply write  $(T, \mathbf{v})$  for  $(T, d, \mathbf{v})$ , and unless specified otherwise, we shall denote also by  $(T, \mathbf{v})$  its equivalence class. For n = 0, we simply write  $\mathbb{T}_{\text{K}}$  and  $d_{\text{GH}}$  for  $\mathbb{T}_{\text{K}}^{(n)}$  and  $d_{\text{GH}}^{(n)}$  and T for  $(T, d, \varrho)$ .

For a rooted *n*-pointed tree  $(T, d, \mathbf{v})$ , with  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$ , we denote the corresponding spanned tree Span<sup>°</sup> $(T, \mathbf{v})$  as:

(26) 
$$\operatorname{Span}^{\circ}(T, \mathbf{v}) = \bigcup_{k=1}^{n} \llbracket \varrho, v_k \rrbracket.$$

The tree  $(\text{Span}^{\circ}(T, \mathbf{v}), d, \varrho)$  will be simply denoted by  $\text{Span}^{\circ}(T, \mathbf{v})$ , whereas we will denote by  $\text{Span}(T, \mathbf{v})$  the rooted *n*-pointed tree  $(\text{Span}^{\circ}(T, \mathbf{v}), d, \mathbf{v})$ . For  $y \in T$ , we also define  $p_{\mathbf{v}}(y)$ , the projection of y on  $\text{Span}^{\circ}(T, \mathbf{v})$ , as the only point of  $\text{Span}^{\circ}(T, \mathbf{v})$  such that:

(27) 
$$\llbracket \varrho, y \rrbracket \cap \operatorname{Span}^{\circ}(T, \mathbf{v}) = \llbracket \varrho, p_{\mathbf{v}}(y) \rrbracket$$

Let us state a technical result which will be used several times in what follows.

**Lemma 5.1.** Let  $n \in \mathbb{N}$ . Let  $(T, d, \mathbf{v})$  and  $(T', d', \mathbf{v}')$  be two compact rooted n-pointed trees and let  $\mathcal{R}$  be a correspondence between them. For every  $(x, x') \in \mathcal{R}$  with  $x' \in \text{Span}^{\circ}(T', \mathbf{v}')$ , we have:

$$d(x, p_{\mathbf{v}}(x)) \le \frac{3}{2} \text{dist} (\mathcal{R}).$$

*Proof.* Let  $(x, x') \in \mathcal{R}$  with  $x' \in \text{Span}^{\circ}(T', \mathbf{v}')$ . First remark that there exist  $k, \ell \in \{0, \ldots, n\}$  such that  $p_{\mathbf{v}}(x) \in [\![v_k, v_\ell]\!]$  and  $x' \in [\![v'_k, v'_\ell]\!]$ . Indeed, let us set:

$$A = \{ v_k \colon p_{\mathbf{v}}(x) \in \llbracket \varrho, v_k \rrbracket \} \quad \text{and} \quad A' = \{ v'_k \colon x' \in \llbracket \varrho', v'_k \rrbracket \}.$$

Notice that  $A \neq \emptyset$  and  $A' \neq \emptyset$ . If there exists  $k \ge 1$  such that  $v_k \in A$  and  $v'_k \in A'$ , then one can take  $\ell = 0$  so that  $v_\ell = \rho$  and  $v'_\ell = \rho'$ . Otherwise, take k and  $\ell$  with  $k \neq \ell$  such that  $v_k \in A$  and  $v'_\ell \in A'$ . In this case, we get  $v_\ell \notin A$ . Clearly we have  $p_{\mathbf{v}}(x) \in [v_k, v_\ell]$  and by a similar argument,  $x' \in [v'_k, v'_\ell]$ . Therefore, we have:

$$2d(x, p_{\mathbf{v}}(x)) = d(x, v_k) + d(x, v_\ell) - d(v_k, v_\ell) \le d'(x', v'_k) + d'(x', v'_\ell) - d'(v'_k, v'_\ell) + 3 \operatorname{dist}(\mathcal{R}).$$

Then, use that  $d'(x', v'_k) + d'(x', v'_\ell) - d'(v'_k, v'_\ell) = 0$ , as  $x' \in \llbracket v'_k, v'_\ell \rrbracket$ , to conclude.  $\Box$ 

If  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  belong to the same equivalence class in  $\mathbb{T}_{\mathrm{K}}^{(n)}$ , then so do  $\mathrm{Span}(T, \mathbf{v})$  and  $\mathrm{Span}(T', \mathbf{v}')$  in  $\mathbb{T}_{\mathrm{K}}^{(n)}$ . Therefore, the function  $(T, \mathbf{v}) \mapsto \mathrm{Span}(T, \mathbf{v})$  is well defined from  $\mathbb{T}_{\mathrm{K}}^{(n)}$  to  $\mathbb{T}_{\mathrm{K}}^{(n)}$ . A first consequence of Lemma 5.1 is that this function is Lipschitz continuous; this result will be completed in Lemma 5.6.

**Lemma 5.2** (Continuity of the map Span). Let  $n \in \mathbb{N}$ . The map  $(T, \mathbf{v}) \mapsto \text{Span}(T, \mathbf{v})$  is 4-Lipschitz continuous from  $\mathbb{T}_{K}^{(n)}$  to  $\mathbb{T}_{K}^{(n)}$ .

*Proof.* Let  $(T, \mathbf{v}), (T', \mathbf{v}')$  be two compact rooted *n*-pointed trees and let  $\mathcal{R}$  be a correspondence between them. Let us set with obvious notations:

(28) 
$$\tilde{\mathcal{R}} = \left\{ \left( x, p'_{\mathbf{v}'}(x') \right) \colon (x, x') \in \mathcal{R}, \ x \in \operatorname{Span}^{\circ}(T, \mathbf{v}) \right\} \\ \cup \left\{ \left( p_{\mathbf{v}}(x), x' \right) \colon (x, x') \in \mathcal{R}, \ x' \in \operatorname{Span}^{\circ}(T', \mathbf{v}') \right\}.$$

Clearly,  $\hat{\mathcal{R}}$  is a correspondence between  $\text{Span}(T, \mathbf{v})$  and  $\text{Span}(T', \mathbf{v}')$ . We now compute its distortion. We consider the case  $x \in \text{Span}(T, \mathbf{v})$ ,  $y' \in \text{Span}(T', \mathbf{v}')$  and  $(x, x'), (y, y') \in \mathcal{R}$ , so that  $(x, p'_{\mathbf{v}'}(x'))$  and  $(p_{\mathbf{v}}(y), y')$  belong to  $\hat{\mathcal{R}}$ . We have:

$$\begin{aligned} \left| d(x, p_{\mathbf{v}}(y)) - d'(p'_{\mathbf{v}'}(x'), y') \right| &= \left| d(x, y) - d(y, p_{\mathbf{v}}(y)) - d'(x', y') + d'(x', p_{\mathbf{v}'}(x')) \right| \\ &\leq \left| d(x, y) - d'(x', y') \right| + d(y, p_{\mathbf{v}}(y)) + d'(x', p_{\mathbf{v}'}(x')) \\ &\leq 4 \operatorname{dist} (\mathcal{R}), \end{aligned}$$

where we used Lemma 5.1 for the last inequality. The other cases can be treated similarly. This implies that dist  $(\tilde{\mathcal{R}}) \leq 4 \operatorname{dist}(\mathcal{R})$  and thus, by definition of  $d_{\operatorname{GH}}^{(n)}$ :

$$d_{\mathrm{GH}}^{(n)} \big( \mathrm{Span}(T, \mathbf{v}), \mathrm{Span}(T', \mathbf{v}') \big) \le 4 \, d_{\mathrm{GH}}^{(n)} \big( (T, \mathbf{v}), (T', \mathbf{v}') \big).$$

5.3. Set of (equivalence classes of) rooted *n*-pointed locally compact trees. Recall the definition of the height  $H(x) = d(\varrho, x)$  of a vertex x in a rooted tree  $(T, d, \rho)$ . For a rooted *n*-pointed tree  $(T, d, \mathbf{v})$  and  $t \ge 0$ , we define the rooted *n*-pointed tree T truncated at level t as  $(r_t(T, \mathbf{v}), d, \mathbf{v})$  with:

(29) 
$$r_t(T, \mathbf{v}) = \left\{ x \in T : H(x) \le t \right\} \cup \left\{ \operatorname{Span}^{\circ}(T, \mathbf{v}) \right\},$$

and the distance on  $r_t(T, \mathbf{v})$  is given by the restriction of the distance d. We shall simply write  $r_t(T, \mathbf{v})$  for  $(r_t(T, \mathbf{v}), d, \mathbf{v})$ . If  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  are in the same equivalence class of  $\mathbb{T}_{\mathrm{K}}^{(n)}$ , so are  $r_t(T, \mathbf{v})$  and  $r_t(T', \mathbf{v}')$ . Thus the function  $r_t$  can be seen as a map from  $\mathbb{T}_{\mathrm{K}}^{(n)}$  to itself. When n = 0, we shall simply write  $r_t(T)$  for  $r_t(T, \varrho)$ . The next lemma is about the continuity of  $r_t$ .

**Lemma 5.3** (Continuity of  $r_t$ ). Let  $n \in \mathbb{N}$ . For  $s, t \ge 0$  and  $(T, \mathbf{v}), (T', \mathbf{v}') \in \mathbb{T}_{\mathrm{K}}^{(n)}$ , we have:

(30) 
$$d_{\text{GH}}^{(n)}\left(r_t(T, \mathbf{v}), r_{t+s}(T', \mathbf{v}')\right) \le 4 \, d_{\text{GH}}^{(n)}\left((T, \mathbf{v}), (T', \mathbf{v}')\right) + s$$

The map  $(t, (T, \mathbf{v})) \mapsto r_t(T, \mathbf{v})$  is continuous from  $\mathbb{R}_+ \times \mathbb{T}_{\mathrm{K}}^{(n)}$  to  $\mathbb{T}_{\mathrm{K}}^{(n)}$ .

*Proof.* Let  $(T, d, \mathbf{v}), (T', d', \mathbf{v}')$  be two compact rooted *n*-pointed trees. Firstly, notice that  $d_{\text{GH}}^{(n)}(r_{t+s}(T, \mathbf{v}), r_t(T, \mathbf{v})) \leq s$ . Secondly, recall Definition (27) of the projection  $p_{\mathbf{v}}$  on  $\text{Span}^{\circ}(T, \mathbf{v})$ . For  $y \in T$ , we also define the projection  $p_t(y)$  on  $r_t(T, \mathbf{v})$  as the only point of  $r_t(T, \mathbf{v})$  such that:

$$\llbracket \varrho, y \rrbracket \cap r_t(T, \mathbf{v}) = \llbracket \varrho, p_t(y) \rrbracket$$

We first prove the analogue of Lemma 5.1. Let  $\mathcal{R}$  be a correspondence between  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$ . Let  $(x, x') \in \mathcal{R}$  with  $x' \in r_t(T', \mathbf{v}')$ . By construction, we have  $p_t(x) \in [\![p_{\mathbf{v}}(x), x]\!]$ . If  $x' \in \operatorname{Span}(T', \mathbf{v}')$ , then we deduce from Lemma 5.1 that  $d(x, p_t(x)) \leq d(x, p_{\mathbf{v}}(x)) \leq \frac{3}{2} \operatorname{dist}(\mathcal{R})$ . If  $x' \in r_t(T', \mathbf{v}') \setminus \operatorname{Span}(T', \mathbf{v}')$ , then we have  $H(x') \leq t$  and thus  $H(x) = d(\varrho, x) \leq d'(\varrho', x') + \operatorname{dist}(\mathcal{R}) \leq t + \operatorname{dist}(\mathcal{R})$ , which implies that  $d(x, p_t(x)) \leq \operatorname{dist}(\mathcal{R})$ . In conclusion, we get  $d(x, p_t(x)) \leq \frac{3}{2} \operatorname{dist}(\mathcal{R})$ . Now, arguing as in the proof of Lemma 5.2, we deduce that  $d_{\operatorname{GH}}^{(n)}(r_t(T, \mathbf{v}), r_t(T', \mathbf{v})) \leq 4 d_{\operatorname{GH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}'))$ . This gives the result.

A rooted *n*-pointed tree  $(T, d, \mathbf{v})$  is locally compact if  $r_t(T, \mathbf{v})$  is a compact rooted tree for all  $t \ge 0$ . Following [6], we set for two locally compact rooted *n*-pointed trees  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$ :

$$d_{\mathrm{LGH}}^{(n)}((T,\mathbf{v}),(T',\mathbf{v}')) = \int_0^\infty \mathrm{e}^{-t} \,\mathrm{d}t \,\left(1 \wedge d_{\mathrm{GH}}^{(n)}\big(r_t(T,\mathbf{v}),r_t(T',\mathbf{v}')\big)\right).$$

Furthermore, we have that  $d_{\text{LGH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')) = 0$  if and only if there exists an isometric one-to-one map  $\varphi$  from (T, d) to (T', d') which preserves the distinguished vertices. The relation  $d_{\text{LGH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')) = 0$  defines an equivalence relation. The set  $\mathbb{T}_{\text{loc}-K}^{(n)}$  of equivalence classes of locally compact rooted trees endowed with  $d_{\text{LGH}}^{(n)}$  is then a metric Polish space. Furthermore,  $\mathbb{T}_{K}^{(n)}$  is an open dense subset of  $\mathbb{T}_{\text{loc}-K}^{(n)}$ . For n = 0, we simply write  $\mathbb{T}_{\text{loc}-K}$  and  $d_{\text{LGH}}$  for  $\mathbb{T}_{\text{loc}-K}^{(n)}$ and  $d_{\text{LGH}}^{(n)}$ . We provide a short proof for the following inequalities.

Lemma 5.4 (Inequalities for  $d_{\text{GH}}^{(n)}$  and  $d_{\text{LGH}}^{(n)}$ ). Let  $n \in \mathbb{N}$ . For  $(T, \mathbf{v}), (T', \mathbf{v}') \in \mathbb{T}_{\text{K}}^{(n)}$ , we have: (31)  $d_{\text{LGH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')) \leq 1 \wedge 4 d_{\text{GH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v})).$ 

For  $(T, \mathbf{v}), (T', \mathbf{v}') \in \mathbb{T}^{(n)}_{loc-K}$  and  $s, t \ge 0$ , we have:

(32) 
$$d_{\text{LGH}}^{(n)}(r_t(T, \mathbf{v}), r_{t+s}(T', \mathbf{v}')) \le 4 d_{\text{LGH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')) + s$$

(33) 
$$d_{\rm GH}^{(n)}(r_t(T, \mathbf{v}), r_t(T', \mathbf{v}')) \le 4 \,\mathrm{e}^t \, d_{\rm LGH}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')).$$

The map  $(t, (T, \mathbf{v})) \mapsto r_t(T, \mathbf{v})$  is continuous from  $\mathbb{R}_+ \times \mathbb{T}^{(n)}_{\text{loc}-K}$  to  $\mathbb{T}^{(n)}_{\text{loc}-K}$  (and to  $\mathbb{T}^{(n)}_K$ ).

Proof. Equation (31) is a direct consequence of (30) with s = 0 and the definition of  $d_{\text{LGH}}^{(n)}$ . Equation (32) follows from similar arguments, using also that  $r_{t'} \circ r_u = r_u \circ r_{t'} = r_{t' \wedge u}$ . For  $t \leq s$ , we have  $4^{-1} d_{\text{GH}}^{(n)} (r_t(T, \mathbf{v}), r_t(T', \mathbf{v}')) \leq d_{\text{GH}}^{(n)} (r_s(T, \mathbf{v}), r_s(T', \mathbf{v}'))$ . Integrating with respect to  $e^{-s}$  ds gives (33). The continuity of the map  $(t, (T, \mathbf{v})) \mapsto r_t(T)$  is a direct consequence of (32).

We deduce from (31) and (33) that all the measurable sets of  $(\mathbb{T}_{\mathrm{K}}^{(n)}, d_{\mathrm{GH}}^{(n)})$  are measurable sets of  $(\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)}, d_{\mathrm{LGH}}^{(n)})$ , and that a converging sequence in  $(\mathbb{T}_{\mathrm{K}}^{(n)}, d_{\mathrm{GH}}^{(n)})$  is also converging in  $(\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)}, d_{\mathrm{LGH}}^{(n)})$ . We also we deduce from (31) that the restriction to  $\mathbb{T}_{\mathrm{K}}^{(n)}$  of a continuous function defined on  $(\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)}, d_{\mathrm{LGH}}^{(n)})$  is also continuous on  $(\mathbb{T}_{\mathrm{K}}^{(n)}, d_{\mathrm{GH}}^{(n)})$ .

Removing from **v** some of the pointed vertices (but the root) is continuous, see the next lemma. For  $(T, \mathbf{v} = (v_0 = \varrho, \dots, v_n)) \in \mathbb{T}_{loc-K}^{(n)}$  and  $0 \in A \subset \{0, \dots, n\}$ , we set:

(34) 
$$\Pi_n^{\circ,A}(T,\mathbf{v}) = (T,\mathbf{v}_A) \quad \text{with} \quad \mathbf{v}_A = (v_i, i \in A).$$

For simplicity, we shall write  $\Pi_n^{\circ}$  for  $\Pi_n^{\circ,A}$  when A is reduced to  $\{0\}$ , so that  $\Pi_n^{\circ}$  corresponds to removing all the pointed vertices but the root.

**Lemma 5.5** (Removing some pointed vertices is continuous). Let  $n \in \mathbb{N}$  and  $0 \in A \subset \{0, \ldots, n\}$ . The map  $\prod_{n=1}^{\circ,A}$  from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(k)}$ , with k the cardinal of A, is 1-Lipschitz continuous.

*Proof.* First, notice that the equivalence class of  $(T, \mathbf{v}_A)$  in  $\mathbb{T}^{(k)}_{\text{loc}-K}$  does not depend of the choice of  $(T, \mathbf{v})$  in its equivalence class in  $\mathbb{T}^{(n)}_{\text{loc}-K}$ . Thus the map  $\Pi^{\circ,A}_n$  is well defined from  $\mathbb{T}^{(n)}_{\text{loc}-K}$  to  $\mathbb{T}^{(k)}_{\text{loc}-K}$ . It is clearly 1-Lipschitz continuous since a correspondence between the trees  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  is also a correspondence between  $(T, \mathbf{v}_A)$  and  $(T, \mathbf{v}'_A)$ .

We give an immediate consequence on the continuity of the maps Span and Span<sup>°</sup>.

**Lemma 5.6** (Continuity of the maps Span and Span<sup>°</sup>). Let  $n \in \mathbb{N}$ . The map  $(T, \mathbf{v}) \mapsto \text{Span}(T, \mathbf{v})$ and  $(T, \mathbf{v}) \mapsto \text{Span}^{\circ}(T, \mathbf{v})$  are 4-Lipschitz continuous from  $\mathbb{T}_{\text{loc}-K}^{(n)}$  to  $\mathbb{T}_{\text{loc}-K}^{(n)}$  and to  $\mathbb{T}_{\text{loc}-K}$  respectively. *Proof.* Notice that  $d_{\text{LGH}}^{(n)}(\text{Span}(T, \mathbf{v}), \text{Span}(T', \mathbf{v})) = d_{\text{GH}}^{(n)}(\text{Span}(T, \mathbf{v}), \text{Span}(T', \mathbf{v}))$ , and thus the map Span from  $\mathbb{T}_{\text{loc}-\text{K}}^{(n)}$  to  $\mathbb{T}_{\text{loc}-\text{K}}^{(n)}$  is 4-Lipschitz continuous, thanks to Lemma 5.2. Then use Lemma 5.5 on the continuity of  $\Pi_n^\circ$  and the fact that  $\text{Span}^\circ = \Pi_n^\circ \circ \text{Span}$  to conclude.

Next, we check that rerooting or reordering the pointed vertices is a continuous operation. For a vector  $\mathbf{v} = (v_0, \ldots, v_n)$  and a permutation  $\pi$  of  $\{0, \ldots, n\}$ , we set  $\mathbf{v}^{\pi} = (v_{\pi(0)}, \ldots, v_{\pi(n)})$ .

Remark 5.7. One can see that the map  $(T, \mathbf{v}) \mapsto (T, \mathbf{v}^{\pi})$  is an isometry on  $\mathbb{T}_{\mathrm{K}}^{(n)}$ . The next lemma is an extension to locally compact case.

**Lemma 5.8** (Permuting the pointed vertices is continuous). Let  $n \in \mathbb{N}$  and let  $\pi$  be a permutation on  $\{0, \ldots, n\}$ . The map  $(T, \mathbf{v}) \mapsto (T, \mathbf{v}^{\pi})$  defined on  $\mathbb{T}^{(n)}_{\text{loc}-K}$  is continuous.

*Proof.* First notice that if  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  are rooted *n*-pointed trees belonging to the same equivalence class of  $\mathbb{T}_{loc-K}^{(n)}$ , so do  $(T, \mathbf{v}^{\pi})$  and  $(T', \mathbf{v}'^{\pi})$ . Thus, the map  $(T, \mathbf{v}) \mapsto (T, \mathbf{v}^{\pi})$  is indeed well-defined on  $\mathbb{T}_{loc-K}^{(n)}$ . We shall use the following notation: we denote by  $r_t^{\circ}$  the truncation  $r_t$  when one forgets about the pointed vertices (but the root):  $r_t^{\circ} = \prod_n^{\circ} \circ r_t$ . (Take care that  $\prod_n^{\circ} \circ r_t \neq r_t \circ \prod_n^{\circ}$ .) To prove the continuity of the map, we consider two cases.

1st case: No rerooting,  $\pi(0) = 0$ . In that case, for every  $t \ge 0$  and every  $(T, \mathbf{v}) \in \mathbb{T}^{(n)}_{\text{loc}-\text{K}}$ , we have that  $r_t^{\circ}(T, \mathbf{v}) = r_t^{\circ}(T, \mathbf{v}^{\pi})$  and thus we get that:

$$d_{\mathrm{LGH}}^{(n)}\big((T,\mathbf{v}^{\pi}),(T',\mathbf{v}'^{\pi})\big) = d_{\mathrm{LGH}}^{(n)}\big((T,\mathbf{v}),(T',\mathbf{v}')\big).$$

This trivially implies the continuity of the map.

2nd case: With rerooting,  $\pi(k_0) = 0$  for some  $k_0 \neq 0$ . Let  $(T, \mathbf{v}), (T', \mathbf{v}') \in \mathbb{T}^{(n)}_{\text{loc}-K}$ , with  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$  and  $\mathbf{v}' = (v'_0 = \varrho', \dots, v'_n)$ , such that  $d^{(n)}_{\text{LGH}}((T, \mathbf{v}), (T', \mathbf{v}')) < 1/2$ . As  $v_{k_0}$  and  $v'_{k_0}$  are always in correspondence as well as  $\varrho$  and  $\varrho'$ , we have, for every  $t \geq 0$  that:

$$|H(v_{k_0}) - H(v'_{k_0})| \le 2d_{\mathrm{GH}}^{(n)} (r_t(T, \mathbf{v}), r_t(T', \mathbf{v}')).$$

Multiplying by  $e^{-t}$  and integrating yields:

$$1 \wedge |H(v_{k_0}) - H(v'_{k_0})| \le 2d_{\text{LGH}}^{(n)}((T, \mathbf{v}), (T', \mathbf{v}')) < 1,$$

and hence:

$$H(v'_{k_0}) \le H(v_{k_0}) + 1.$$

We set  $h_0 = H(v_{k_0}) + 1$ . Then, for every  $t \ge 0$ , we have:

$$r_t^{\circ}(T, \mathbf{v}^{\pi}) \subset r_{t+h_0}^{\circ}(T, \mathbf{v})$$
 and thus  $r_t(T, \mathbf{v}^{\pi}) = r_t \Big( r_{t+h_0}^{\circ}(T, \mathbf{v}), \mathbf{v}^{\pi} \Big),$ 

and the same holds for T'. Consequently, applying Lemma 5.3, we have:

$$d_{\rm LGH}^{(n)}((T, \mathbf{v}^{\pi}), (T', \mathbf{v}'^{\pi})) \leq 4 \int_{0}^{+\infty} dt \, e^{-t} \left( 1 \wedge d_{\rm GH}^{(n)} \left( \left( r_{t+h_0}^{\circ}(T, \mathbf{v}), \mathbf{v}^{\pi} \right), \left( r_{t+h_0}^{\circ}(T', \mathbf{v}'), \mathbf{v}'^{\pi} \right) \right) \right) \\ = 4 \int_{0}^{+\infty} dt \, e^{-t} \left( 1 \wedge d_{\rm GH}^{(n)} \left( r_{t+h_0}(T, \mathbf{v}), r_{t+h_0}(T', \mathbf{v}') \right) \right) \\ \leq 4 \, e^{h_0} \, d_{\rm LGH}^{(n)} \left( (T, \mathbf{v}), (T', \mathbf{v}') \right),$$

where we used for the second inequality that  $d_{\text{GH}}^{(n)}((\tilde{T}, \mathbf{v}^{\pi}), (\tilde{T}', \mathbf{v}'^{\pi})) = d_{\text{GH}}^{(n)}((\tilde{T}, \mathbf{v}), (\tilde{T}, \mathbf{v}'))$  for  $(\tilde{T}, \mathbf{v}), (\tilde{T}', \mathbf{v}') \in \mathbb{T}_{\mathrm{K}}^{(n)}$ . The continuity of the map follows.

We shall also consider the set of trees whose root is not a branching vertex:

(35) 
$$\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n),0} = \big\{ (T, \mathbf{v}) \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)} : \varrho \notin \mathrm{Br}(T) \big\}.$$

We shall simply write  $\mathbb{T}^0_{\text{loc}-K}$  for  $\mathbb{T}^{(n),0}_{\text{loc}-K}$  when n = 0.

**Lemma 5.9.** The set  $\mathbb{T}_{loc-K}^{(n),0}$  is a Borel subset of  $\mathbb{T}_{loc-K}^{(n)}$ .

*Proof.* For a rooted tree T, we define its diameter by diam  $(T) = \sup\{d(x, y) : x, y \in T\}$ . Notice that  $H(T) \leq \operatorname{diam}(T) \leq 2H(T)$ . Clearly the function diam is constant on all equivalent classes of  $\mathbb{T}_{\mathrm{K}}^{(n)}$  and thus of  $\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)}$ . If diam  $(T) = 2H(T) < +\infty$ , then we deduce that the root is a branching vertex. Recall  $\prod_{n}^{\circ}$  for (34). More generally, we get that:

$$\mathbb{T}_{\text{loc}-K}^{(n),0} = \bigcup_{n \in \mathbb{N}^*} D_{1/n} \quad \text{with} \quad D_t = \Big\{ T \in \mathbb{T}_{\text{loc}-K}^{(n)} : \text{ diam } \big( r_t \circ \Pi_n^\circ(T) \big) = 2t \Big\}.$$

Since the functions diam,  $r_t$  and  $\Pi_n^{\circ}$  are continuous, we deduce that  $D_t$  is closed, and hence  $\mathbb{T}_{loc-K}^{(n),0}$  is a Borel subset of  $\mathbb{T}_{loc-K}$ .

We now define the set of discrete trees. We say that a rooted *n*-pointed tree  $(T, d, \mathbf{v})$  is a discrete tree if T is equal to the tree spanned by its distinguished vertices:  $T = \text{Span}^{\circ}(T, \mathbf{v})$ . We define the set of (equivalence classes of) discrete trees with at most n leaves as:

(36) 
$$\mathbb{T}_{\mathrm{dis}}^{(n)} = \left\{ (T, \mathbf{v}) \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)} : (T, \mathbf{v}) = \mathrm{Span}(T, \mathbf{v}) \right\}.$$

As a direct consequence of the continuity of the map Span we get the following result.

**Lemma 5.10.** Let  $n \in \mathbb{N}$ . The set of discrete trees  $\mathbb{T}_{dis}^{(n)}$  is a closed subset of  $\mathbb{T}_{K}^{(n)}$  and of  $\mathbb{T}_{loc-K}^{(n)}$ .

We end this section with partial measurability result on the number of vertices at a given height of a tree.

Remark 5.11. It is immediate to check that the map  $(T, \mathbf{v}) \mapsto (d(v_i, v_j), i, j \in \{0, \dots, n\})$  is injective 1/2-Lipschitz continuous from  $(\mathbb{T}_{loc-K}^{(n)}, d_{LGH}^{(n)})$  to  $\mathbb{R}^{(n+1)\times(n+1)}$  endowed with the supremum norm (*i.e.* the maximum of the distances between coordinates). It is also bi-measurable thanks to Lusin's theorem from [32] or [11, Exercise 6.10.54 p.60].

Let  $\mathbb{T}_{loc-K}^{no \, leaf}$  be the set of trees with no leaves:

$$\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{\mathrm{no\,leaf}} = \big\{ T \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}} : \ \mathrm{Lf}(T) = \emptyset \big\}.$$

For  $T \in \mathbb{T}_{loc-K}^{no \, leaf}$  and  $t \ge 0$ , let  $\tilde{N}_t(T)$  denotes the finite number of vertices at height t of T:

(37) 
$$\tilde{N}_t(T) = \operatorname{Card}\left(\left\{x \in T : \ H(x) = t\right\}\right)$$

We have the following result.

**Lemma 5.12** (Measurability of  $\tilde{N}_t$ ). The set  $\mathbb{T}_{loc-K}^{no \, leaf}$  is a Borel subset of  $\mathbb{T}_{loc-K}$  and the map  $(t,T) \mapsto \tilde{N}_t(T)$  is measurable from  $\mathbb{R}_+ \times \mathbb{T}_{loc-K}^{no \, leaf}$  to  $\mathbb{N}$ .

*Proof.* Let  $t \ge 0$  and let  $\Theta_n(t)$  be the set of discrete trees such that all the pointed vertices (but the root) are leaves at height t:

$$\Theta_n(t) = \left\{ T \in \mathbb{T}^{(n)}_{\text{dis}} : d(\varrho, v_i) = t \text{ and } d(v_i, v_j) > 0 \text{ for all } i, j \in \{1, \dots, n\} \right\}.$$

Thanks to Remark 5.11,  $\Theta_n(t)$  is a Borel set of  $\mathbb{T}_{dis}^{(n)} \subset \mathbb{T}_K^{(n)} \subset \mathbb{T}_{loc-K}^{(n)}$ . For  $T \in \mathbb{T}_{loc-K}$ , we get that  $\{T' \in \mathbb{T}_{dis}^{(n)} : \Pi_n^{\circ}(T') = T\}$  is finite. We deduce from Lusin's theorem, see [32], that

 $\Pi_n^{\circ}$  restricted to  $\mathbb{T}_{\text{dis}}^{(n)}$  is bi-measurable. This implies that the set  $\Pi_n^{\circ}(\Theta_n(t))$  is a Borel subset of  $\mathbb{T}_{\text{loc}-K}$ . We deduce that the set of trees with no leaves,  $\mathbb{T}_{\text{loc}-K}^{\text{no leaf}}$ , which is formally defined by:

$$\mathbb{T}_{\rm loc-K}^{\rm no \, leaf} = \bigcap_{k \in \mathbb{N}^*} \bigcup_{n \in \mathbb{N}} r_k^{-1} \Big( \Pi_n^{\circ} \big( \Theta_n(k) \big) \Big),$$

is a Borel subset of  $\mathbb{T}_{loc-K}$ . We also get that  $\{T \in \mathbb{T}_{loc-K}^{no \, leaf} : \tilde{N}_t(T) = n\} = r_t^{-1} (\Pi_n^{\circ}(\Theta_n(t)));$  this implies that the map  $\tilde{N}_t$  is measurable. Since  $t \mapsto \tilde{N}_t(T)$  is non-decreasing and left-continuous, we deduce that the map  $(t,T) \mapsto \tilde{N}_t(T)$  is measurable from  $\mathbb{R}_+ \times \mathbb{T}_{loc-K}^{no \, leaf}$  to  $\mathbb{N}$ .  $\Box$ 

5.4. Grafting a tree on another one. We start by recalling the grafting operation of [2] which is slightly different from the function Graft which will be introduced in Section 5.9.

Let  $n, k \in \mathbb{N}$  and  $i \in \{0, \ldots, n\}$  be given. Let  $(T, d, \mathbf{v})$  be a locally compact rooted *n*-pointed tree with  $\mathbf{v} = (v_0 = \varrho, \ldots, v_n)$  and  $(T', d', \mathbf{v}')$  be a locally compact rooted *k*-pointed tree with  $\mathbf{v}' = (v'_0 = \varrho', \ldots, v'_k)$ . We define the tree  $T \circledast_i T'$  as the tree obtained by grafting T' on the *i*-th distinguished vertex of the tree T. We set  $\mathbf{v} \circledast \mathbf{v}'$  the concatenation of the vector  $\mathbf{v}$  and the vector  $(v'_1, \ldots, v'_k)$  which is the vector  $\mathbf{v}'$  where the coordinate  $v'_0 = \varrho'$  is removed, and:

$$T \circledast_i T' = T \sqcup (T' \setminus \{\varrho'\}),$$
  
$$\forall x, x' \in T \circledast_i T', \ d^{\circledast}(x, x') = \begin{cases} d(x, x') & \text{if } x, x' \in T, \\ d'(x, x') & \text{if } x, x' \in T', \\ d(x, v_i) + d'(\varrho', x') & \text{if } x \in T, \ x' \in T', \end{cases}$$

where  $\sqcup$  denotes the disjoint union of two sets. By construction  $(T \circledast_i T', d^{\circledast}, \mathbf{v} \circledast \mathbf{v}')$  is a locally compact rooted n + k pointed tree. It is easy to see that the equivalence class of  $T \circledast_i T'$  in  $\mathbb{T}_{\text{loc}-K}^{(n+k)}$ does not depend of the choice of the representatives in the equivalence classes of T and T' and hence the map  $((T, d, \mathbf{v}), (T', d', \mathbf{v}')) \mapsto (T \circledast_i T', d^{\circledast}, \mathbf{v} \circledast \mathbf{v}')$  is well defined from  $\mathbb{T}_{\text{loc}-K}^{(n)} \times \mathbb{T}_{\text{loc}-K}^{(k)}$ into  $\mathbb{T}_{\text{loc}-K}^{(n+k)}$ . We shall simply write  $(T \circledast_i T', \mathbf{v} \circledast \mathbf{v}')$ , or even  $T \circledast_i T'$ , for  $(T \circledast_i T', d^{\circledast}, \mathbf{v} \circledast \mathbf{v}')$ . The next lemma asserts that this grafting procedure is continuous.

**Lemma 5.13** (Continuity of the grafting map). Let  $n, k \in \mathbb{N}$  and  $i \in \{0, \ldots, n\}$ . The map  $((T, \mathbf{v}), (T', \mathbf{v}')) \mapsto (T \circledast_i T', \mathbf{v} \circledast \mathbf{v}')$ , is continuous from  $\mathbb{T}_{loc-K}^{(n)} \times \mathbb{T}_{loc-K}^{(k)}$  to  $\mathbb{T}_{loc-K}^{(n+k)}$ .

*Proof.* Let  $(T_1, \mathbf{v}_1), (T'_1, \mathbf{v}'_1) \in \mathbb{T}^{(n)}_{\text{loc}-K}$  and  $(T_2, \mathbf{v}_2), (T'_2, \mathbf{v}'_2) \in \mathbb{T}^{(k)}_{\text{loc}-K}$ . Set  $T = T_1 \circledast_i T_2, T' = T'_1 \circledast_i T'_2, \mathbf{v} = \mathbf{v}_1 \circledast \mathbf{v}_2$ , and  $\mathbf{v}' = \mathbf{v}'_1 \circledast \mathbf{v}'_2$ .

First suppose that the trees are compact, that is  $(T_1, \mathbf{v}_1), (T'_1, \mathbf{v}'_1) \in \mathbb{T}_{\mathrm{K}}^{(n)}$  and  $(T_2, \mathbf{v}_2), (T'_2, \mathbf{v}'_2) \in \mathbb{T}_{\mathrm{K}}^{(k)}$ . Let  $\mathcal{R}_1$  be a correspondence between (elements of the classes)  $(T_1, \mathbf{v}_1)$  and  $(T'_1, \mathbf{v}'_1)$  and let  $\mathcal{R}_2$  be a correspondence between (elements of the classes)  $(T_2, \mathbf{v}_2)$  and  $(T'_2, \mathbf{v}'_2)$ . We set  $\mathcal{R} = \mathcal{R}_1 \cup \mathcal{R}_2$  with  $\rho_2$  and  $\rho'_2$  replaced respectively by  $v_i$  and  $v'_i$ . It defines a correspondence between  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$ . For every  $(x, x'), (y, y') \in \mathcal{R}$ , we have:

$$\left| d^{\circledast}(x,y) - d'^{\circledast}(x',y') \right| = \begin{cases} \left| d_1(x,y) - d'_1(x',y') \right| \le \operatorname{dist} \left(\mathcal{R}_1\right) & \text{if } (x,x'), (y,y') \in \mathcal{R}_1, \\ \left| d_2(x,y) - d'_2(x',y') \right| \le \operatorname{dist} \left(\mathcal{R}_2\right) & \text{if } (x,x'), (y,y') \in \mathcal{R}_2, \end{cases}$$

and if  $(x, x') \in \mathcal{R}_1$  and  $(y, y') \in \mathcal{R}_2$ , we have:

$$\begin{aligned} \left| d^{\circledast}(x,y) - d'^{\circledast}(x',y') \right| &= \left| d_1(x,v_i) + d_2(\rho_2,y) - d'_2(\rho'_2,y') - d'_1(x',v'_i) \right| \\ &\leq \left| d_1(x,v_i) - d_1(x',v'_i) \right| + \left| d_2(\rho_2,y) - d'_2(\rho'_2,y') \right| \\ &\leq \text{dist} \left( \mathcal{R}_1 \right) + \text{dist} \left( \mathcal{R}_2 \right). \end{aligned}$$

This gives:

(38) 
$$d_{\rm GH}^{(n+k)}((T,\mathbf{v}),(T',\mathbf{v}')) \leq d_{\rm GH}^{(n)}((T_1,\mathbf{v}_1),(T_1',\mathbf{v}_1')) + d_{\rm GH}^{(k)}((T_2,\mathbf{v}_2),(T_2',\mathbf{v}_2')).$$

Now consider  $(T_1, \mathbf{v}_1), (T'_1, \mathbf{v}'_1) \in \mathbb{T}^{(n)}_{\text{loc}-K}$  and  $(T_2, \mathbf{v}_2), (T'_2, \mathbf{v}'_2) \in \mathbb{T}^{(k)}_{\text{loc}-K}$ . Without loss of generality we assume that  $H(v'_i) \geq H(v_i)$ . Remark that, for every  $t \geq 0$ , we have, with  $a_+ = \max(a, 0)$ :

$$r_t(T, \mathbf{v}) = r_t(T_1, \mathbf{v}_1) \circledast_i r_{(t-H(v_i))_+}(T_2, \mathbf{v}_2).$$

Therefore, we have:

$$\begin{split} d_{\mathrm{LGH}}^{(n+k)} \left( (T,\mathbf{v}), (T',\mathbf{v}') \right) \\ &= \int_{0}^{+\infty} \mathrm{d}t \, \mathrm{e}^{-t} \left( 1 \wedge d_{\mathrm{GH}}^{(n+k)} \left( r_{t}(T,\mathbf{v}), r_{t}(T',\mathbf{v}') \right) \right) \\ &= \int_{0}^{+\infty} \mathrm{d}t \, \mathrm{e}^{-t} \left( 1 \wedge d_{\mathrm{GH}}^{(n+k)} \left( r_{t}(T_{1},\mathbf{v}_{1}) \circledast_{i} r_{(t-H(v_{i}))_{+}}(T_{2},\mathbf{v}_{2}), r_{t}(T'_{1},\mathbf{v}'_{1}) \circledast_{i} r_{(t-H(v'_{i}))_{+}}(T'_{2},\mathbf{v}'_{2}) \right) \right) \\ &\leq \int_{0}^{+\infty} \mathrm{d}t \, \mathrm{e}^{-t} \left( 1 \wedge d_{\mathrm{GH}}^{(n)} \left( r_{t}(T_{1},\mathbf{v}_{1}), r_{t}(T'_{1},\mathbf{v}'_{1}) \right) \right) \\ &\qquad + \int_{0}^{+\infty} \mathrm{d}t \, \mathrm{e}^{-t} \left( 1 \wedge d_{\mathrm{GH}}^{(k)} \left( r_{(t-H(v_{i}))_{+}}(T_{2},\mathbf{v}_{2}), r_{(t-H(v'_{i}))_{+}}(T'_{2},\mathbf{v}'_{2}) \right) \right) \\ &\leq d_{\mathrm{LGH}}^{(n)} \left( (T_{1},\mathbf{v}_{1}), (T'_{1},\mathbf{v}'_{1}) \right) + 4 \, \mathrm{e}^{-H(v'_{i})} \, d_{\mathrm{LGH}}^{(k)} \left( (T_{2},\mathbf{v}_{2}), (T'_{2},\mathbf{v}'_{2}) \right) + H(v'_{i}) - H(v_{i}) \\ &\leq 3 \, d_{\mathrm{LGH}}^{(n)} \left( (T_{1},\mathbf{v}_{1}), (T'_{1},\mathbf{v}'_{1}) \right) + 4 d_{\mathrm{LGH}}^{(k)} \left( (T_{2},\mathbf{v}_{2}), (T'_{2},\mathbf{v}'_{2}) \right), \end{split}$$

where we used Equation (38) for the first inequality and Lemma 5.3 for the second one. We are done.  $\hfill \Box$ 

Remark 5.14. We shall use a version of the grafting procedure where, instead of grafting on  $v_i$ , we shall graft on the branch  $[\![\varrho, v_i]\!]$  at height h provided that  $H(v_i) \ge h$ .

Let  $n \in \mathbb{N}$  and  $i \in \{0, ..., n\}$  be given. For  $h \in \mathbb{R}_+$  and  $(T, \mathbf{v}) \in \mathbb{T}_{\mathrm{K}}^{(n)}$ , we denote by  $x_{i,h}$ the unique vertex of T that satisfies  $x_{i,h} \in [\![\varrho, v_i]\!]$  and  $H(x_{i,h}) = H(v_i) \wedge h$ . Then, the map  $(h, (T, \mathbf{v})) \mapsto (T, (\mathbf{v}, x_{i,h}))$  is clearly continuous from  $\mathbb{R}_+ \times \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n)}$  to  $\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n+1)}$ . We then define the grafting map  $\circledast_{i,h}$  by:

(39) 
$$(h, (T, \mathbf{v}), (T', \mathbf{v}')) \mapsto T \circledast_{i,h} T' = (T \circledast_{i,h} T', \mathbf{v} \circledast \mathbf{v}'),$$

as the composition of

[adding the vertex  $x_{i,h}$ ]:  $(h, (T, \mathbf{v})) \mapsto (T, \tilde{\mathbf{v}})$  with  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$  and  $\tilde{\mathbf{v}} = (\mathbf{v}, x_{i,h}) = (\tilde{v}_0 = \varrho, \dots, \tilde{v}_n = v_n, \tilde{v}_{n+1} = x_{i,h}),$ 

[grafting]: 
$$((T, \tilde{\mathbf{v}}), (T', \mathbf{v}')) \mapsto (T \circledast_{n+1} T', \tilde{\mathbf{v}} \circledast \mathbf{v}')$$
 and  
[removing the  $(n+1)$ -th pointed vertex]:  $(T'' = T \circledast_{n+1} T', \tilde{\mathbf{v}} \circledast \mathbf{v}') \mapsto (T'', \mathbf{v} \circledast \mathbf{v}')$ .

Since all those maps are continuous, we get that the map  $\circledast_{i,h}$  is continuous (and hence measurable) from  $\mathbb{R}_+ \times \mathbb{T}^{(n)}_{loc-K} \times \mathbb{T}^{(k)}_{loc-K}$  to  $\mathbb{T}^{(n+k)}_{loc-K}$ .

We shall also be interested in a grafting on the left or on the right of  $i \in \{1, \ldots, n\}$ , which is the same as the grafting (39), but for the order of the coordinates of the vector  $\mathbf{v} \circledast \mathbf{v}'$ . For  $\epsilon \in \{g, d\}$ , we define the grafting map  $\circledast_{i,h}^{\epsilon}$  by:

(40) 
$$(h, (T, \mathbf{v}), (T', \mathbf{v}')) \mapsto T \circledast_{i,h}^{\epsilon} T' = (T \circledast_{i,h} T', \mathbf{v}^{\epsilon,i}),$$

where, with  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$  and  $\mathbf{v}' = (v'_0 = \varrho, \dots, v'_k)$ , we set:

$$\mathbf{v}^{g,i} = (v_0 = \varrho, \dots, v_{i-1}, v'_1, \dots, v'_k, v_i, \dots, v_n)$$
$$\mathbf{v}^{d,i} = (v_0 = \varrho, \dots, v_i, v'_1, \dots, v'_k, v_{i+1}, \dots, v_n).$$

Thanks to Lemma 5.8, we deduce from the continuity of the map  $\circledast_{i,h}$ , that the maps  $\circledast_{i,h}^{\epsilon}$  are continuous.

We summarize the results from Remark 5.14 in the following lemma.

**Lemma 5.15** (Continuity of the grafting maps). Let  $n, k \in \mathbb{N}$ ,  $i \in \{0, ..., n\}$  and  $\epsilon \in \{g, d\}$ . The maps  $(h, (T, \mathbf{v}), (T', \mathbf{v}')) \mapsto T \circledast_{i,h} T'$  and  $(h, (T, \mathbf{v}), (T', \mathbf{v}')) \mapsto T \circledast_{i,h}^{\epsilon} T'$ , are continuous from  $\mathbb{R}_+ \times \mathbb{T}_{loc-K}^{(n)} \times \mathbb{T}_{loc-K}^{(k)}$  to  $\mathbb{T}_{loc-K}^{(n+k)}$ .

5.5. Set of (equivalence classes of) marked trees. We shall consider trees with a marked infinite branch; for this reason we introduce the notion of marked trees. In this part, we do not record an order on the marked vertices as in the n-pointed trees.

We say that  $(T, S, d, \varrho)$  is a marked rooted tree if  $(T, d, \varrho)$  is a rooted tree and the set of marks S is a sub-tree of T with the same root (that is  $\varrho \in S$ ) endowed with the restriction of the distance d. A correspondence between two compact marked rooted trees  $(T, S, d, \varrho)$  and  $(T', S', d', \varrho')$  is a set  $\mathcal{R} \subset T \times T'$  such that  $\mathcal{R}$  is a correspondence between  $(T, d, \varrho)$  and  $(T', d', \varrho')$  and  $\mathcal{R} \cap (S \times S')$  is also a correspondence between  $(S, d, \varrho)$  and  $(S', d', \varrho')$ . Then, we set:

$$d_{\mathrm{GH}}^{[2]}((T,S),(T',S')) = \inf \frac{1}{2} \operatorname{dist} (\mathcal{R}),$$

where the infimum is taken over all the correspondences  $\mathcal{R}$  between  $(T, S, d, \varrho)$  and  $(T', d', S', \varrho')$ . An easy extension of [6] gives that  $d_{\mathrm{GH}}^{[2]}$  is a pseudo-distance, and that  $d_{\mathrm{GH}}^{[2]}(T,T') = 0$  if and only if there exists an isometric one-to-one map  $\varphi$  from (T, d) to (T', d') which preserves the root and which is also one-to-one from S to S'. The relation  $d_{\mathrm{GH}}^{[2]}((T, S), (T', S')) = 0$  defines an equivalence relation. The set  $\mathbb{T}_{\mathrm{K}}^{[2]}$  of equivalence classes of compact marked rooted trees  $(T, S, d, \varrho)$  endowed with  $d_{\mathrm{GH}}^{[2]}$  is then a metric Polish space. We simply write (T, S) for  $(T, S, d, \varrho)$ , and unless specified otherwise, we shall denote also by (T, S) its equivalence class. Since  $d_{\mathrm{GH}}^{[2]}((T, S), (T', S')) \ge d_{\mathrm{GH}}(T, T') \lor d_{\mathrm{GH}}(S, S')$ , we deduce that the map  $(T, S) \mapsto (T, S)$ from  $\mathbb{T}_{\mathrm{K}}^{[2]}$  to  $(\mathbb{T}_{\mathrm{K}})^2$  (endowed with the maximum distance on the coordinates) is continuous. For  $t \ge 0$ , we define the truncation function  $r_t^{[2]}$  of a marked rooted tree  $(T, S, d, \varrho)$  as the marked rooted tree  $r_t^{[2]}(T, S) = (r_t(T), r_t(S), d, \varrho)$ , where we recall that  $r_t(T) = \{x \in T : H(x) \le t\}$ . If (T, S) and (T', S') are in the same equivalence class of  $\mathbb{T}_{\mathrm{K}}^{[2]}$ , so are  $r_t^{[2]}(T, S)$  and  $r_t^{[2]}(T', S')$ ; thus the function  $r_t^{[2]}$  can be seen as a map from  $\mathbb{T}_{\mathrm{K}}^{[2]}$  to itself. Similarly to (30), we have for  $t, s \ge 0$ and  $(T, S), (T', S') \in \mathbb{T}_{\mathrm{K}}^{[2]}$ :

(41) 
$$d_{\text{GH}}^{[2]}\left(r_t^{[2]}(T,S), r_{t+s}^{[2]}(T',S')\right) \le 4 \, d_{\text{GH}}^{[2]}\left((T,S), (T',S')\right) + s.$$

22

This implies that the map  $(t, (T, S)) \mapsto r_t^{[2]}(T, S)$  is continuous from  $\mathbb{R}_+ \times \mathbb{T}_K^{[2]}$  to  $\mathbb{T}_K^{[2]}$ .

A marked rooted tree  $(T, S, d, \varrho)$  is locally compact if  $r_t^{[2]}(T, S)$  is a compact marked rooted tree for all  $t \ge 0$ . Following [6], we consider for two locally compact marked rooted trees (T, S) and (T', S'):

(42) 
$$d_{\text{LGH}}^{[2]}((T,S),(T',S')) = \int_0^\infty e^{-t} dt \left( 1 \wedge d_{\text{GH}}^{[2]}\left(r_t^{[2]}(T,S),r_t^{[2]}(T',S')\right) \right).$$

Furthermore, we have that  $d_{\text{LGH}}^{[2]}((T,S),(T',S')) = 0$  if and only if there exists an isometric oneto-one map  $\varphi$  from (T,d) to (T',d') which is one-to-one from S to S' and preserves the roots. Thus the relation  $d_{\text{LGH}}^{[2]}((T,S),(T',S')) = 0$  defines an equivalence relation, see [7, Proposition 5.3]. The set  $\mathbb{T}_{\text{loc}-K}^{[2]}$  of equivalence classes of locally compact marked rooted trees  $(T,S,d,\varrho)$  endowed with  $d_{\text{LGH}}^{[2]}$  is then a metric Polish space. Furthermore,  $\mathbb{T}_{K}^{[2]}$  is an open dense subset of  $\mathbb{T}_{\text{loc}-K}^{[2]}$ . Similar equations to (31), (32) and (33) holds with  $d_{\text{LGH}}^{(n)}$  and  $d_{\text{GH}}^{(n)}$  replaced by  $d_{\text{LGH}}^{[2]}$  and  $d_{\text{GH}}^{[2]}$ . For future use, let us give the equations corresponding to (32) and (33). For  $(T,S), (T',S') \in \mathbb{T}_{\text{loc}-K}^{[2]}$ and  $s, t \geq 0$ , we have:

(43) 
$$d_{\text{LGH}}^{[2]}\left(r_t^{[2]}(T,S), r_{t+s}^{[2]}(T',S')\right) \le 4 d_{\text{LGH}}^{[2]}\left((T,S), (T',S')\right) + s,$$

(44) 
$$d_{\rm GH}^{[2]}\left(r_t^{[2]}(T,S), r_t^{[2]}(T',S')\right) \le 4 \,\mathrm{e}^t \, d_{\rm LGH}^{[2]}\left((T,S), (T',S')\right).$$

We also we have the following result consequences of (41) and (43).

**Lemma 5.16** (Continuity of the truncation map). Let  $n \in \mathbb{N}$ . The map  $(t, (T, S)) \mapsto r_t^{[2]}(T, S)$ is continuous from  $\mathbb{R}_+ \times \mathbb{T}_{\mathrm{K}}^{[2]}$  to  $\mathbb{T}_{\mathrm{K}}^{[2]}$  and from  $\mathbb{R}_+ \times \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{[2]}$  to  $\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{[2]}$  (and to  $\mathbb{T}_{\mathrm{K}}^{[2]}$ ).

We give in the next lemma an example of a  $\mathbb{T}_{K}^{[2]}$  and  $\mathbb{T}_{loc-K}^{[2]}$  valued function.

**Lemma 5.17** (Continuity of Span<sup>°</sup>). Let  $n \in \mathbb{N}$ . The map  $(T, d, \mathbf{v}) \mapsto (\Pi_n^{\circ}(T), \operatorname{Span}^{\circ}(T, \mathbf{v}), d, \varrho)$ from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$  (resp. from  $\mathbb{T}_K^{(n)}$  to  $\mathbb{T}_K^{[2]}$ ) is injective, bi-measurable and 16-Lipschitz (resp. 4-Lipschitz) continuous.

*Proof.* We first consider the compact case. Let  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  be rooted *n*-pointed compact trees and let  $\mathcal{R}$  be a correspondence between them. Recall the definition of  $p_{\mathbf{v}}$  in (27) as the projection on  $\operatorname{Span}^{\circ}(T, \mathbf{v})$  and the correspondence  $\tilde{\mathcal{R}}$  from (28). We set  $\mathcal{R}^{[2]} = \mathcal{R} \cup \tilde{\mathcal{R}}$ . By construction  $\mathcal{R}^{[2]}$  is a correspondence between  $(T, \operatorname{Span}^{\circ}(T, \mathbf{v}))$  and  $(T', \operatorname{Span}^{\circ}(T', \mathbf{v}'))$ . From the proof of Lemma 5.2, we get that dist  $(\mathcal{R}^{[2]}) \leq 4 \operatorname{dist}(\mathcal{R})$ . This directly implies that:

(45) 
$$d_{\mathrm{GH}}^{[2]}\left(\left(T, \mathrm{Span}^{\circ}(T, \mathbf{v})\right), \left(T', \mathrm{Span}^{\circ}(T', \mathbf{v}')\right)\right) \leq 4 \, d_{\mathrm{GH}}^{(n)}\left((T, \mathbf{v}), (T', \mathbf{v}')\right).$$

This gives that the map  $(T, d, \mathbf{v}) \mapsto (T, \operatorname{Span}^{\circ}(T, \mathbf{v}), d, \varrho)$  from  $\mathbb{T}_{\mathrm{K}}^{(n)}$  to  $\mathbb{T}_{\mathrm{K}}^{[2]}$  is 4-Lipschitz continuous.

We now consider the locally compact case. Let  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  belong to  $\mathbb{T}_{loc-K}^{[2]}$ . We have:

$$\begin{aligned} d_{\mathrm{LGH}}^{[2]} \left( \left(T, \mathrm{Span}^{\circ}(T, \mathbf{v})\right), \left(T', \mathrm{Span}^{\circ}(T', \mathbf{v}')\right) \right) \\ &= \int_{0}^{\infty} \mathrm{e}^{-t} \, \mathrm{d}t \left(1 \wedge d_{\mathrm{GH}}^{[2]} \left(r_{t}^{[2]} \left((T, \mathrm{Span}^{\circ}(T, \mathbf{v})\right), r_{t}^{[2]} \left(T', \mathrm{Span}^{\circ}(T', \mathbf{v}')\right) \right) \right) \\ &\leq 4 \int_{0}^{\infty} \mathrm{e}^{-t} \, \mathrm{d}t \left(1 \wedge d_{\mathrm{GH}}^{[2]} \left(\left(r_{t}(T, \mathbf{v}), \mathrm{Span}^{\circ}(T, \mathbf{v})\right), \left(r_{t}(T', \mathbf{v}'), \mathrm{Span}^{\circ}(T', \mathbf{v}')\right) \right) \right) \\ &\leq 16 \int_{0}^{\infty} \mathrm{e}^{-t} \, \mathrm{d}t \left(1 \wedge d_{\mathrm{GH}}^{(n)} \left(r_{t}(T, \mathbf{v}), r_{t}(T', \mathbf{v}')\right) \right) \\ &= 16 \, d_{\mathrm{LGH}}^{(n)} \left((T, \mathbf{v}), (T', \mathbf{v}')\right), \end{aligned}$$

where we used (41) (with T and S replaced respectively by  $r_t(T, \mathbf{v})$  and Span<sup> $\circ$ </sup>(T,  $\mathbf{v}$ ) and similarly for T' and S') for the first inequality, and (45) (with  $(T, \mathbf{v})$  replaced by  $r_t(T, \mathbf{v})$ ) as well as the relation  $\operatorname{Span}^{\circ}(r_t(T, \mathbf{v})) = \operatorname{Span}^{\circ}(T, \mathbf{v})$  for the second. This gives that the map  $(T, d, \mathbf{v}) \mapsto$  $(T, \operatorname{Span}^{\circ}(T, \mathbf{v}), d, \varrho)$  from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$  is 16-Lipschitz continuous. Clearly those maps are injective and thus bi-measurable thanks to Lusin's theorem [32].

*Remark* 5.18. Let us stress that for  $(T, \mathbf{v})$  a rooted *n*-pointed compact tree, the rooted tree  $r_t^{[2]}(T, \operatorname{Span}^{\circ}(T, \mathbf{v})) = \left(r_t(T), r_t(\operatorname{Span}^{\circ}(T, \mathbf{v}))\right) \text{ and the rooted tree } \left(r_t(T), \operatorname{Span}^{\circ}(r_t(T, \mathbf{v}))\right) = r_t^{[2]}(T, \operatorname{Span}^{\circ}(T, \mathbf{v}))$  $(r_t(T), \operatorname{Span}^\circ(T, \mathbf{v}))$  differ if and only if t is smaller than the height of  $\operatorname{Span}^\circ(T, \mathbf{v})$ .

Let  $(T, S, d, \rho)$  be a marked locally compact rooted tree. To simplify, we shall only write (T, S)for  $(T, S, d, \varrho)$ . We define the projection of  $z \in T$  on  $S, p_S(z) \in S$ , as the element of S uniquely defined by:

$$\llbracket \varrho, p_S(z) \rrbracket = \llbracket \varrho, z \rrbracket \cap S$$

Now, we consider the truncation of a marked tree at a given height, say t, of the marked sub-tree. For  $t \ge 0$  and  $\varepsilon \in \{-,+\}$ , we set:

$$r_t^{[2],\varepsilon}(T,S) = \left(r_{t,1}^{[2],\varepsilon}(T,S), r_t(S)\right)$$

with:

$$r_{t,1}^{[2],+}(T,S) = \left\{ x \in T : H(p_S(x)) \le t \right\},\$$
$$r_{t,1}^{[2],-}(T,S) = \left\{ x \in T : H(p_S(x)) < t \right\} \cup \left\{ x \in S : H(x) = t \right\}$$

See Figure 1 for an instance of  $r_t^{[2],\varepsilon}(T,S)$ , where S is an infinite branch. For  $\varepsilon \in \{+,-\}$ , we also denote by  $r_t^{[2],\varepsilon}(T,S)$  the marked rooted tree  $(r_t^{[2],\varepsilon}(T,S), d, \varrho)$  endowed with the restriction of the distance d and the root  $\varrho$ . Furthermore, if (T,S) and (T',S') belong to the same equivalence class of  $\mathbb{T}_{\text{loc}-K}^{[2]}$  or  $\mathbb{T}_K^{[2]}$ , then so do  $r_t^{[2],\varepsilon}(T,S)$  and  $r_t^{[2],\varepsilon}(T',S')$ . Thus the map  $(t,(T,S)) \mapsto r_t^{[2],\varepsilon}(T,S)$  is a well defined map from  $\mathbb{R}_+ \times \mathbb{T}_{\text{loc}-K}^{[2]}$  to  $\mathbb{T}_{\text{loc}-K}^{[2]}$  for  $\varepsilon \in \{+,-\}$ .

24



FIGURE 1. Example of restrictions of a tree T with a marked spine S (in bold).

Remark 5.19 (Examples). We give elementary examples. For  $\varepsilon \in \{+, -\}$  and t > 0, we have that  $r_t^{[2],\varepsilon}(T, \{\varrho\}) = (T, \{\varrho\})$  and  $r_0^{[2],-}(T, \{\varrho\}) = (\{\varrho\}, \{\varrho\})$  as well as  $r_0^{[2],+}(T, \{\varrho\}) = (T, \{\varrho\})$ . We also have for  $t \in \mathbb{R}_+$  that  $r_t^{[2],\varepsilon}(T,T) = (r_t(T), r_t(T))$ .

Remark 5.20 (The map  $r_t^{[2],\varepsilon}$  is not continuous). Let  $\varepsilon \in \{+,-\}$  and t > 0. The function  $r_t^{[2],\varepsilon}$  is not continuous from  $\mathbb{T}_{loc-K}^{[2]}$  to itself. Indeed take t = 1 without loss of generality and consider T = [0,2] and  $S_{\delta} = [0,\delta]$ , with  $\delta \in [0,2]$ ,  $\varrho = 0$  and the Euclidean distance. Notice that  $([0,1], [0,1]) = (S_1, S_1) \neq (T, S_1)$ . Then we have that  $\lim_{\delta \to 1} d_{\mathrm{GH}}^{[2]}((T, S_{\delta}), (T, S_1)) = 0$ ,  $r_1^{[2],\varepsilon}(T, S_{\delta}) = (T, S_{\delta})$  for  $\delta < 1$ ,  $r_1^{[2],\varepsilon}(T, S_{\delta}) = (S_1, S_1) = (S_1, S_1)$  and  $r_1^{[2],+}(T, S_1) = (T, S_1)$ .

We have the following measurability result.

**Lemma 5.21** (Measurability of some truncation maps). Let  $\varepsilon \in \{+, -\}$ . The map  $(t, (T, S)) \mapsto r_t^{[2],\varepsilon}(T,S)$  is measurable from  $\mathbb{R}_+ \times \mathbb{T}_{loc-K}^{[2]}$  to  $\mathbb{T}_{loc-K}^{[2]}$ .

Proof. Let a > 0. For a marked tree  $(T, S) = (T, S, d, \varrho)$ , we define its partial dilatation  $R_a(T, S) = (T, S, d_a, \varrho)$  as the marked tree with  $d_a(x, y) = ad(x, p_S(x)) + d(p_S(x), p_S(y)) + ad(y, p_S(y))$ . Intuitively the distances on T are multiplied by a outside S. The equivalence class of  $R_a(T, S)$  in  $\mathbb{T}^{[2]}_{\text{loc}-\text{K}}$  does not depend of the choice of (T, S) in its equivalence class in  $\mathbb{T}^{[2]}_{\text{loc}-\text{K}}$ ; so the map  $R_a$  is well defined on  $\mathbb{T}^{[2]}_{\text{loc}-\text{K}}$  to itself. Notice that the map  $R_a$  is continuous and one-to-one with inverse  $R_{1/a}$ . It is immediate to check that, for  $t \geq 0$ :

$$r_t^{[2],-} = \lim_{a \to 0+} R_{1/a} \circ r_t^{[2]} \circ R_a.$$

This and Lemma 5.16 imply the measurability of the map  $(t, (T, S)) \mapsto r_t^{[2],-}(T, S)$ . Then, notice that  $\lim_{s \downarrow t} r_s^{[2],-} = r_t^{[2],+}$  to get the measurability of the map  $(t, (T, S)) \mapsto r_t^{[2],+}(T, S)$ .

We end this section by proving (in a very similar way) that the map  $r_*^{[2]}$  below, which consists in cleaning the root, that is, in erasing the bushes at the root of a marked tree is measurable. For  $(T, S) = (T, S, d, \rho)$  a marked locally compact rooted tree, we set:

(46) 
$$r_*^{[2]}(T,S) = \left(r_{*,1}^{[2]}(T,S),S\right) \text{ with } r_{*,1}^{[2]}(T,S) = \left\{x \in T : p_S(x) \neq \varrho\right\} \cup \{\varrho\}.$$

We also denote by  $r_*^{[2]}(T, S)$  the marked rooted tree  $(r_*^{[2]}(T, S), d, \varrho)$  endowed with the restriction of the distance d and the root  $\varrho$ . Furthermore, if (T, S) and (T', S') belong to the same equivalence class of  $\mathbb{T}_{loc-K}^{[2]}$ , then so do  $r_*^{[2]}(T, S)$  and  $r_*^{[2]}(T', S')$ . Thus the map  $r_*^{[2]}$  is well-defined from  $\mathbb{T}_{loc-K}^{[2]}$  to  $\mathbb{T}_{loc-K}^{[2]}$ .

**Lemma 5.22** (Measurability of the root cleaning map). The map  $r_*^{[2]}$  is measurable from  $\mathbb{T}_{loc-K}^{[2]}$  to  $\mathbb{T}_{loc-K}^{[2]}$ .

Proof. Let a > 0. For a marked tree  $(T, S) = (T, S, d, \varrho)$ , we define its partial dilatation  $R'_a(T, S) = (T, S, d'_a, \varrho)$  as the marked tree with  $d'_a(x, y) = F_a(t)d(x, y)$  if  $p_S(x) = p_S(y)$  with  $t = H(p_S(x))$ , and otherwise  $d'_a(x, y) = F_a(t)d(x, p_S(x)) + ad(p_S(x), p_S(y)) + F_a(s)d(y, p_S(y))$  with  $t = H(p_S(x))$ ,  $s = H(p_S(y))$ , and the function  $F_a$  defined for  $t \ge 0$  by  $F_a(t) = t \land a + a^{-2}(a-t)_+$  if  $a \le 1$ , and  $F_a(t) = 1/F_{1/a}(at)$  if a > 1. The equivalence class of  $R'_a(T, S)$  in  $\mathbb{T}^{[2]}_{\text{loc}-K}$  does not depend of the choice of (T, S) in its equivalence class in  $\mathbb{T}^{[2]}_{\text{loc}-K}$ ; so the map  $R'_a$  is well defined on  $\mathbb{T}^{[2]}_{\text{loc}-K}$  to itself. Notice that the map  $R'_a$  is continuous and one-to-one with inverse  $R'_{1/a}$ . It is immediate to check that for t > 0:

$$r_*^{[2]} = \lim_{a \to 0+} R_{1/a} \circ r_t^{[2]} \circ R_a.$$

This and Lemma 5.16 imply the measurability of the map  $r_*^{[2]}$ .

5.6. Set of (equivalence classes of ) trees with one infinite marked branch. Let us denote by  $T_0 = (\varrho, \{\varrho\})$  the rooted tree reduced to its root. Notice that  $r_0^{[2],+}(T,S) = \{(T_0,T_0)\}$  if and only if  $[\![\varrho,x]\!] \cap S = \{\varrho\}$  implies  $x = \varrho$ . Let  $T_1 = ([0,\infty), d, 0)$  be the tree consisting of only one infinite branch. We consider the set (of equivalence classes) of locally compact rooted trees with one infinite marked branch and its subset of trees whose root is not a branching vertex:

(47) 
$$\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{\mathrm{spine}} = \left\{ (T,S) \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{[2]} : S = T_1 \text{ in } \mathbb{T}_{\mathrm{loc}-\mathrm{K}} \right\},$$

(48) 
$$\mathbb{T}^{\text{spine},0}_{\text{loc}-K} = \left\{ (T,S) \in \mathbb{T}^{\text{spine}}_{\text{loc}-K} : \varrho \notin Br(T) \right\}.$$

**Lemma 5.23.** The sets  $\mathbb{T}^{\text{spine}}_{\text{loc}-\text{K}}$  and  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$  are Borel subsets of  $\mathbb{T}^{[2]}_{\text{loc}-\text{K}}$ .

*Proof.* Consider the projection  $\tilde{\Pi} : (T, S) \mapsto S$  from  $\mathbb{T}^{[2]}_{loc-K}$  to  $\mathbb{T}_{loc-K}$ , which is by construction 1-Lipschitz and thus continuous. As  $\mathbb{T}^{\text{spine}}_{loc-K} = \tilde{\Pi}^{-1}(\{T_1\})$ , we get that  $\mathbb{T}^{\text{spine}}_{loc-K}$  is Borel.

Notice that for  $(T,S) \in \mathbb{T}_{loc-K}^{spine}$ , then, by definition of  $r_t^{[2],+}$ , we get that the root is not a branching vertex of (T,S) if and only if  $r_0^{[2],+}(T,S) = (T_0,T_0)$ . Then, the set  $\mathbb{T}_{loc-K}^{spine,0} = \mathbb{T}_{loc-K}^{spine} \cap (r_0^{[2],+})^{-1} \left( \left\{ (T_0,T_0) \right\} \right)$  is Borel as the map  $r_0^{[2],+}$  is measurable according to Lemma 5.21.  $\Box$ 

We shall be mainly consider elements of  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$  in what follows. For simplicity, we shall write  $T^* = (T, S)$  for an element of  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$ . For  $t \ge 0$  and  $T^* = (T, S)$  in  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$ , we have  $r_t^{[2],+}(T^*) = (r_{t,1}^{[2],+}(T), r_t(S))$  where the rooted tree  $r_t(S)$  is given by  $(\llbracket \varrho, x \rrbracket, \varrho)$  with  $x \in S$ 

uniquely characterized by  $d(\varrho, x) = t$ . We shall consider a slight modification of  $r_t^{[2],+}$  on  $\mathbb{T}_{loc-K}^{spine,0}$ , say  $\tilde{r}_t^{[2],+}$ , where one keeps track only of  $(\varrho, x)$  instead of  $r_t(S)$ :

(49) 
$$\tilde{r}_t^{[2],+}(T^*) = \left(r_{t,1}^{[2],+}(T), (\varrho, x)\right).$$

It is left to the reader to check that  $\tilde{r}_t^{[2],\varepsilon}$  is defined on  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$  and  $\mathbb{T}^{(1)}_{\text{loc}-\text{K}}$ -valued. Similarly to Lemma 5.21, we get the following result.

**Lemma 5.24.** The function  $(t, T^*) \mapsto \tilde{r}_t^{[2],+}(T^*)$  from  $\mathbb{R}_+ \times \mathbb{T}^{\text{spine},0}_{\text{loc}-K}$  to  $\mathbb{T}^{(1)}_{\text{loc}-K}$  is measurable.

5.7. Another representation for discrete trees. Let  $n \in \mathbb{N}$  be fixed. Let  $(T, \mathbf{v})$ , with  $\mathbf{v} = (v_0 = \varrho, \ldots, v_n)$ , be a locally compact rooted *n*-pointed tree. We will decompose the tree  $\operatorname{Span}(T, \mathbf{v})$  as a sequence of edges. To do so, we introduce some notations. Let  $A \subset \{0, \ldots, n\}$  be non-empty. We set  $\mathbf{v}_A = (v_i, i \in A)$ . We denote by  $v_A$  the most recent common ancestor of  $\mathbf{v}_A$ , which is the only element of T such that:

(50) 
$$\llbracket \varrho, v_A \rrbracket = \bigcap_{k \in A} \llbracket \varrho, v_k \rrbracket.$$

Notice that  $v_{\{i\}} = v_i$ . Recall that for  $x \in T$ ,  $T_x$  is the sub-tree of T above x and rooted at x. Let  $\mathcal{P}_n^+$  be the set of all subsets  $A \subset \{1, \ldots, n\}$  such that  $A \neq \emptyset$ . For  $A \in \mathcal{P}_n^+$ , if  $T_{v_A} \cap \operatorname{Span}^{\circ}(T, \mathbf{v}_{A^c}) \neq \emptyset$  with  $A^c = \{0, 1, 2, \cdots, n\} \setminus A$ , we set  $w_A = v_A$ , otherwise we define  $w_A \in [\rho, v_A]$  as the only element of T such that:

(51) 
$$\llbracket \varrho, w_A \rrbracket = \operatorname{Span}^{\circ}(T, \mathbf{v}_{A^{\circ}}) \cap \operatorname{Span}^{\circ}(T, (\varrho, \mathbf{v}_A)).$$

Equivalently  $w_A$  is the only element in  $\llbracket \varrho, v_A \rrbracket$  such that  $w_A = v_{A \cup \{k_0\}}$  for some  $k_0 \in A^c$  and for all  $k \in A^c$ , we have  $v_{A \cup \{k\}} \in \llbracket \varrho, w_A \rrbracket$ . Notice that  $w_{\{1,\ldots,n\}} = \varrho$ . We also record the lengths of all the branches  $\llbracket w_A, v_A \rrbracket$ :

(52) 
$$\mathbf{L}_n(T, \mathbf{v}) = \left(\ell_A(T, \mathbf{v}), A \in \mathcal{P}_n^+\right) \quad \text{with} \quad \ell_A(T, \mathbf{v}) = d(w_A, v_A).$$



FIGURE 2. A discrete trees spanned by the leaves  $\{1, 2, 3\}$ .

TABLE 1. Quantities of interest for the discrete tree from Figure 2.

I	$A \subset \mathcal{P}_3^+$	{1}	$\{2\}$	$\{3\}$	$\{1, 2\}$	$\{1,3\}$	$\{2,3\}$	$\{1, 2, 3\}$
	$v_A$	1	2	3	a	a	b	a
	$w_A$	a	b	b	a	a	a	Q
	$\ell_A$	d(a,1)	d(b,2)	d(b,3)	0	0	d(a,b)	$d(\varrho, a)$

For instance, we record the quantity of interest in Table 1 for the discrete tree spanned by the leaves  $\{1, 2, 3\}$  from Figure 2. We can see that each branch of the discrete tree appears (through their length) once and only once in  $\mathbf{L}_3(T, \mathbf{v})$ .

Set  $\hat{\mathbf{v}} = (\hat{\mathbf{v}}_0 = \varrho, (v_A, A \in \mathcal{P}_n^+)) \in T^{2^n}$ , so that  $(T, \hat{\mathbf{v}})$  is a locally compact rooted  $(2^n - 1)$ pointed tree with the same root  $\varrho$  as T. Notice that all the vertices in  $\mathbf{v}$  appear in  $\hat{\mathbf{v}}$  (possibly more than once), and that  $w_A$  also appears in  $\hat{\mathbf{v}}$  for all  $A \in \mathcal{P}_n^+$ . Recall the set of discrete trees defined at the end of Section 5.3. The next lemma states that  $\mathbf{L}_n$  codes continuously for discrete trees. Set  $\operatorname{Im}(\mathbf{L}_n) \subset \mathbb{R}_+^{\mathcal{P}_n^+}$  (with  $\mathbb{R}_+^{\mathcal{P}_n^+} = \mathbb{R}_+^{2^n-1}$ ) for the image of  $\mathbf{L}_n$ .

**Lemma 5.25** (Regularity of the branch lengths as a function of the tree). Let  $n \in \mathbb{N}^*$ . The map  $(T, \mathbf{v}) \mapsto (T, \hat{\mathbf{v}})$  is well defined from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(2^n-1)}$ , and it is continuous. The function  $\mathbf{L}_n$  is well defined from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathrm{Im}(\mathbf{L}_n) \subset \mathbb{R}_+^{\mathcal{P}_n^+}$  and is continuous. Furthermore,  $\mathrm{Im}(\mathbf{L}_n)$  is closed and  $\mathbf{L}_n$  is a one-to-one bi-measurable map from  $\mathbb{T}_{dis}^{(n)}$  to  $\mathrm{Im}(\mathbf{L}_n)$ .

*Proof.* If  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  belong to the same equivalence class in  $\mathbb{T}_{loc-K}^{(n)}$ , then we deduce from (50) and (51) that  $(T, \hat{\mathbf{v}})$  and  $(T', \hat{\mathbf{v}}')$  belong also to the same equivalence class. This implies that the function  $(T, \mathbf{v}) \mapsto (T, \hat{\mathbf{v}})$  is well defined from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(2^n-1)}$ . We deduce from (50) and (51) that this function is in fact continuous on  $\mathbb{T}_{loc-K}^{(n)}$ . We also get that the function  $\mathbf{L}_n$  is well defined from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{R}_+^{\mathcal{P}_n^+}$ .

We shall now precise the image of the function  $\mathbf{L}_n$  and prove its continuity. Recall  $x_+ = \max(x, 0)$  denotes the positive part of  $x \in \mathbb{R}$ . We define the function L from  $\mathbb{R}^{(n+1)\times(n+1)}_+$  to  $\mathbb{R}^{\mathcal{P}^n_h}_+$  by, for  $\mathbf{d} = (\mathbf{d}_{ij}, 0 \leq i, j \leq n)$  and  $A \in \mathcal{P}^+_n$ :

$$L_A(d) = \frac{1}{4} \inf \left\{ \left( d_{ii'} + d_{ij'} + d_{ji'} + d_{jj'} - 2d_{ij} - 2d_{i'j'} \right)_+ : i, j \in A \text{ and } i', j' \in A^c \right\},\$$

where  $A^c = \{0, \ldots, n\} \setminus A$ . We also define the function D from  $\mathbb{R}^{\mathcal{P}^+_n}$  to  $\mathbb{R}^{(n+1)\times(n+1)}_+$  by, for  $\ell = (\ell_A, A \in \mathcal{P}^+_n)$  and  $i, j \in \{0, \ldots, n\}$ :

(53) 
$$D_{ij}(\ell) = \sum_{A \in \mathcal{P}_n^+} \ell_A \left( \mathbf{1}_{\{i \in A, j \notin A\}} + \mathbf{1}_{\{i \notin A, j \in A\}} \right).$$

The functions L and D are continuous. Consider the closed subset  $\mathcal{Q}^{(n)}$  of  $\mathbb{R}^{(n+1)\times(n+1)}_+$  satisfying the so-called four-point condition, that is the set of all  $(d_{ij}, 0 \leq i, j \leq n) \in \mathbb{R}^{(n+1)\times(n+1)}_+$  such that:

 $d_{ij} + d_{i'j'} \le \max(d_{ii'} + d_{jj'}, d_{ij'} + d_{ji'})$  for all  $i, j, i', j' \in \{0, \dots, n\}$ .

Notice that the four-point condition is also used to characterize metric spaces which are real trees, see Evans[?]. Then, one can check that the function L is one-to-one from  $\mathcal{Q}^{(n)}$  to  $L(\mathcal{Q}^{(n)})$  with inverse D. We also get that  $L(\mathcal{Q}^{(n)})$  is closed (indeed if  $(\ell^k = L(d^k), k \in \mathbb{N})$  is a sequence of elements of  $L(\mathcal{Q}^{(n)})$  converging to a limit, say  $\ell$ , then it is bounded and thus the sequence  $(d^k, k \in \mathbb{N})$  is also bounded. Hence there is a converging sub-sequence, and denote by d its limit which belongs to  $\mathcal{Q}^{(n)}$  as this set is closed. Since L is continuous, we get that  $L(d) = \ell$  and thus  $\ell$  belongs to  $L(\mathcal{Q}^{(n)})$ , which gives that  $L(\mathcal{Q}^{(n)})$  is closed). Since for  $(T, \mathbf{v}) \in \mathbb{T}^{(n)}_{\text{loc}-\text{K}}$ , we have that  $\mathbf{L}_n(T, \mathbf{v}) = L(d(v_i, v_j), 0 \leq i, j \leq n)$ , we deduce that the function  $\mathbf{L}_n$  is continuous from  $\mathbb{T}^{(n)}_{\text{loc}-\text{K}}$  to  $L(\mathcal{Q}^{(n)})$ .

28

We now prove that  $\operatorname{Im}(\mathbf{L}_n) = L(\mathcal{Q}^{(n)})$  and that  $\mathbf{L}_n$  is one-to-one from  $\mathbb{T}_{\operatorname{dis}}^{(n)}$  to  $L(\mathcal{Q}^{(n)})$ . Let  $\ell = (\ell_A, A \in \mathcal{P}_n^+) \in L(\mathcal{Q}^{(n)})$ . Thus, there exists a sequence  $d = (d_{ij}, 0 \leq i, j \leq n) \in \mathcal{Q}^{(n)}$  which satisfies the four-point condition and such that  $L(d) = \ell$ . Since d satisfies the four-point condition, we get that there exists a discrete tree  $(T, d, \mathbf{v}) \in \mathbb{T}_{\operatorname{dis}}^{(n)}$  such that  $d(v_i, v_j) = d_{ij}$  for all  $i, j \in \{0, \ldots, n\}$ . This proves that  $\operatorname{Im}(\mathbf{L}_n) = L(\mathcal{Q}^{(n)})$ . Then use that L is one-to-one from  $\mathcal{Q}^{(n)}$  to  $L(\mathcal{Q}^{(n)})$  with inverse D and that two discrete trees  $(T, d, \mathbf{v})$  and  $(T', d', \mathbf{v}')$  are equal in  $\mathbb{T}_{\operatorname{dis}}^{(n)}$  if and only if  $d(v_i, v_j) = d'(v'_i, v'_j)$  for all  $i, j \in \{0, \ldots, n\}$  to deduce that  $\mathbf{L}_n$  is one-to-one from  $\mathbb{T}_{\operatorname{dis}}^{(n)}$  to  $L(\mathcal{Q}^{(n)})$  and thus bi-measurable thanks to Lusin's theorem from [32].

5.8. The splitting operator for a pointed tree. We want now to decompose the pointed tree  $(T, \mathbf{v})$  along the branches of  $\text{Span}^{\circ}(T, \mathbf{v})$ . We keep notations from Section 5.7.

Let  $(T, \mathbf{v})$ , with  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$ , be a locally compact rooted *n*-pointed tree. Recall Definition (27) of the projection  $p_{\mathbf{v}}$  on  $\operatorname{Span}(T, \mathbf{v})$ . For  $A \in \mathcal{P}_n^+$ , consider the rooted 1-pointed tree:

(54) 
$$\hat{T}_A(T, \mathbf{v}) = \left(T_A(T, \mathbf{v}), (\varrho_A, v_A)\right) \in \mathbb{T}^{(1)}_{\text{loc}-\text{K}},$$

with root  $\rho_A = w_A$  and

$$T_A(T, \mathbf{v}) = \left\{ x \in T : p_{\mathbf{v}}(x) \in ]\!] w_A, v_A]\!] \right\} \cup \left\{ w_A \right\}$$

By construction, we have that  $\ell_A(T, \mathbf{v}) = d(\varrho_A, v_A)$ .

Notice that  $\ell_A(T, \mathbf{v}) = 0$  if and only if  $\hat{T}_A(T, \mathbf{v})$  is reduced to its root, that is,  $(\{\varrho_A\}, (\varrho_A, \varrho_A))$ . Notice also that  $\ell_A(T, \mathbf{v}) > 0$  implies that  $\hat{T}_A$  belongs to  $\mathbb{T}^{(1),0}_{\text{loc}-K}$ , the set of trees in  $\mathbb{T}^{(1)}_{\text{loc}-K}$  such that the root is not a branching point (see Definition (35)). We also define the rooted 1-pointed tree  $\hat{T}_{\{0\}}(T, \mathbf{v}) \in \mathbb{T}^{(1)}_{\text{loc}-K} = (T_{\{0\}}(T, \mathbf{v}), (\varrho, \varrho))$  by:

$$T_{\{0\}}(T, \mathbf{v}) = \left\{ x \in T : \, ]\!]\varrho, x]\!] \cap \operatorname{Span}^{\circ}(T, \mathbf{v}) = \emptyset \right\},\$$

with root  $\rho$  and distinguished vertex also  $\rho$ . If  $(T, \mathbf{v})$  and  $(T', \mathbf{v}')$  belong to the same equivalence class in  $\mathbb{T}^{(n)}_{\text{loc}-\text{K}}$ , then we get that  $\hat{T}_A(T, \mathbf{v})$  and  $\hat{T}_A(T', \mathbf{v}')$  belong also to the same equivalent class in  $\mathbb{T}^{(1)}_{\text{loc}-\text{K}}$  for  $A \in \mathcal{P}_n = \mathcal{P}_n^+ \cup \{\{0\}\}$ . Thus, the map Split<sub>n</sub> defined on  $\mathbb{T}^{(n)}_{\text{loc}-\text{K}}$  by:

(55) 
$$\operatorname{Split}_{n}(T, \mathbf{v}) = \left(\hat{T}_{A}(T, \mathbf{v}), A \in \mathcal{P}_{n}\right)$$

takes values in  $\left(\mathbb{T}_{loc-K}^{(1)}\right)^{2^n}$ . We give an instance of the function  $\text{Split}_n$  in Figure 3.

**Lemma 5.26** (Measurability of the splitting map). Let  $n \in \mathbb{N}^*$ . The map  $\operatorname{Split}_n$  from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\left(\mathbb{T}_{\operatorname{loc}-K}^{(1)}\right)^{2^n}$  is measurable.

*Proof.* The proof is divided into three steps.

Step 1: The map  $\hat{T}_{\{0\}}$  is measurable. Let  $(T, \mathbf{v}) \in \mathbb{T}_{\text{loc}-\text{K}}^{(n)}$ . By construction, we have that  $r_0^{[2],+}(T, \text{Span}^{\circ}(T, \mathbf{v})) = (T_{\{0\}}(T, \mathbf{v}), T_0)$ . We deduce from Lemma 5.21 on the measurability of  $r_t^{[2],\varepsilon}$ , that the map  $(T, \mathbf{v}) \mapsto \hat{T}_{\{0\}} = (T_{\{0\}}(T, \mathbf{v}), (\varrho, \varrho))$  is measurable.

Step 2: A measurable truncation function. Let  $n \ge 1$ . Let  $(T, \mathbf{v})$  be a rooted *n*-pointed tree. Recall the definition of  $\hat{T}_A(T, \mathbf{v})$  from (54). We set  $q(T, \mathbf{v}) = \hat{T}_{\{1,2,\ldots,n\}}(T, \mathbf{v})$  so that q is a map from  $\mathbb{T}_{\text{loc}-K}^{(n)}$  to  $\mathbb{T}_{\text{loc}-K}^{(1)}$ . Recall the measurable truncation functions  $r_t^{[2],+}$  and  $r_*^{[2]}$  from (49) and (46), respectively.



FIGURE 3. The splitting of the left hand tree with respect to  $\mathbf{v} = \{\varrho, 1, 2, 3\}$ . In this instance,  $T_{\{1,2\}}$  and  $T_{\{1,3\}}$  are reduced to their own root.

We set:

$$q'(T, \mathbf{v}) = r_*^{[2]} \circ r_{d(\varrho, w_{\{1, \dots, b\}})}^{[2], +} \left(T, \operatorname{Span}^{\circ}(T, \mathbf{v})\right).$$

Thanks to Lemma 5.17, the map  $(T, \mathbf{v}) \mapsto (T, \operatorname{Span}^{\circ}(T, \mathbf{v}))$  is continuous from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$ . Thanks to Lemma 5.25 and Remark 5.11, we get that the map  $(T, \mathbf{v}) \mapsto d(\varrho, w_{\{1,\ldots,b\}})$  is continuous from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\mathbb{R}_+$ . Then, use Lemmas 5.21 and 5.22 on the measurability of  $r_t^{[2],\varepsilon}$  and  $r_*^{[2]}$  to conclude that the map q' from  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$  is measurable and it has the same image as the map  $(T, (\varrho, v)) \mapsto (T, \llbracket \varrho, v \rrbracket)$  from  $\mathbb{T}_{\operatorname{loc}-K}^{(1)}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$ . According to Lemma 5.17 (with n = 1), this latter map is injective and measurable. Hence the map q, which is the composition of q' and this latter map, is measurable.

Step 3: Conclusion. Let  $A \subset \{1, \ldots, n\}$  be non-empty. Notice that  $\hat{T}_A$  is the image of  $(T, \mathbf{v})$  by: the expansion procedure  $(T, \mathbf{v}) \mapsto (T, \hat{\mathbf{v}})$  from the first part of Lemma 5.25, the rerooting at  $w_A$  from Lemma 5.8, the reducing procedure from Lemma 5.5 where one forgets about all  $w_{A'}$  and  $v_{A'}$  for  $A' \subset A^c$ , and then the function q from Step 2. This implies that the function  $(T, \mathbf{v}) \mapsto \hat{T}_A(T, \mathbf{v})$  is measurable from  $\mathbb{T}_{\text{loc}-\text{K}}^{(n)}$  to  $\mathbb{T}_{\text{loc}-\text{K}}^{(1)}$ .

5.9. The grafting procedure. Let  $n \in \mathbb{N}^*$ . Let  $\ell = (\ell_A, A \in \mathcal{P}_n^+) \in \text{Im}(\mathbf{L}_n)$ . According to Lemma 5.25, there exists a unique (up to the equivalence in  $\mathbb{T}_K^{(n)}$ ) rooted *n*-pointed discrete tree  $(S, \mathbf{v})$  (that is  $S = \text{Span}^{\circ}(S, \mathbf{v})$ ) such that  $\mathbf{L}_n(S, \mathbf{v}) = \ell$ . Recall  $v_A$  and  $w_A$  defined in Section 5.7 for  $A \in \mathcal{P}_n^+$  so that:

(56) 
$$S = \bigcup_{A \in \mathcal{P}_n^+} \llbracket w_A, v_A \rrbracket$$

where the sets  $(]\!]w_A, v_A[\![, A \in \mathcal{P}_n^+)$  are pairwise disjoint.

Recall that  $\mathbb{T}_{\text{loc}-K}^{\text{spine},0}$  denotes the set (of equivalence classes) of locally compact rooted trees with one infinite marked branch such that the root is not a branching vertex. Let  $T^* = (T_A^*, A \in \mathcal{P}_n^+)$ be a family of elements of equivalence classes in  $\mathbb{T}_{\text{loc}-K}^{\text{spine},0}$ . Then, we define the tree  $(T, \mathbf{v}) = \text{Graft}_n(\ell, T^*)$ , where T is the tree S with that the branches  $]\!]w_A, v_A]$  are replaced by the trees given by the first component of  $r_{\ell_A}^{[2],+}(T_A^*)$  (where the second component has been identified to  $[\![w_A, v_A]\!]$ ).

We now provide a more formal construction of  $\operatorname{Graft}_n(\ell, T^*)$ . Let  $\ell \in \operatorname{Im}(\mathbf{L}_n)$ , and consider the rooted *n*-pointed discrete tree  $(S, \mathbf{v}) = \mathbf{L}_n^{-1}(\ell) \in \mathbb{T}_{\operatorname{dis}}^{(n)}$  and  $\mathbf{v} = (v_0 = \varrho, \dots, v_n)$ . Set  $\hat{\mathbf{v}} = (\hat{\mathbf{v}}_0 = \varrho, (v_A, A \in \mathcal{P}_n^+)) \in T^{2^n}$ , with  $v_A$  the most recent common ancestor of  $(v_i, i \in A)$  defined in (50). Thus, we get that  $(S, \hat{\mathbf{v}}) \in \mathbb{T}_{\operatorname{dis}}^{(2^n-1)}$  is a rooted  $(2^n - 1)$ -pointed discrete tree with the same root  $\varrho$  as S.

In a first step, we build by a backward induction an "increasing" sequence of discrete trees  $((S_k, \mathbf{v}_k), k \in \{0, \dots, 2^n - 1\})$  such that  $(S_k, \mathbf{v}_k) \in \mathbb{T}^{(k)}_{\text{dis}}$  with root  $\varrho$ . We set  $(S_{2^n-1}, \mathbf{v}_{2^n-1}) = (S, \hat{\mathbf{v}})$ . Recall that x is a leaf of a tree T with root  $\varrho$  if  $x \in [\![\varrho, y]\!] \subset T$  implies y = x. Assume that  $(S_{k+1}, \mathbf{v}_{k+1})$  is defined for some  $k \geq 0$ . We endow the sets  $\mathcal{P}^+_n$  and  $\mathcal{P}_n = \mathcal{P}^+_n \cup \{\{0\}\}$  with the lexicographical order, so that the maximum of the subset of  $\mathcal{P}^+_n$  is well-defined. We set:

 $A_{k+1} = \max\{A \in \mathcal{P}_n^+, v_A \in \mathbf{v}_{k+1} \text{ and } v_A \text{ is a leaf of } (S_{k+1}, \mathbf{v}_{k+1})\}.$ 

Then, we define  $\mathbf{v}_k$  as the sequence  $\mathbf{v}_{k+1}$  where  $v_{A_{k+1}}$  has been removed (notice that the first element of  $\mathbf{v}_k$  is still the root  $\varrho$ ), and we set  $(S_k, \mathbf{v}_k) = \text{Span}(S, \mathbf{v}_k) \in \mathbb{T}_{\text{dis}}^{(k)}$ . We also set  $B_k = \max\{B \in \mathcal{P}_n : v_B = w_{A_{k+1}}\}$ . By construction,  $v_{B_k} = w_{A_{k+1}}$  belongs to the sequence  $\mathbf{v}_k$  and is therefore an element of  $\mathbf{v}$  for some index, and, with a slight abuse of notation, we simply denote this index by  $B_k$ . We have, using the grafting operation from Section 5.4 that:

(57) 
$$(S_{k+1}, \mathbf{v}_{k+1}) = (S_k, \mathbf{v}_k) \circledast_{B_k} [0, \ell_{A_{k+1}}],$$

where the equality holds in  $\mathbb{T}_{\text{loc}-K}^{(k+1)}$  (and in  $\mathbb{T}_{\text{dis}}^{(k+1)}$ ) and by convention [0, t] denotes the discrete 1-pointed tree ([0, t], (0, t)) with root 0. Notice that  $\ell_{A_{k+1}} = 0$  if and only if  $\text{Span}^{\circ}(S, \mathbf{v}_k) = \text{Span}^{\circ}(S, \mathbf{v}_{k+1})$ . Eventually, notice that  $(S_0, \mathbf{v}_0) = (\{\varrho\}, \varrho)$  is the rooted tree reduced to its root  $\varrho = v_{\{0\}}$  and  $B_0 = \{0\}$ . Let us stress, that in Section 5.4, the vector  $\mathbf{v}_{k+1}$  is obtained by adding the distinguished vertex  $\ell_{A_{k+1}}$  of  $[0, \ell_{A_{k+1}}]$  to  $\mathbf{v}_k$ . However here we identify  $[0, \ell_{A_{k+1}}]$  with  $[v_{B_k} = w_{A_{k+1}}, v_{A_{k+1}}]$  and add the distinguished vertex  $v_{A_{k+1}}$  to  $\mathbf{v}_k$  in order to obtain  $\mathbf{v}_{k+1}$ .

For instance, we give in Table 2 the sequences  $(A_k, 1 \le k \le 2^n - 1)$  and  $(B_k, 0 \le k \le 2^n - 2)$  for the tree of Figure 2.

TABLE 2. The sequences  $(A_{k+1}, 0 \le k \le 6)$ ,  $(B_k, 0 \le k \le 6)$  and  $(\ell_{A_{k+1}}, 0 \le k \le 6)$  for the tree of Figure 2.

k	0	1	2	3	4	5	6
$A_{k+1}$	$\{1,2\}$	$\{1, 2, 3\}$	$\{1,3\}$	{1}	$\{2,3\}$	$\{2\}$	{3}
$B_k$	{0}	$\{1,3\}$	$\{1,2\}$	$\{1, 2, 3\}$	$\{1, 2, 3\}$	$\{2,3\}$	$\{2,3\}$
$\ell_{A_{k+1}}$	$d(\varrho, a)$	0	0	d(1,a)	d(a,b)	d(2,b)	d(3,b)

Remark 5.27. The family  $\{A_k, k \in \{1, 2^n - 1\}\}$  is exactly equal to  $\mathcal{P}_n^+$ . Furthermore the sequence  $\ell \in \text{Im}(\mathbf{L}_n) \subset \mathbb{R}_+^{2^n-1}$  provides implicitly two unique ordered sequences  $\mathcal{A}(\ell) = (A_k, k \in \{1, 2^n - 1\})$  (of all elements of  $\mathcal{P}_n^+$ ) and  $\mathcal{B}(\ell) = (B_k, k \in \{0, 2^n - 2\})$  (of elements of  $\mathcal{P}_n^+ \cup \{\{0\}\})$ ,

and an "increasing" way to built  $\mathbf{L}_n^{-1}(\ell)$  recursively by adding at step  $k \in \{0, 2^n - 2\}$  a branch of length  $\ell_{A_{k+1}}$  (and graft it on  $v_{B_k}$  chosen among  $\mathbf{v}_k$ ). It is obvious from the construction that if  $\ell$  and  $\ell'$  are two sequences in  $\text{Im}(\mathbf{L}_n)$  with the same zeros (that is,  $\ell_A = 0$  if and only if  $\ell'_A = 0$ ), then we have  $\mathcal{A}(\ell) = \mathcal{A}(\ell')$  and  $\mathcal{B}(\ell) = \mathcal{B}(\ell')$ . Thus, the sets  $\mathcal{A}(\ell)$  and  $\mathcal{B}(\ell)$  are implicitly coded by the zeros of  $\ell$ .

In a second step, given  $\mathcal{A}(\ell)$  and  $\mathcal{B}(\ell)$  from Remark 5.27 and a sequence  $T^* = (T_A^*, A \in \mathcal{P}_+^n)$ in  $\mathbb{T}_{loc-K}^{spine,0}$ , we build by a forward induction an "increasing" sequence of marked locally compact trees  $((T_k, \mathbf{v}_k), k \in \{0, \ldots, 2^n - 1\})$  such that  $(T_k, \mathbf{v}_k)$  belongs to  $\mathbb{T}_{loc-K}^{(k)}$ , has root  $\varrho$ , and the components of the vector  $\mathbf{v}_k$  can be ranked as the root  $\varrho = v_{\{0\}}$  and  $(v_{A_i}, 1 \leq i \leq k)$ . Recall also the truncation function  $\tilde{r}_t^{[2],+}$  given in (49). We set  $(T_0, \mathbf{v}_0) = (\{\varrho\}, \varrho)$  and for  $k \in \{0, 2^n - 2\}$ :

(58) 
$$(T_{k+1}, \mathbf{v}_{k+1}) = (T_k, \mathbf{v}_k) \circledast_{B_k} \tilde{r}_{\ell_{A_{k+1}}}^{[2], +} (T_{A_{k+1}}^*),$$

where the distinguished vertex of  $\tilde{r}_{\ell_{A_{k+1}}}^{[2],+}(T^*_{A_{k+1}})$  is identified with  $v_{A_{k+1}}$  (and its root with  $v_{B_k}$ ). Then, we set:

(59) Graft<sub>n</sub>( $\ell, T^*$ ) = ( $T_{2^n-1}, \mathbf{v}$ ) with  $\mathbf{v} = (v_{\{k\}}, 0 \le k \le n)$ .



FIGURE 4. Example of a replacement of the branch  $]\!]w_{\{1,\dots,n\}}, v_{\{1,\dots,n\}}]\!]$ . Upper left: The tree S with the branch  $]\!]w_{\{1,\dots,n\}}, v_{\{1,\dots,n\}}]\!]$  in bold. Upper right: The branch  $]\!]w_{\{1,\dots,n\}}, v_{\{1,\dots,n\}}]\!]$  replaced by the first component of the marked tree  $r_{\ell_{\{1,\dots,n\}}}^{[2],+}(T^*_{\{1,\dots,n\}})$ . Lower: The tree  $T^*_{\{1,\dots,n\}}$  with its marked infinite branch.

It is easy to check that the equivalence class of  $(T_{2^n-1}, \mathbf{v})$  in  $\mathbb{T}_{\text{loc}-K}^{(n)}$  does not depend on the choice of  $T^* = (T^*_A, A \in \mathcal{P}^+_n)$  in their own equivalence class. Thus, the map Graft<sub>n</sub> defined by:

 $(\ell, T^*) \mapsto \operatorname{Graft}_n(\ell, T^*)$ 

is well defined from  $\operatorname{Im}(\mathbf{L}_n) \times \left(\mathbb{T}_{\operatorname{loc}-K}^{\operatorname{spine},0}\right)^{\mathcal{P}_n^+}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$ . The main result of this section is the measurability of the map  $\operatorname{Graft}_n$ .

**Lemma 5.28** (Measurability of the grafting map). Let  $n \in \mathbb{N}^*$ . The map  $\operatorname{Graft}_n$  from  $\operatorname{Im}(\mathbf{L}_n) \times \left(\mathbb{T}^{\operatorname{spine},0}_{\operatorname{loc}-\mathrm{K}}\right)^{\mathcal{P}_n^+}$  to  $\mathbb{T}^{(n)}_{\operatorname{loc}-\mathrm{K}}$  is measurable.

Proof. For  $J \,\subset\, \mathcal{P}_{+}^{n}$ , we write  $I_{J} = \left\{\ell \in \operatorname{Im}(\mathbf{L}_{n}) : \ell_{A} = 0 \text{ if and only if } A \in J\right\}$ . Thus, the closed set  $\operatorname{Im}(\mathbf{L}_{n})$  of  $\mathbb{R}_{+}^{\mathcal{P}_{+}^{n}}$  can be written as the union of  $I_{J}$  over all the subsets J of  $\mathcal{P}_{+}^{n}$ . Furthermore, the sets  $(I_{J}, J \subset \mathcal{P}_{+}^{n})$  are Borel sets (as  $\operatorname{Im}(\mathbf{L}_{n})$  is a Borel set), and they are pairwise disjoint. Thanks to Remark 5.27, the maps  $\ell \mapsto \mathcal{A}(\ell)$  and  $\ell \mapsto \mathcal{B}(\ell)$  are constant over  $I_{J}$ . We deduce from Equation (59) and recursion (58), Lemma 5.13 on the continuity of the grafting procedure and Lemma 5.24 on the measurability of  $(t,T) \mapsto \tilde{r}_{t}^{[2],+}(T)$  that the function  $\operatorname{Graft}_{n}$  from  $I_{J} \times \left(\mathbb{T}_{\operatorname{loc}-K}^{\operatorname{spine},0}\right)^{\mathcal{P}_{n}^{+}}$  to  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  is measurable (as long as  $I_{J}$  is not empty). Since there is a finite number of such sets  $I_{J}$ , we deduce that the function  $\operatorname{Graft}_{n}$  from  $\operatorname{Im}(\mathbf{L}_{n}) \times \left(\mathbb{T}_{\operatorname{loc}-K}^{\operatorname{spine},0}\right)^{\mathcal{P}_{n}^{+}}$ 

*Remark* 5.29. Since the map  $\mathbf{L}_n$  is continuous one-to-one from  $\mathbb{T}_{dis}^{(n)}$  to  $\text{Im}(\mathbf{L}_n)$ , we deduce that the map:

$$(T,T^*) \mapsto \operatorname{Graft}_n(\mathbf{L}_n(T),T^*)$$

from  $\mathbb{T}_{\text{dis}}^{(n)} \times \left(\mathbb{T}_{\text{loc}-K}^{\text{spine},0}\right)^{\mathcal{P}_n^+}$  to  $\mathbb{T}_{\text{loc}-K}^{(n)}$  is measurable. Without ambiguity, we shall simply write  $\text{Graft}_n(T,T^*)$  for  $\text{Graft}_n(\mathbf{L}_n(T),T^*)$ .

Remark 5.30. Intuitively, the maps  $\operatorname{Graft}_n$  and  $\operatorname{Split}_n$  should be the inverse one of the other. More precisely, we have the following result. For every  $(T, (\varrho, v)) \in \mathbb{T}^{(1)}_{\operatorname{loc}-\mathrm{K}}$ , we define the tree  $\operatorname{Sp}(T) = (T', S') \in \mathbb{T}^{\operatorname{spine},0}_{\operatorname{loc}-\mathrm{K}}$  by  $T' = \Pi^{\circ}_1(T \circledast_1[0,\infty))$  with the marked spine  $S = \Pi^{\circ}_1(\llbracket \varrho, v \rrbracket \circledast_1[0,\infty))$ . Then then we have, for every  $(T, \mathbf{v}) \in \mathbb{T}^{(n),0}_{\operatorname{loc}-\mathrm{K}}$  (that is, the root of T is not a branching vertex, see Definition (35)), that the following equality hold in  $\mathbb{T}^{(n)}_{\operatorname{loc}-\mathrm{K}}$ :

(60) 
$$\operatorname{Graft}_n\left(\operatorname{Span}_n(T,\mathbf{v}), \operatorname{Sp}\left(\operatorname{Split}_n(T,\mathbf{v})\right)\right) = (T,\mathbf{v}),$$

where  $\operatorname{Sp}(T_A, A \in \mathcal{P}_n) = (\operatorname{Sp}(T_A), A \in \mathcal{P}_n^+).$ 

5.10. A measure associated with trees in  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}}$  or  $\mathbb{T}^{(1)}_{\text{loc}-\text{K}}$ . Recall  $T_0 = (\{\varrho\}, \varrho) \in \mathbb{T}_{\text{loc}-\text{K}}$  is the tree reduced to its root. We define

(61) 
$$\mathbb{T}^*_{\text{loc}-K} = \mathbb{T}_{\text{loc}-K} \setminus \{T_0\}$$

endowed with the distance:

$$d_{\text{LGH}}^*(T, T') = d_{\text{LGH}}(T, T') + \left| H(T)^{-1} - H(T')^{-1} \right|.$$

Clearly  $(\mathbb{T}^*_{\text{loc}-K}, d^*_{\text{LGH}})$  is Polish with the topology induced by the topology on  $\mathbb{T}_{\text{loc}-K}$  (as H is continuous on  $\mathbb{T}_{\text{loc}-K}$ ), and for all  $\varepsilon > 0$ , the sets  $B_{\mathbb{T}^*_{\text{loc}-K}}(\varepsilon) = \{T \in \mathbb{T}^*_{\text{loc}-K} : H(T) \ge \varepsilon\}$  are closed and bounded. Furthermore, every bounded set is a subset of  $B_{\mathbb{T}^*_{\text{loc}-K}}(\varepsilon)$  for  $\varepsilon > 0$  small enough. Set  $E = \mathbb{R}_+ \times \mathbb{T}^*_{\text{loc}-K}$  endowed with the distance  $d_E((u,T), (u',T')) = |u - u'| + d^*_{\text{LGH}}(T,T')$ , so that  $(E, d_E)$  is a Polish space. Every bounded set of E is a subset of  $B_E(\varepsilon) = [0, \varepsilon^{-1}] \times B_{\mathbb{T}^*_{\text{loc}-K}}(\varepsilon)$  for  $\varepsilon > 0$  small enough. We define  $\mathbb{M}(E)$ , the set of point measures on E which are bounded on bounded sets, that is finite on  $B_E(\varepsilon)$  for all  $\varepsilon > 0$ . We say that a sequence  $(\mathcal{M}_n, n \in \mathbb{N})$  of elements of  $\mathbb{M}(E)$  converges to a limit  $\mathcal{M}$ , if  $\lim_{n\to\infty} \mathcal{M}_n(f) = \mathcal{M}(f)$ 

for all continuous functions on E with bounded support. According to [14, Proposition 9.1.IV] the space  $\mathbb{M}(E)$  is Polish and the Borel  $\sigma$ -field is the smallest  $\sigma$ -field such that the application  $\mathcal{M} \mapsto \mathcal{M}(A)$  is measurable for every Borel set A of E.

We build a tree from a point measure  $\mathcal{M} = \sum_{i \in I} \delta_{(h_i, T_i)} \in \mathbb{M}(E)$  by grafting  $T_i$  at height  $h_i$  on an infinite spine. Recall the infinite spine  $T_1 = ([0, \infty), 0)$  endowed with the Euclidean distance is an element of  $\mathbb{T}^{\text{spine},0}_{\text{loc}-\text{K}} \subset \mathbb{T}_{\text{loc}-\text{K}}$ . For  $T \in \mathbb{T}_{\text{loc}-\text{K}}$ , let  $(\tilde{T}, d, \varrho)$  denote a rooted locally compact tree in the equivalent class T. With obvious notation, we define the tree T' as follow:

$$\forall x, x' \in T', \ d(x, x') = \begin{cases} d_i(x, x') & \text{if } x, x' \in \tilde{T}_i, i \in I \\ |x - x'| & \text{if } x, x' \in \tilde{T}_1, \\ d_i(x, \varrho_i) + |h_i - x| & \text{if } x \in \tilde{T}_i, \ x' \in \tilde{T}_1, i \in I, \\ d_i(x, \varrho_i) + d_j(x', \varrho_j) + |h_i - h_j| & \text{if } x \in \tilde{T}_i, \ x' \in \tilde{T}_j \text{ with } i \neq j, i, j \in I, \end{cases}$$

where  $\sqcup$  denotes the disjoint union. By construction T' is a tree rooted at  $\varrho = \varrho_1$ , the root of  $\tilde{T}_1$ . Because  $\mathcal{M}$  is finite on bounded sets of E, it is not difficult to check that T' is locally compact. It is easy to see that the equivalence class of  $\operatorname{Tree}(\mathcal{M}) = (T', \tilde{T}_1)$  in  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$  does not depend of the choice of the representatives in the equivalence classes of  $T_1$  and  $T_i$  for  $i \in I$ . Hence, identifying  $\operatorname{Tree}(\mathcal{M})$  with its equivalence class, we get that the map Tree is well defined from  $\mathbb{M}(E)$  into  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$ .

**Lemma 5.31** (Regularity of the map Tree). The map Tree from  $\mathbb{M}(E)$  to  $\mathbb{T}^{[2]}_{\text{loc}-K}$  (or  $\mathbb{T}^{\text{spine}}_{\text{loc}-K}$ ) is continuous.

Proof. We only give the principal arguments of the proof. Let  $(\mathcal{M}_n, n \in \mathbb{N})$  a sequence of point measures, elements of  $\mathbb{M}(E)$ , which converges to  $\mathcal{M}$ . Let  $\varepsilon > 0$  be fixed such that  $\mathcal{M}(\partial B_E(\varepsilon)) =$ 0. For *n* large enough, we have  $\mathcal{M}_n(B_E(\varepsilon)) = \mathcal{M}(B_E(\varepsilon))$  and the atoms of  $\mathcal{M}_n$  in  $B_E(\varepsilon)$ converge to the atoms of  $\mathcal{M}$  in  $B_E(\varepsilon)$ . Using correspondence between the representations of the atoms, and similar arguments as in the proof of Lemma 5.13, we deduce that the distance between  $\operatorname{Tree}(\mathcal{M}_n)$  and  $\operatorname{Tree}(\mathcal{M})$  (in  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$ ) is small if  $\varepsilon > 0$  is small (to prove this statement in detail, one can use the distance on  $\mathbb{M}(E)$  given in [13, Equation (A2.6.1)]). This means that  $\lim_{n\to\infty} d_{\operatorname{LGH}}^{[2]}(\operatorname{Tree}(\mathcal{M}_n), \operatorname{Tree}(\mathcal{M})) = 0$ , and thus the map Tree is continuous on  $\mathbb{T}_{\operatorname{loc}-K}^{[2]}$ .

We shall now prove that the restriction of the map Tree to a subset of  $\mathbb{M}(E)$  is injective and bi-measurable. For this reason, we consider the subset of  $\mathbb{T}_{loc-K}$  of (equivalence classes of) trees not reduced to its root and such that the root is not a branching vertex (recall Definitions (61) and (35) with n = 0):

(62) 
$$\mathbb{T}^{0,*}_{\mathrm{loc}-\mathrm{K}} = \mathbb{T}^*_{\mathrm{loc}-\mathrm{K}} \cap \mathbb{T}^0_{\mathrm{loc}-\mathrm{K}}$$

As a direct consequence of Lemma 5.9,  $\mathbb{T}^{0,*}_{loc-K}$  is a Borel subset of  $\mathbb{T}_{loc-K}$  and thus of  $\mathbb{T}^*_{loc-K}$ . In particular, the following subset of  $\mathbb{M}(E)$  is a Borel set:

$$\tilde{\mathbb{M}}(E) = \Big\{ \mathcal{M} \in \mathbb{M}(E) : \mathcal{M}\big(\mathbb{R}_+ \times (\mathbb{T}^{0,*}_{\mathrm{loc}-\mathrm{K}})^c\big) = 0 \Big\}.$$

We now introduce a map  $\mathcal{M}$  from  $\mathbb{T}_{loc-K}^{spine}$  to  $\mathbb{M}(E)$  as follow. Let  $T^* = (T, T_1)$  be a rooted locally compact tree with an infinite marked spine. In particular, we have  $T_1 \subset T$  and  $T_1$  is equivalent to  $([0,\infty), d, 0)$ . Let  $(T_i^\circ, i \in I)$  be the family of the connected components of  $T \setminus T_1$ . For every  $i \in I$ , let us denote by  $x_i$  the MRCA of  $T_i^\circ$ , that is, the unique point of  $T_1$  such that for every  $x \in T_i^{\circ}$ ,  $[\![\varrho, x]\!] \cap T_1 = [\![\varrho, x_i]\!]$ . We then set  $T_i = T_i^{\circ} \cup \{x_i\}$  viewed as a locally compact tree rooted at  $x_i$ . Then, we define the point measure  $\mathcal{M}(T^*)$  on  $\mathbb{R}_+ \times \mathbb{T}^*_{\text{loc}-K} \subset \mathbb{R}_+ \times \mathbb{T}_{\text{loc}-K}$  by:

$$\mathcal{M}(T^*) = \sum_{i \in I} \delta_{(H(x_i), T_i)}.$$

As  $\mathcal{M}(T^*)$  does not depends on the representatives chosen in the equivalence class of  $T^*$  in  $\mathbb{T}^{\text{spine}}_{\text{loc}-\text{K}}$ , we deduce that  $\mathcal{M}: T^* \mapsto \mathcal{M}(T^*)$  is a map from  $\mathbb{T}^{\text{spine}}_{\text{loc}-\text{K}}$  to  $\mathbb{M}(E)$ . We now give the main result of this section.

**Proposition 5.32** (Regularity of the maps Tree and  $\mathcal{M}$ ). The map  $\mathcal{M}$  is bi-measurable from  $\mathbb{T}^{\text{spine}}_{\text{loc}-K}$  to  $\tilde{\mathbb{M}}(E)$  with  $\tilde{\mathbb{M}}(E) = \text{Im}(\mathcal{M})$ . The map Tree is bi-measurable from  $\tilde{\mathbb{M}}(E)$  to  $\mathbb{T}^{\text{spine}}_{\text{loc}-K}$ . Furthermore, the map Tree  $\circ \mathcal{M}$  is the identity map on  $\mathbb{T}^{\text{spine}}_{\text{loc}-K}$  and  $\mathcal{M} \circ \text{Tree}$  is the identity map on  $\tilde{\mathbb{M}}(E)$ .

*Proof.* By construction, the roots of all the trees  $T_i$  in the point measure  $\mathcal{M}(T^*)$  are not branching vertices, so that  $\mathcal{M}(T^*)$  belongs to  $\tilde{\mathbb{M}}(E) \subset \mathbb{M}(E)$ . We also get by construction that  $\operatorname{Tree}(\mathcal{M}(T^*)) = T^*$ . This implies that  $\mathcal{M}$  is injective and thus bi-measurable thanks to Lusin's theorem [32].

We also have by construction that  $\mathcal{M} \circ \text{Tree}(\mathcal{M}) = \mathcal{M}$  for  $\mathcal{M} \in \tilde{\mathbb{M}}(E)$ . This implies that  $\text{Im}(\mathcal{M}) = \tilde{\mathbb{M}}(E)$  and also that Tree restricted to  $\tilde{\mathbb{M}}(E)$  is injective and thus bi-measurable thanks to Lusin's theorem.

We extend the map  $T^* \mapsto \mathcal{M}(T^*)$  to  $\mathbb{T}^{(1)}_{\text{loc}-K}$  in the following way. For  $(T, \mathbf{v} = (\varrho, v_1)) \in \mathbb{T}^{(1)}_{\text{loc}-K}$ , we graft the infinite spine  $T_1$  on  $v_1$  and consider the rooted locally compact tree with an infinite marked spine  $\operatorname{Sp}(T) \in \mathbb{T}^{\text{spine}}_{\text{loc}-K}$  defined in Remark 5.30. Then, we define  $\mathcal{M}(T, \mathbf{v})$  as  $\mathcal{M}(\operatorname{Sp}(T))$ . From the continuity of the grafting procedure, see Lemma 5.13 and the continuity of  $\Pi_1^\circ$ , see Lemma 5.5, and the measurability of the map  $\mathcal{M}$ , we deduce that the map  $(T, \mathbf{v}) \mapsto \mathcal{M}(T, \mathbf{v})$ , which we still denote by  $\mathcal{M}$  is measurable. In fact, we have the stronger following result. Consider the set of (equivalence class of) *n*-pointed rooted locally compact tree such that the root is not a branching vertex and the pointed vertices are not equal to the root:

(63) 
$$\mathbb{T}_{\text{loc}-K}^{(n),0,*} = \left\{ (T, \mathbf{v}) \in \mathbb{T}_{\text{loc}-K}^{(n),0} : d(\varrho, v_i) > 0 \text{ for all } i \in \{1, \dots, n\} \right\},$$

where  $\mathbf{v} = (\varrho, v_1, \ldots, v_n)$ . According to Lemma 5.9 and Remark 5.11, the set  $\mathbb{T}_{loc-K}^{(n),0,*}$  is a Borel subset of  $\mathbb{T}_{loc-K}^{(n)}$ . Recall from (62) that the Borel set  $\mathbb{T}_{loc-K}^{0,*}$  is the set of (equivalence class of) 1-pointed rooted locally compact trees such that the root is not a branching vertex and the pointed vertex is not equal to the root.

**Corollary 5.33** (Recovering  $(T, \mathbf{v})$  from  $\mathcal{M}(T, \mathbf{v})$ ). The following map from  $\mathbb{T}^{(1)}_{\text{loc}-K}$  to  $\mathbb{R}_+ \times \mathbb{M}(E)$  defined by:

$$(T, \mathbf{v}) \mapsto (d(\varrho, v), \mathcal{M}(T, \mathbf{v}))$$

is measurable and its restriction to  $\mathbb{T}^{(1),0,*}_{loc-K}$  is injective and bi-measurable.

Proof. Set  $\mathbb{M}^*(E) = \{\mathcal{M} \in \mathbb{M}(E) : \mathcal{M}(\{0\} \times \mathbb{T}^*_{\text{loc}-K}) = 0\}$ . For  $\mathcal{M} \in \mathbb{M}^*(E)$ , we get that  $\text{Tree}(\mathcal{M})$  belongs to  $\mathbb{T}^{\text{spine},0}_{\text{loc}-K}$ . So, we can define the map g on  $\mathbb{R}_+ \times \mathbb{M}^*(E)$  by  $g(a, \mathcal{M}) = \text{Graft}_1([0, a], \text{Tree}(\mathcal{M}))$ , where the tree [0, a] has root 0 and pointed vertex a. Thanks to the continuity of the grafting procedure, see Lemma 5.28 and of the function Tree, see Lemma 5.31, we deduce that g is continuous.

Let  $(T, \mathbf{v}) \in \mathbb{T}_{\text{loc}-K}^{(1),0,*}$ . As the root of T is not a branching vertex, we get that  $\mathcal{M}(T, \mathbf{v})$  belongs to  $\mathbb{M}^*(E)$ , and thus  $g(d(\varrho, v), \mathcal{M}(T, \mathbf{v}))$ , where  $\mathbf{v} = (\varrho, v)$ , is well defined and in fact equal to  $(T, \mathbf{v})$  thanks to (60) with n = 1. This implies that the map  $(T, \mathbf{v}) \mapsto (d(\varrho, v), \mathcal{M}(T, \mathbf{v}))$  defined on  $\mathbb{T}_{\text{loc}-K}^{(1),0,*}$  is injective, and thus bi-measurable by Lusin's theorem [32].

We extend this result to *n*-pointed trees. Recall from (55) that, for  $(T, \mathbf{v}) \in \mathbb{T}_{loc-K}^{(n)}$ , we have  $\operatorname{Split}_n(T, \mathbf{v}) = \left(\hat{T}_A(T, \mathbf{v}), A \in \mathcal{P}_n\right)$  and set  $\mathcal{M}_A[T, \mathbf{v}] = \mathcal{M}(\hat{T}_A(T, \mathbf{v}))$  for  $A \in \mathcal{P}_n^+$ .

**Corollary 5.34** (Recovering  $(T, \mathbf{v})$  from the  $\mathcal{M}_A[T, \mathbf{v}]$ ). The following map from  $\mathbb{T}^{(n)}_{\text{loc}-K}$  to  $\mathbb{T}^{(n)}_{\text{dis}} \times \mathbb{M}(E)^{\mathcal{P}^+_n}$  defined by:

$$(T, \mathbf{v}) \mapsto \left( \operatorname{Span}_n(T, \mathbf{v}), \left( \mathcal{M}_A[T, \mathbf{v}], A \in \mathcal{P}_n^+ \right) \right)$$

is measurable and its restriction to  $\mathbb{T}^{(n),0,*}_{loc-K}$  is injective and bi-measurable.

*Proof.* Using the measurability of the functions Span from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(n)}$  (see Lemma 5.6),  $\mathbf{L}_n$  from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{R}_+^{\mathcal{P}_n^+}$  (see Lemma 5.25), Split<sub>n</sub> from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\left(\mathbb{T}_{loc-K}^{(1)}\right)^{2^n}$  (see Lemma 5.26) and the map  $(T, \mathbf{v}) \mapsto \mathcal{M}(T, \mathbf{v})$  from  $\mathbb{T}_{loc-K}^{(1)}$  to  $\mathbb{M}(E)$  (see Corollary 5.33), we deduce that the following map, say  $g_1$ , from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(n)} \times (\mathbb{R}_+ \times \mathbb{M}(E))^{\mathcal{P}_n^+}$  is measurable:

$$g_1: (T, \mathbf{v}) \mapsto \left( \operatorname{Span}(T, \mathbf{v}), \left( (\ell_A(T, \mathbf{v}), \mathcal{M}_A[T, \mathbf{v}]), A \in \mathcal{P}_n^+ \right) \right).$$

Notice that  $(T, \mathbf{v}) \in \mathbb{T}_{loc-K}^{(n),0,*}$  implies that  $\hat{T}_{\{0\}}$  is reduced to its root. Using the measurable functions  $\operatorname{Graft}_n$  and the map defined in Corollary 5.33, we easily deduce that  $g_1$  restricted to  $\mathbb{T}_{loc-K}^{(n),0,*}$  is injective and thus bi-measurable by Lusin's theorem [32]. Since  $\mathbf{L}_n(T, \mathbf{v})$  is also equal to  $\mathbf{L}_n(\operatorname{Span}(T, \mathbf{v}))$ , we deduce that the following map  $g_2$ , from  $\mathbb{T}_{loc-K}^{(n)}$  to  $\mathbb{T}_{loc-K}^{(n)} \times \mathbb{M}(E)^{\mathcal{P}_n^+}$  is measurable:

$$g_2: (T, \mathbf{v}) \mapsto \left( \operatorname{Span}(T, \mathbf{v}), \left( \mathcal{M}_A[T, \mathbf{v}], A \in \mathcal{P}_n^+ \right) \right)$$

Furthermore, its restriction to  $\mathbb{T}_{loc-K}^{(n),0,*}$  is also injective and thus bi-measurable.

#### 

#### 6. BACKBONE DECOMPOSITION

#### 6.1. A discrete random tree constructed by successive grafts.

6.1.1. A random tree. In this section, for  $a \ge 0$ , we denote by  $([0, a], (\rho = 0, a)) \in \mathbb{T}^{(1)}_{\text{dis}}$  the (equivalent class of the) tree [0, a] endowed with the usual distance on  $\mathbb{R}$ , rooted at  $\rho = 0$  and pointed at a; and when there is no possible confusion we simply denote it by [0, a].

Let t > 0 and let  $\nu$  be a probability measure on [0, t]. Let  $\xi = (\xi_k, k \in \mathbb{N}^*)$  be a sequence of independent random variables with distribution  $\nu$  and let  $((K_k, \varepsilon_k), k \in \mathbb{N}^*)$  be a sequence of independent random variables independent of the sequence  $\xi$ , with  $K_k$  uniformly distributed on  $\{1, \ldots, k\}$  independent of  $\varepsilon_k$  uniformly distributed on  $\{g, d\}$ . For every integer  $n \ge 2$ , we set  $(\xi_1^{(n)}, \ldots, \xi_{n-1}^{(n)})$  the increasing order statistic of  $(\xi_1, \ldots, \xi_{n-1})$ . Then we define the family of pointed trees  $((\mathbf{T}_1^{(n)}, \mathbf{v}_1^{(n)}), \ldots, (\mathbf{T}_n^{(n)}, \mathbf{v}_n^{(n)}))$ , with  $(\mathbf{T}_k^{(n)}, \mathbf{v}_k^{(n)}) \in \mathbb{T}_{\text{dis}}^{(k)}$ , recursively by:  $\mathbf{T}_1^{(n)} = [0, t]$ , that is,  $(\mathbf{T}_1^{(n)}, \mathbf{v}_1^{(n)}) = ([0, t], (0, t)) \in \mathbb{T}_{\text{dis}}^{(k)} \subset \mathbb{T}_{\text{loc}-K}^{(1)}$ . For every k ∈ {1,..., n − 1}, conditionally given the random variable (**T**<sup>(n)</sup><sub>k</sub>, **v**<sup>(n)</sup><sub>k</sub>) in **T**<sup>(k)</sup><sub>dis</sub>, we define the **T**<sup>(k+1)</sup><sub>loc-K</sub>-valued random variable (**T**<sup>(n)</sup><sub>k+1</sub>, **v**<sup>(n)</sup><sub>k+1</sub>) by grafting a branch of length t − ξ<sup>(n)</sup><sub>k+1</sub> uniformly on the left or on the right of a uniformly chosen vertex among the k vertices of **T**<sup>(n)</sup><sub>k</sub> at level ξ<sup>(n)</sup><sub>k+1</sub>, and the new leaf (which is, as all the other leaves, at level t) is added to the vector recording the pointed vertices. Formally, using the grafting procedure (39) from Remark 5.14, we set:

(64) 
$$(\mathbf{T}_{k+1}^{(n)}, \mathbf{v}_{k+1}^{(n)}) = (\mathbf{T}_{k}^{(n)}, \mathbf{v}_{k}^{(n)}) \circledast_{K_{k}, \xi_{k+1}^{(n)}}^{\varepsilon_{k+1}} \left[ 0, t - \xi_{k+1}^{(n)} \right]$$

According to Lemma 5.15 the grafting procedure is continuous and thus measurable, we deduce that  $(\mathbf{T}_k^{(n)}, \mathbf{v}_k^{(n)})$  is a  $\mathbb{T}_{\mathbf{K}}^{(k)}$ -valued random variable for every  $1 \le k \le n$ .

By construction, we also get that  $(\mathbf{T}_k^{(n)}, \mathbf{v}_k^{(n)})$  belongs to  $\mathbb{T}_{dis}^{(k)}$ , and is thus a  $\mathbb{T}_{dis}^{(k)}$ -valued random variable, for all  $k \in \{1, \ldots, n\}$ . To simplify the notations, we set  $\mathbf{T}_n = (\mathbf{T}_n^{(n)}, \mathbf{v}_n^{(n)})$  for the  $\mathbb{T}_K^{(n)}$ -valued (and  $\mathbb{T}_{dis}^{(n)}$ -valued) random variable. By construction, we also get that  $\mathbf{T}_n$  is a planar tree with the leaves  $v_1, \ldots, v_n$  ranked from left to right, where  $\mathbf{v}_n^{(n)} = (v_0 = \varrho, \ldots, v_n)$ .

Recall that for a rooted tree T,  $\mathscr{L}^T$  denotes its length measure; and we simply write  $\mathscr{L}$  when there is no ambiguity. Informally, for  $x \in T$  chosen according to  $\mathscr{L}(dx)$ , we denote by  $T \circledast_x [0, a]$ the tree T on which a segment of length a is grafted at the vertex x. It is possible to make this construction in a measurable way, in this direction see for example [8, Equation (1.6)] and Section 3 therein. As we shall consider the discrete tree  $\mathbf{T}_n$  for T, we refer to Remark 6.2 below for a direct proof. The next lemma relates the distributions of  $\mathbf{T}_n$  and of  $\mathbf{T}_{n+1}$ ; its proof is given in the next section.

**Lemma 6.1.** Let  $t \ge 0$ . Assume that the probability distribution  $\nu$  has a positive density  $f_{\text{dens}}$  with respect to the Lebesgue measure on [0, t]. For  $n \in \mathbb{N}^*$ , G a measurable non-negative function defined on  $\mathbb{T}_{\text{dis}}^{(n+1)}$ , and  $\varepsilon$  a  $\{g, d\}$ -valued uniform random variable independent of  $\mathbf{T}_n$ , we have:

(65) 
$$\mathbb{E}\left[\int_{\mathbf{T}_n} \mathscr{L}(\mathrm{d}x) f_{\mathrm{dens}}(H(x)) G\left(\mathbf{T}_n \circledast_x^{\varepsilon} \left[0, t - H(x)\right]\right)\right] = \frac{n+1}{2} \mathbb{E}\left[G\left(\mathbf{T}_{n+1}\right)\right]$$

Remark 6.2. We comment on the left-hand side of (65), and more precisely we check that the integral  $\mathcal{I} = \int_{\mathbf{T}_n} \mathscr{L}(\mathrm{d}x) f_{\mathrm{dens}}(H(x)) G(\mathbf{T}_n \circledast_x^{\varepsilon} [0, t - H(x)])$  is a non-negative random variable. Recall that  $\mathbf{T}_n = (\mathbf{T}_n^{(n)}, \mathbf{v}_n^{(n)})$ . Then, we can write  $\mathcal{I}$  as follows:

$$\mathcal{I} = \sum_{k=1}^{n} \int_{\xi_{k-1}^{(n)}}^{t} \mathrm{d}h \, f_{\mathrm{dens}}(h) \, G\big(\mathbf{T}_{n} \circledast_{k,h}^{\varepsilon} [0,t-h]\big),$$

with the convention that  $\xi_0^{(n)} = 0$ . Therefore, using the continuity of the grafting function, see Lemma 5.15, we obtain that  $\mathcal{I}$  is a non-negative real-valued random variable, and thus its expectation is well defined.

6.1.2. Proof of Lemma 6.1. The proof is based on two technical lemmas. We first consider the case t = 1 and  $\nu$  the uniform distribution on [0, 1]. Let us denote by  $\mathbf{T}_n^{\text{unif}}$  for  $\mathbf{T}_n$  when  $\nu$  is the uniform distribution on [0, 1].

**Lemma 6.3.** For  $n \in \mathbb{N}^*$ , G a measurable non-negative functional defined on  $\mathbb{T}_{dis}^{(n+1)}$ , and  $\varepsilon$  a  $\{g,d\}$ -valued uniform random variable independent of  $\mathbf{T}_n^{unif}$ , we have:

(66) 
$$\mathbb{E}\left[\int_{\mathbf{T}_{n}^{\text{unif}}} \mathscr{L}(\mathrm{d}x) G\left(\mathbf{T}_{n}^{\text{unif}} \circledast_{x}^{\varepsilon} \left[0, t - H(x)\right]\right)\right] = \frac{n+1}{2} \mathbb{E}\left[G\left(\mathbf{T}_{n+1}^{\text{unif}}\right)\right].$$

*Remark* 6.4. From (66), we see  $\frac{n+1}{2}$  is just the mean length of  $\mathbf{T}_n^{\text{unif}}$ .

*Proof.* To simplify notation, we write  $\mathbf{T}_n$  for  $\mathbf{T}_n^{\text{unif}}$ . We give a proof by induction. For n = 1, this is a direct consequence of the construction of  $\mathbf{T}_2 = \mathbf{T}_2^{(2)}$  from  $\mathbf{T}_1^{(2)} = \mathbf{T}_1 = [0, 1]$  given by (64).

Let  $n \in \mathbb{N}^*$  and assume that (66) holds for n replaced by any  $k \in \{1, \ldots, n-1\}$ . We will use for the proof a special representation of planar binary trees. Let T be a compact planar binary tree rooted at  $\rho$ , with all leaves at height 1. Let us denote by x its lowest branching vertex, set h = H(x) and set  $\tilde{T}^g$  (resp.  $\tilde{T}^d$ ) the left (resp. right) sub-tree above x. In our settings, we have:

$$h = H(x)$$
 and  $T = \left(\llbracket \varrho, x \rrbracket \circledast_x \tilde{T}^{\mathrm{g}}\right) \circledast_x^{\mathrm{d}} \tilde{T}^{\mathrm{d}} = \left(\llbracket \varrho, x \rrbracket \circledast_x \tilde{T}^{\mathrm{d}}\right) \circledast_x^{\mathrm{g}} \tilde{T}^{\mathrm{g}}$ 

where one remove the vertex x from the pointed vertices after the graftings. For convenience, we consider the scaled left and right trees  $T^{g} = (1-h)^{-1}\tilde{T}^{g}$  and  $T^{d} = (1-h)^{-1}\tilde{T}^{d}$ , so that  $T^{g}$  and  $T^{d}$  are rooted bounded binary planar trees with all their leaves at height 1. We call  $(h, T^{g}, T^{d})$  the decomposition of T according to its lowest branching vertex.

Let  $(\xi_1^{(n+1)}, \mathbf{T}_{n+1}^{g}, \mathbf{T}_{n+1}^{d})$  be the decomposition of  $\mathbf{T}_{n+1}$  according to its lowest branching vertex (which is indeed at height  $\xi_1^{(n+1)}$  by construction). Denote by  $I_{n+1}$  the number of leaves of  $\mathbf{T}_{n+1}^{g}$ . Using a Pólya urn starting with two balls of color g and d, we get that, by construction,  $I_{n+1}$ is the number of balls of color g in the urn after n draws. Thus  $I_{n+1}$  is uniform on  $\{1, \ldots, n\}$ and independent of  $\xi_1^{(n+1)}$ . Notice that if U is a uniform random variable on [0, 1], for every  $h \in (0, 1)$ , conditionally given  $\{U \ge h\}$ , the random variable  $(1 - h)^{-1}(U - h)$  is still uniformly distributed on [0, 1]. This gives that, conditionally on  $\{\xi_1^{(n+1)} = h\}$  and  $\{I_{n+1} = i\}$ , the two trees  $\mathbf{T}_{n+1}^{g}$  and  $\mathbf{T}_{n+1}^{d}$  are independent and distributed respectively as  $\mathbf{T}_i$  and  $\mathbf{T}_{n+1-i}$ .

We consider a measurable non-negative functional G defined on the space of rooted compact binary planar trees with a finite number of leaves, all of them at height 1 of the form:

(67) 
$$G(T) = g_1(h) g_2(T^{\rm g}) g_3(T^{\rm d}),$$

where the  $g_i$ 's are measurable non-negative functionals and  $(h, T^g, T^d)$  is the decomposition of T according to its lowest branching vertex. Setting  $f_j(i) = \mathbb{E}[g_j(\mathbf{T}_i)]$  for  $j \in \{2, 3\}$ , we have since  $\xi_1^{(n+1)}$  is distributed according to a  $\beta(1, n)$  distribution:

(68) 
$$\mathbb{E}\left[G(\mathbf{T}_{n+1})\right] = \left(\int_0^1 g_1(h) n(1-h)^{n-1} \,\mathrm{d}h\right) \frac{1}{n} \sum_{i=1}^n f_2(i) f_3(n+1-i)$$
$$= \left(\int_0^1 g_1(h) (1-h)^{n-1} \,\mathrm{d}h\right) \sum_{i=1}^n f_2(i) f_3(n+1-i).$$

On the other hand, let  $(\xi_1^{(n)}, \mathbf{T}_n^{\mathrm{g}}, \mathbf{T}_n^{\mathrm{d}})$  be the decomposition of  $\mathbf{T}_n$  according to its lowest branching vertex. Let  $x \in \mathbf{T}_n$  and set h = H(x).

• If  $h < \xi_1^{(n)}$ , the decomposition of  $\mathbf{T}_n \otimes_x^g [0, 1-h]$  according to its lowest branching vertex is given by  $(h, [0, 1], (1-h)^{-1}\mathbf{T}'_n)$  where  $\mathbf{T}'_n$  is as the tree  $\mathbf{T}_n$  but for its lowest branch whose length is  $\xi_1^{(n)} - h$  instead of  $\xi_1^{(n)}$ . Notice that the shapes of the tree  $\mathbf{T}'_n$  and  $\mathbf{T}_n$ are the same. Then using again the property of conditioned uniform random variables, we deduce that conditionally on  $\{\xi_1^{(n)} \ge h\}$ , the tree  $(1-h)^{-1}\mathbf{T}'_n$  is distributed as  $\mathbf{T}_n$ . Thus, we get:

(69) 
$$\mathbb{E}\left[\int_{\mathbf{T}_n} \mathbf{1}_{\{H(x)<\xi_1^{(n)}\}} \mathscr{L}(\mathrm{d}x) G(\mathbf{T}_n \circledast_x^{\mathrm{g}} [0,1-h])\right] = \mathbb{E}\left[\int_0^{\xi_1^{(n)}} g_1(h) \,\mathrm{d}h\right] f_2(1) f_3(n).$$

By symmetry, we have:

(70) 
$$\mathbb{E}\left[\int_{\mathbf{T}_n} \mathbf{1}_{\{H(x)<\xi_1^{(n)}\}} \mathscr{L}(\mathrm{d}x) G(\mathbf{T}_n \circledast_x^{\mathrm{d}} [0,1-h])\right] = \mathbb{E}\left[\int_0^{\xi_1^{(n)}} g_1(h) \,\mathrm{d}h\right] f_2(n) f_3(1).$$

• For  $x \in \tilde{\mathbf{T}}_n^{\mathrm{g}}$ , the decomposition of  $\mathbf{T}_n \circledast_x^{\varepsilon} [0, 1-h]$  according to its lowest branching vertex is given by  $(\xi_1^{(n)}, (1-h)^{-1}\mathcal{T}', \mathbf{T}_n^{\mathrm{d}})$ , where  $\mathcal{T}' = \tilde{\mathbf{T}}_n^{\mathrm{g}} \circledast_x^{\varepsilon} [0, 1-h]$ . Notice that the length measure on the tree  $\mathbf{T}_n^{\mathrm{g}}$  is obtained by scaling by  $(1-\xi_1^{(n)})^{-1}$  the length measure on  $\mathbf{T}_n$ restricted to  $\tilde{\mathbf{T}}_n^{\mathrm{g}}$ . We deduce that:

$$\begin{split} \mathbb{E}\left[\int_{\mathbf{T}_{n}} \mathbf{1}_{\{x\in\tilde{\mathbf{T}}_{n}^{g}\}} \mathscr{L}^{\mathbf{T}_{n}}(\mathrm{d}x) G\left(\mathbf{T}_{n} \circledast_{x}^{\varepsilon} [0, 1-h]\right)\right] \\ &= \mathbb{E}\left[g_{1}(\xi_{1}^{(n)})g_{3}(\mathbf{T}_{n}^{d}) \int_{\mathbf{T}_{n}} \mathbf{1}_{\{x\in\tilde{\mathbf{T}}_{n}^{g}\}} \mathscr{L}^{\mathbf{T}_{n}}(\mathrm{d}x) g_{2}\left((1-\xi_{1}^{(n)})^{-1}\left(\tilde{\mathbf{T}}_{n}^{g} \circledast_{x}^{\varepsilon} [0, 1-H(x)]\right)\right)\right)\right] \\ &= \mathbb{E}\left[g_{1}(\xi_{1}^{(n)})g_{3}(\mathbf{T}_{n}^{d}) \int_{\mathbf{T}_{n}^{g}} (1-\xi_{1}^{(n)}) \mathscr{L}^{\mathbf{T}_{n}^{g}}(\mathrm{d}y) g_{2}\left(\mathbf{T}_{n}^{g} \circledast_{y}^{\varepsilon} [0, 1-H(y)]\right)\right] \\ &= \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] \frac{1}{n-1} \sum_{i=1}^{n-1} f_{3}(n-i) \mathbb{E}\left[\int_{\mathbf{T}_{i}} \mathscr{L}^{\mathbf{T}_{i}}(\mathrm{d}y) g_{2}\left(\mathbf{T}_{i} \circledast_{y}^{\varepsilon} [0, 1-H(y)]\right)\right]. \end{split}$$

where we used the distribution of  $(\mathbf{T}_n^{\mathrm{g}}, \mathbf{T}_n^{\mathrm{d}})$  conditionally on  $\xi_1^{(n)}$  and  $I_n$  for the last equality. Using that, by induction, (66) holds for n = i, we get:

(71) 
$$\mathbb{E}\left[\int_{\mathbf{T}_{n}} \mathbf{1}_{\{x \in \tilde{\mathbf{T}}_{n}^{g}\}} \mathscr{L}(\mathrm{d}x) G(\mathbf{T}_{n} \circledast_{x}^{\varepsilon} [0, 1-h])\right]$$
$$= \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] \frac{1}{n-1} \sum_{i=1}^{n-1} \frac{i+1}{2} f_{2}(i+1) f_{3}(n-i)$$
$$= \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] \frac{1}{n-1} \sum_{i=2}^{n} \frac{i}{2} f_{2}(i) f_{3}(n-i+1).$$

• By symmetry, for  $x \in \tilde{\mathbf{T}}_n^d$ , we get:

(72) 
$$\mathbb{E}\left[\int_{\mathbf{T}_{n}} \mathbf{1}_{\{x \in \tilde{\mathbf{T}}_{n}^{d}\}} \mathscr{L}(\mathrm{d}x) G(\mathbf{T}_{n} \circledast_{x}^{\varepsilon} [0, 1-h])\right]$$
$$= \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] \frac{1}{n-1} \sum_{i=2}^{n} \frac{i}{2} f_{3}(i) f_{2}(n-i+1)$$
$$= \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] \frac{1}{n-1} \sum_{i=1}^{n-1} \frac{n-i+1}{2} f_{2}(i) f_{3}(n-i+1).$$

Summing (69) times  $\mathbb{P}(\varepsilon = g) = 1/2$ , (70) times  $\mathbb{P}(\varepsilon = d) = 1/2$ , (71) and (72), and using that  $\xi_1^{(n)}$  has distribution  $\beta(1, n - 1)$  so that:

$$\mathbb{E}\left[\int_{0}^{\xi_{1}^{(n)}} g_{1}(h) \,\mathrm{d}h\right] = \frac{1}{n-1} \mathbb{E}\left[(1-\xi_{1}^{(n)})g_{1}(\xi_{1}^{(n)})\right] = \int_{0}^{1} g_{1}(h)(1-h)^{n-1} \,\mathrm{d}h$$

we deduce that:

$$\mathbb{E}\left[\int_{\mathbf{T}_n} \mathscr{L}(\mathrm{d}x) G\left(\mathbf{T}_n \circledast_x^{\varepsilon} \left[0, 1 - H(x)\right]\right)\right]$$
$$= \left(\int_0^1 g_1(h)(1-h)^{n-1} \mathrm{d}h\right) \sum_{i=1}^n \frac{n+1}{2} f_2(i) f_3(n+1-i).$$

Thanks to (68), we deduce that (66) holds for G given by (67). Then use a monotone class argument to conclude that (66) holds for any measurable non-negative G. This concludes the proof by induction.

We now consider  $t \ge 0$  and assume that the probability distribution  $\nu$  has a positive density  $f_{\text{dens}}$  with respect to the Lebesgue measure on [0, t]. Let F denote the cumulative distribution function of  $\nu$ . By the assumptions on  $f_{\text{dens}}$ , F is one to one from [0, t] onto [0, 1] and its inverse  $F^{-1}$  is continuous. For a compact rooted real tree  $(T, d, \varrho)$ , we define:

$$\forall x \in T, \ H^{f_{\text{dens}}}(x) = F^{-1}(H(x)),$$
  
$$\forall x, y \in T, \ d^{f_{\text{dens}}}(x, y) = H^{f_{\text{dens}}}(x) + H^{f_{\text{dens}}}(y) - 2H^{f_{\text{dens}}}(x \wedge y).$$

The scaling map  $R^{f_{\text{dens}}}: (T, d, \varrho) \longmapsto (T, d^{f_{\text{dens}}}, \varrho)$  is then well-defined from  $\{T \in \mathbb{T}_{\mathrm{K}}: H(T) \leq 1\}$  to  $\mathbb{T}_{\mathrm{K}}$ . We shall now prove it is continuous.

**Lemma 6.5.** The map  $R^{f_{\text{dens}}}$  from  $\{T \in \mathbb{T}_{K}, H(T) \leq 1\}$  to  $\mathbb{T}_{K}$  is continuous.

*Proof.* Let  $\varepsilon > 0$ . As  $F^{-1}$  is uniformly continuous with our assumptions, there exists  $\delta > 0$  such that, for every  $x, y \in [0, 1]$ :

$$|x-y| < \delta \Longrightarrow \left| F^{-1}(x) - F^{-1}(y) \right| \le \frac{\varepsilon}{2}.$$

Let  $T, T' \in \mathbb{T}_{\mathrm{K}}$  with  $H(T) \leq 1$  and  $H(T') \leq 1$  such that  $d_{\mathrm{GH}}(T, T') < \delta/8$ . Then, there exists a correspondence  $\mathcal{R}$  between (elements in the equivalence classes) T and T' such that dist  $(\mathcal{R}) \leq 2d_{\mathrm{GH}}(T, T') + \delta/4 < \delta/2$ .

For every  $(x, x'), (y, y') \in \mathcal{R}$ , we have:

$$\begin{aligned} \left| d^{f_{\text{dens}}}(x,y) - d'^{f_{\text{dens}}}(x',y') \right| &= \left| H^{f_{\text{dens}}}(x) + H^{f_{\text{dens}}}(y) - 2H^{f_{\text{dens}}}(x \wedge y) - H^{f_{\text{dens}}}(x') \\ &- H^{f_{\text{dens}}}(y') + 2H^{f_{\text{dens}}}(x' \wedge y') \right| \\ &\leq \left| F^{-1}(H(x)) - F^{-1}(H(x')) \right| + \left| F^{-1}(H(y)) - F^{-1}(H(y')) \right| \\ &+ 2 \left| F^{-1}(H(x \wedge y)) - F^{-1}(H(x' \wedge y')) \right|. \end{aligned}$$

As  $(x, x') \in \mathcal{R}$ , we have  $|H(x) - H(x')| \leq \text{dist } (\mathcal{R}) < \delta$  and consequently,  $|F^{-1}(H(x)) - F^{-1}(H(x'))| < \varepsilon/2$ . Similarly, we have  $|F^{-1}(H(y)) - F^{-1}(H(y'))| < \varepsilon/2$ . We also have:

$$\begin{aligned} \left| H(x \wedge y) - H(x \wedge y) \right| &= \frac{1}{2} \left| H(x) + H(y) - d(x, y) - H(x') - H(y') + d'(x', y') \right| \\ &\leq \frac{1}{2} \left| H(x) - H(x') \right| + \frac{1}{2} \left| H(y) - H(y') \right| + \frac{1}{2} \left| d(x, y) - d'(x', y') \right| \\ &\leq \frac{3}{2} \text{dist} \ (\mathcal{R}) \\ &< \delta. \end{aligned}$$

This gives  $\left|F^{-1}(H(x \wedge y)) - F^{-1}(H(x' \wedge y'))\right| < \varepsilon/2.$ 

To conclude, we have  $\operatorname{dist}^{f_{\operatorname{dens}}}(\mathcal{R}) < 2\varepsilon$  which implies that  $d_{\operatorname{GH}}^{f_{\operatorname{dens}}}(T,T') < \varepsilon$ . This gives the continuity of the map  $R^{f_{\operatorname{dens}}}$ .

We now prove Lemma 6.1. Recall that  $\mathbf{T}_n$  denotes the trees constructed with the probability measure  $\nu(\mathrm{d}x) = f_{\mathrm{dens}}(x) \,\mathrm{d}x$  and  $\mathbf{T}_n^{\mathrm{unif}}$  the trees constructed with the uniform distribution on [0,1] as studied in the first step. By construction, for all  $n \in \mathbb{N}^*$ , the random variables  $R^{f_{\mathrm{dens}}}(\mathbf{T}_n^{\mathrm{unif}})$  and  $\mathbf{T}_n$  have the same distribution. Notice also that, for every  $T \in \mathbb{T}_K$  and every non-negative measurable function g on  $\mathbb{R}_+ \times \mathbb{T}_{\mathrm{loc}-\mathrm{K}}$ , we have:

$$\int_{T} \mathscr{L}^{T}(\mathrm{d}y) g(H(y), T) = \int_{R^{f_{\mathrm{dens}}}(T)} \mathscr{L}^{R^{f_{\mathrm{dens}}}(T)}(\mathrm{d}x) f_{\mathrm{dens}}(H^{f_{\mathrm{dens}}}(x)) g(H^{f_{\mathrm{dens}}}(x), R^{f_{\mathrm{dens}}}(T)).$$

Let G be a measurable non-negative functional defined on the space of rooted compact binary planar trees with a finite number of leaves, all of them at height t. We first have:

$$\begin{split} & \mathbb{E}\left[\int_{\mathbf{T}_{n}}\mathscr{L}^{\mathbf{T}_{n}}(\mathrm{d}x)\,f_{\mathrm{dens}}\big(H(x)\big)\,G\Big(\mathbf{T}_{n}\,\circledast_{x}^{\varepsilon}\,\left[0,t-H(x)\right]\Big)\right] \\ &= \mathbb{E}\left[\int_{R^{f_{\mathrm{dens}}}(\mathbf{T}_{n}^{\mathrm{unif}})}\mathscr{L}^{R^{f_{\mathrm{dens}}}(\mathbf{T}_{n}^{\mathrm{unif}})}(\mathrm{d}x)\,f_{\mathrm{dens}}\big(H^{f_{\mathrm{dens}}}(x)\big)\,G\Big(R^{f_{\mathrm{dens}}}(\mathbf{T}_{n}^{\mathrm{unif}})\,\circledast_{x}^{\varepsilon}\,\left[0,t-H^{f_{\mathrm{dens}}}(x)\right]\Big)\right] \\ &= \mathbb{E}\left[\int_{\mathbf{T}_{n}^{\mathrm{unif}}}\mathscr{L}^{\mathbf{T}_{n}^{\mathrm{unif}}}(\mathrm{d}y)\,G\circ R^{f_{\mathrm{dens}}}\Big(\mathbf{T}_{n}^{\mathrm{unif}}\,\circledast_{y}^{\varepsilon}\,\left[0,1-H(y)\right]\Big)\right]. \end{split}$$

Applying Lemma 6.1, and then that  $R^{f_{\text{dens}}}(\mathbf{T}_{n+1}^{\text{unif}})$  and  $\mathbf{T}_{n+1}$  have the same distribution, we get the result.

6.1.3. An infinite tree with no leaves. Let  $f_{\text{int}}$  be a positive locally integrable function on  $[0, +\infty)$ . Let S be a Poisson point measure on  $\mathbb{R}_+$  with intensity  $f_{\text{int}}(t) \, \mathrm{d}t$ . We denote by  $(\xi_i, i \ge 1)$  the increasing sequence of the atoms of S and by N the process  $(N_t = S([0, t]), t \ge 0)$ .

Let  $(\varepsilon_n, n \ge 1)$  be independent random variables uniformly distributed on  $\{g, d\}$  and let  $(K_n, n \ge 1)$  be independent random variables uniformly distributed on  $\{1, 2, \ldots, n\}$  respectively, all these variables being independent and independent of S.

We define a tree-valued process  $(\mathfrak{T}_t, t \geq 0)$  where, for every  $t \geq 0$ , the random tree  $\mathfrak{T}_t$  has height t and  $N_t + 1$  leaves, all of them at this height t. Before going into this construction, we first define a function Growth<sub>n</sub> on rooted n-pointed trees. We define Growth<sub>n</sub>( $(T, \mathbf{v}), h$ ) as the tree obtained by grafting on all the pointed vertices of T a branch of length h and pointing the new leaves with the order naturally induced by  $\mathbf{v}$ . Set  $A_n = \{0, n + 1, \ldots, 2n\}$ . Formally, we have:

$$\operatorname{Growth}_{n}((T,\mathbf{v}),h) = \Pi_{2n}^{\circ,A_{n}}\left(\left((T,\mathbf{v})\circledast_{1}[0,h]\right)\circledast_{2}[0,h]\right)\cdots\circledast_{n}[0,h]\right)$$

where  $\Pi_{2n}^{\circ,A_n}$ , defined in (34), removes the first *n* pointed vertices: for  $(T, \mathbf{v}) \in \mathbb{T}_{loc-K}^{(2n)}$ , we have  $\Pi_{2n}^{\circ,A_n}(T,\mathbf{v}) = (T,\mathbf{v}')$ , with  $\mathbf{v} = (v_0 = \varrho, v_1, \ldots, v_{2n})$  and  $\mathbf{v}' = (v'_0 = \varrho, v'_1 = v_{n+1}, \ldots, v'_n = v_{2n})$ . Thanks to the continuity of the grafting procedure (see Lemma 5.15) and the continuity of  $\Pi_{2n}^{\circ,A_n}$  (see Lemma 5.5), we get that Growth<sub>n</sub> is a continuous map from  $\mathbb{T}_{loc-K}^{(n)}$  to itself.

We can now construct the process  $(\mathfrak{T}_t, t \geq 0)$  inductively. For  $0 \leq t \leq \xi_1$ , we set  $\mathfrak{T}_t = ([0,t], (0,t))$  and  $N_t = 0$ .

Let  $n \in \mathbb{N}^*$  and assume that  $(\mathfrak{T}_{\xi_n}, \mathbf{v}_n)$  is a tree of height  $\xi_n$  with n leaves, all of them at height  $\xi_n$  and pointed (*i.e.* the vector  $\mathbf{v}_n$  is composed of the root of  $\mathfrak{T}_{\xi_n}$  and all its leaves). Then, we define the process on  $(\xi_n, \xi_{n+1}]$  by setting, for every  $t \in (\xi_n, \xi_{n+1}]$ ,

$$\mathfrak{T}_t = \operatorname{Growth}_n(\mathfrak{T}_{\xi_n}, t - \xi_n) \circledast_{K_n, \xi_n}^{\varepsilon_n} [0, t - \xi_n] \text{ and } N_t = n.$$

Standard properties of Poisson processes give the following result.

**Lemma 6.6.** For every  $n \ge 1$  and every t > 0, conditionally given  $N_t = n - 1$ , the tree  $\mathfrak{T}_t$  is distributed as the tree  $\mathbf{T}_n$  of Section 6.1.1 associated with the density  $f_{dens}$  on [0, t] given by:

(73) 
$$f_{\text{dens}}(u) = \frac{f_{\text{int}}(u)}{F(t)} \mathbf{1}_{[0,t]}(u) \quad with \quad F(t) = \int_0^t f_{\text{dens}}(u) \, \mathrm{d}u.$$

Recall the definition of the function  $\Pi_n^{\circ}$  which removes the pointed vertices. It is easy to see that the process  $(\Pi_{N_t+1}^{\circ}(\mathfrak{T}_t), t \geq 0)$  satisfies the Cauchy property in  $\mathbb{T}_{\text{loc}-K}$  as  $r_s(\Pi_{N_t+1}^{\circ}(\mathfrak{T}_t)) = r_s(\Pi_{N'_t+1}^{\circ}(\mathfrak{T}_{t'}))$  for every  $s \leq t \leq t'$ . Thus this sequence converges a.s. in  $\mathbb{T}_{\text{loc}-K}$ , and we write:

(74) 
$$\mathfrak{T}^{\mathrm{ske}} = \lim_{t \to +\infty} \Pi^{\circ}_{N_t+1}(\mathfrak{T}_t).$$

The tree  $\mathfrak{T}^{\text{ske}}$  is a  $\mathbb{T}_{\text{loc}-\text{K}}$ -valued random variable which has no leaves and thus belongs to  $\mathbb{T}^{\text{no leaf}}_{\text{loc}-\text{K}}$ , and the process  $(\mathfrak{T}^{\text{ske}}_t = r_t(\mathfrak{T}^{\text{ske}}), t \ge 0)$  is distributed as the process  $(\Pi^{\circ}_{N_t+1}(\mathfrak{T}_t), t \ge 0)$ . The tree  $\mathfrak{T}^{\text{ske}}$  will serve as a backbone for the description of the genealogical tree of the conditioned CSBP.

We present now an ancillary result which is a consequence of Lemma 6.1 on two tree-valued processes that have the same one-dimensional marginal.

We first consider the process  $(\mathfrak{T}_t, t \geq 0)$  associated with the intensity  $f_{\text{int}} \equiv 1$ , that is,  $f_{\text{int}}(t) = 1$  for all  $t \geq 0$ . Then we construct a sequence  $\mathbf{t} = (\mathbf{t}_n, n \geq 1)$  of increasing real trees, with  $\mathbf{t}_n \in \mathbb{T}_{\mathrm{K}}^{(n)}$  for every  $n \geq 1$ , all of them of height 1. Let  $(\varepsilon_k, k \geq 1)$  be independent random

variables uniformly distributed on {g, d}. We define the sequence **t** by induction by setting first  $\mathbf{t}_1 = ([0, 1], (0, 1))$ . Let  $n \ge 1$  and assume that  $(\mathbf{t}_n, \mathbf{v}_n)$  is a tree of  $\mathbb{T}_{\mathbf{K}}^{(n)}$  with height 1 and with n leaves all of them at height 1. Conditionally given  $\mathbf{t}_n$ , let  $V_{n+1}$  be a random element on  $\mathbf{t}_n$  uniformly chosen according to the length measure; that is  $V_{n+1}$  is distributed according to the measure  $c_n \mathscr{L}$ , with  $\mathscr{L}$  the length measure on  $\mathbf{t}_n$  and the normalization  $c_n = 1/\mathscr{L}(\mathbf{t}_n)$ . Notice that  $V_{n+1}$  is a.s. not a leaf nor the root of  $\mathbf{t}_n$ . Then we set:

$$\mathbf{t}_{n+1} = \mathbf{t}_n \circledast_{V_{n+1}}^{\varepsilon_{n+1}} \left[ 0, 1 - H(V_{n+1}) \right].$$

In particular, for every measurable nonnegative function G, we have:

(75) 
$$\mathbb{E}\left[G(\mathbf{t}_{n+1})|\mathbf{t}_1,\ldots,\mathbf{t}_n,\varepsilon_{n+1}\right] = \int_{\mathbf{t}_n} \frac{\mathscr{L}(\mathrm{d}x)}{\mathscr{L}(\mathbf{t}_n)} G\left(\mathbf{t}_n \circledast_x^{\varepsilon_{n+1}} \left[0,1-H(x)\right]\right).$$

Recall the measurable function  $N_t$  from Definition (37) which records the number of vertices at level t of a tree without leaves. We have the following result.

**Proposition 6.7.** Let  $n \ge 1$  and  $f_{\text{int}} \equiv 1$ . For all measurable non-negative functional G defined on  $\mathbb{T}_{\text{dis}}^{(n)}$ , we have, with  $\mathscr{L}$  the length measure on  $\mathbf{t}_n$ :

(76) 
$$\mathbb{E}\left[G(\mathfrak{T}_1) \mid N_1 = n - 1\right] = \frac{2^{n-1}}{n!} \mathbb{E}\left[G(\mathbf{t}_n) \prod_{k=1}^{n-1} \mathscr{L}(\mathbf{t}_k)\right],$$

and for all measurable non-negative functional G defined on  $\mathbb{T}_{K}$  (or on  $\mathbb{T}_{loc-K}$ ):

(77) 
$$\mathbb{E}\left[G(\mathfrak{T}_{1}^{\text{ske}}) \mid \tilde{N}_{1}(\mathfrak{T}^{\text{ske}}) = n\right] = \mathbb{E}\left[G \circ \Pi_{n}^{\circ}(\mathfrak{T}_{1}) \mid N_{1} = n - 1\right].$$

*Proof.* By construction, we have that the process  $((\mathfrak{T}_t^{\text{ske}}, \tilde{N}_t(\mathfrak{T}^{\text{ske}})), t \ge 0)$  is distributed as the process  $((\Pi_{N_t+1}^{\circ}(\mathfrak{T}_t), N_t+1), t \ge 0)$ . This gives (77).

We now prove (76) by induction. Thanks to Lemma 6.6, conditionally given  $N_1 = n - 1$ , the tree  $\mathfrak{T}_1$  is distributed as  $\mathbf{T}_n^{\text{unif}}$ . For n = 1, we have  $\mathbf{T}_n^{\text{unif}} = \mathbf{t}_1 = ([0, 1], (0, 1))$  hence Equation (76) holds. Let us suppose that (76) holds for some  $n \ge 1$ . Applying Lemma 6.1, one gets:

$$\mathbb{E}\left[G(\mathbf{T}_{n+1}^{\mathrm{unif}})\right] = \frac{2}{n+1} \mathbb{E}\left[\mathscr{L}(\mathbf{T}_n^{\mathrm{unif}}) \int_{\mathbf{T}_n^{\mathrm{unif}}} \frac{\mathscr{L}^{\mathbf{T}_n^{\mathrm{unif}}}(\mathrm{d}x)}{\mathscr{L}^{\mathbf{T}_n^{\mathrm{unif}}}(\mathbf{T}_n^{\mathrm{unif}})} G\left(\mathbf{T}_n^{\mathrm{unif}} \circledast_x^{\varepsilon} \left[0, 1 - H(x)\right]\right)\right].$$

Now we apply the induction assumption for the right-hand side of the previous equation to get:

$$\mathbb{E}\left[G(\mathbf{T}_{n+1}^{\text{unif}})\right] = \frac{2}{n+1} \frac{2^{n-1}}{n!} \left[ \mathscr{L}^{\mathbf{t}_n}(\mathbf{t}_n) \int_{\mathbf{t}_n} \frac{\mathscr{L}^{\mathbf{t}_n}(\mathrm{d}x)}{\mathscr{L}^{\mathbf{t}_n}(\mathbf{t}_n)} G\left(\mathbf{t}_n \circledast_x^{\varepsilon_{n+1}} \left[0, 1 - H(x)\right]\right) \prod_{k=1}^{n-1} \mathscr{L}^{\mathbf{t}_n}(\mathbf{t}_k) \right]$$
$$= \frac{2^n}{(n+1)!} \mathbb{E}\left[G(\mathbf{t}_{n+1}) \prod_{k=1}^n \mathscr{L}^{\mathbf{t}_{n+1}}(\mathbf{t}_k)\right]$$

by definition of the tree  $\mathbf{t}_{n+1}$  and by (75). This gives that (76) holds with *n* replaced by n + 1. This concludes the proof by induction.

6.2. Brownian CRTs and Kesten trees. Brownian CRTs are random trees in  $\mathbb{T}_{loc-K}$  that code for the genealogy of continuous-state branching processes.

Before recalling the definition of such trees, we give some additional notations. For a locally compact rooted tree  $\mathbf{t}$ , we define the population at level a as the sub-set:

$$\mathcal{Z}_{\mathbf{t}}(a) = \{ u \in \mathbf{t}, \ H(u) = a \}.$$

We denote by  $(\mathbf{t}^{(i),*}, i \in I)$  the connected components of the open set  $\mathbf{t} \setminus r_a(\mathbf{t})$ . For every  $i \in I$ , let  $\varrho_i$  be the MRCA of  $\mathbf{t}^{(i),*}$ , which is equivalently characterized by  $[\![\varrho, \varrho_i]\!] = \bigcap_{u \in \mathbf{t}^{(i),*}} [\![\varrho, u]\!]$ ; notice that  $\varrho_i \in \mathcal{Z}_{\mathbf{t}}(a)$ . We then set  $\mathbf{t}^{(i)} = \mathbf{t}^{(i),*} \cup \{\varrho_i\}$  so that  $\mathbf{t}^{(i)}$  is a locally compact rooted tree with root  $\varrho_i$ , and we consider the point measure on  $\mathcal{Z}_{\mathbf{t}}(a) \times \mathbb{T}_{\text{loc}-\text{K}}$ :

$$\mathcal{N}_a^{\mathbf{t}} = \sum_{i \in I} \delta_{(\varrho_i, \mathbf{t}^{(i)})}$$

We then recall the definition of the excursion measure  $\mathbb{N}^{\theta}$  for  $\beta > 0$   $\theta \ge 0$  associated with a Brownian CRT from [19]. There exists a measure  $\mathbb{N}^{\theta}$  on  $\mathbb{T}_{K}$  (and hence on  $\mathbb{T}_{loc-K}$ ) such that:

- (i) **Existence of a local time.** For every  $a \ge 0$  and for  $\mathbb{N}^{\theta}[d\mathcal{T}]$ -a.e.  $\mathcal{T}$ , there exists a finite measure  $\Lambda_a$  on  $\mathcal{T}$  such that
  - (a)  $\Lambda_0 = 0$  and, for every a > 0,  $\Lambda_a$  is supported on  $\mathcal{Z}_{\mathcal{T}}(a)$ .
  - (b) For every a > 0,  $\mathbb{N}^{\theta}[d\mathcal{T}]$ -a.e., we have  $\{\Lambda_a \neq 0\} = \{H(\mathcal{T}) > a\}$ .
  - (c) For every a > 0,  $\mathbb{N}^{\theta}[d\mathcal{T}]$ -a.e., we have for every continuous function  $\varphi$  on  $\mathcal{T}$ :

$$\begin{split} \langle \Lambda_a, \varphi \rangle &= \lim_{\varepsilon \to 0+} \frac{1}{c_{\varepsilon}^{\theta}} \int \mathcal{N}_a^{\mathcal{T}}(\mathrm{d} u, \mathrm{d} \mathcal{T}') \varphi(u) \mathbf{1}_{\{H(\mathcal{T}') \ge \varepsilon\}} \\ &= \lim_{\varepsilon \to 0+} \frac{1}{c_{\varepsilon}^{\theta}} \int \mathcal{N}_{a-\varepsilon}^{\mathcal{T}}(\mathrm{d} u, \mathrm{d} \mathcal{T}') \varphi(u) \mathbf{1}_{\{H(\mathcal{T}') \ge \varepsilon\}}. \end{split}$$

- (ii) **Branching property.** For every a > 0, the conditional distribution of the point measure  $\mathcal{N}_a^{\mathcal{T}}(\mathrm{d} u, \mathrm{d} \mathcal{T}')$ , under the probability measure  $\mathbb{N}^{\theta}[\mathrm{d} \mathcal{T} \mid H(\mathcal{T}) > a]$  and given  $r_a(\mathcal{T})$ , is that of a Poisson point measure on  $\mathcal{Z}_{\mathcal{T}}(a) \times \mathbb{T}_{\mathrm{loc}-\mathrm{K}}$  with intensity  $\Lambda_a(\mathrm{d} u)\mathbb{N}^{\theta}[\mathrm{d} \mathcal{T}']$ .
- (iii) **Regularity of the local time process.** We can choose a modification of the process  $(\Lambda_a, a \ge 0)$  in such a way that the mapping  $a \mapsto \Lambda_a$  is  $\mathbb{N}^{\theta}[d\mathcal{T}]$ -a.e. continuous for the weak topology of finite measures on  $\mathcal{T}$ .
- (iv) Link with CSBP. Under  $\mathbb{N}^{\theta}[d\mathcal{T}]$ , the process  $(\langle \Lambda_a^{\mathcal{T}}, 1 \rangle, a \geq 0)$  is distributed as a CSBP under its canonical measure with branching mechanism:

$$\psi(\lambda) = \beta \lambda^2 + 2\beta \theta \lambda, \quad \lambda \ge 0.$$

We now extend the definition of the measure  $\mathbb{N}^{\theta}$  (only on  $\mathbb{T}_{loc-K}$ ) for  $\theta < 0$  by a Girsanov transformation, following [7]. For  $t \geq 0$ , set  $\mathcal{G}_t = \sigma(r_t(\mathcal{T}))$  and  $Z_t = \Lambda_t(\mathcal{T})$ , the latter notation is consistent with Section 2.2. The CSBP process  $Z = (Z_t, t \geq 0)$  is Markov with respect to the filtration  $(\mathcal{G}_t, t \geq 0)$ . For  $\theta < 0$  and t > 0, we set:

(78) 
$$\mathbb{N}^{-\theta}[\mathrm{d}\mathcal{T}]_{|\mathcal{G}_t} = \mathrm{e}^{2\theta Z_t} \,\mathbb{N}^{\theta}[\mathrm{d}\mathcal{T}]_{|\mathcal{G}_t}.$$

Then properties (i) to (iv) still hold for every  $\theta \in \mathbb{R}$ . This Girsanov transformation is consistent with the Girsanov transformation of CSBP given by (14). Let us stress that the measure  $\mathbb{N}^{\theta}$  on  $\mathbb{T}_{\text{loc}-K}$  depends also on the parameter  $\beta > 0$ .

The so-called Kesten tree with parameters  $(\beta, \theta) \in \mathbb{R}^*_+ \times \mathbb{R}_+$  can be defined as the genealogical tree associated with the continuous-state branching process with the same parameters, conditioned on non-extinction (see for instance [28]). This latter process can also be defined by

adding to the initial process a particular immigration. We use this second approach to extend the definition of the Kesten tree for  $\theta < 0$ .

Using our framework, the Kesten tree with parameters  $(\beta, \theta) \in \mathbb{R}^*_+ \times \mathbb{R}$  is the random tree  $\mathcal{T}^* = \text{Tree}(\mathcal{M})$  (see Section 5.10 for the definition of the map Tree), with  $\mathcal{M}(dh, dT)$  a Poisson point measure on  $E = \mathbb{R}_+ \times \mathbb{T}^*_{\text{loc}-K}$  with intensity  $2\beta \mathbf{1}_{\{h>0\}} dh \mathbb{N}^{\theta}[dT]$ . The Kesten tree belongs a.s. to the space  $\mathbb{T}^{\text{spine},0}_{\text{loc}-K}$ .

6.3. Backbone decomposition. The decomposition of a (sub)critical Brownian CRT  $\mathcal{T}$  according to a spine  $[\![\emptyset, x]\!]$ , where  $x \in \mathcal{T}$  is a leaf picked at random at level t > 0, that is according to the local time  $\Lambda_t(dx)$ , is given in Theorem 4.5 in [19]. In our setting, it can be rephrased in the next theorem. Recall that, for t > 0, the (discrete) 1-pointed tree  $[0, t] \in \mathbb{T}_{loc-K}^{(1)}$  denotes the segment [0, t] endowed with the Euclidean distance, with the root 0 and the distinguished vertex t. Recall that  $\mathbb{T}_{loc-K}^{\text{spine},0}$  defined in Section 5.6 is the set of locally compact rooted trees with one infinite marked branch such that the root is not a branching vertex. From Remark 5.29, we recall that informally the 1-pointed tree  $(T, (\varrho, v_1)) = \text{Graft}_1([0, t], T^*)$ , with  $T^* \in \mathbb{T}_{loc-K}^{\text{spine},0}$  is the tree where the marked branch has been cut at height t (the part above t being removed), and the cut marked branch is identified with [0, t], so that the root  $\varrho$  of T is identified with 0 and the distinguished vertex  $v_1$  of T is also identified with t.

**Theorem 6.8** ([19]). Let  $\beta > 0$ ,  $\theta \ge 0$  and t > 0. Let  $\mathcal{T}^*$  be under  $\mathbb{E}$  a Kesten tree with parameter  $(\beta, \theta)$ . For every non-negative measurable functional F on  $\mathbb{T}^{(1)}_{loc-K}$  (or  $\mathbb{T}^{(1)}_{K}$ ), we have, with  $\varrho$  the root of  $\mathcal{T}$ :

(79) 
$$\mathbb{N}^{\theta}\left[\int_{\mathcal{T}} \Lambda_t(\mathrm{d}v) F(\mathcal{T},(\varrho,v))\right] = \mathrm{e}^{-2\beta\theta t} \mathbb{E}\left[F\left(\mathrm{Graft}_1([0,t],\mathcal{T}^*)\right)\right].$$

We extend this result to the super-critical case  $\theta < 0$ .

**Corollary 6.9.** Let  $\beta > 0$ ,  $\theta \in \mathbb{R}$  and t > 0. Let  $\mathcal{T}^*$  be under  $\mathbb{E}$  a Kesten tree with parameter  $(\beta, \theta)$ . For every non-negative measurable functional F on  $\mathbb{T}^{(1)}_{\text{loc}-K}$ , Equation (79) holds.

*Proof.* We first to prove (79) for functionals F of the form:

(80) 
$$F(T, \mathbf{v}) = e^{-\langle \Phi, \mathcal{M}(T, \mathbf{v}) \rangle},$$

where  $(T, \mathbf{v}) \in \mathbb{T}^{(1)}_{\text{loc}-K}$  and  $\Phi$  is a continuous non-negative function with bounded support defined on  $\mathbb{R}_+ \times \mathbb{T}^*_{\text{loc}-K}$  (with  $\mathbb{T}^*_{\text{loc}-K} = \mathbb{T}_{\text{loc}-K} \setminus \{T_0\}$  where  $T_0 \in \mathbb{T}_{\text{loc}-K}$  is the tree reduced to its root, see Section 5.10). And the measure  $\mathcal{M}(T, \mathbf{v})$  on  $\mathbb{R}_+ \times \mathbb{T}^*_{\text{loc}-K}$  is defined at the end of Section 5.10.

For simplicity, we simply write  $(\mathcal{T}, v)$  for the 1-pointed tree  $(\mathcal{T}, (\varrho, v))$ . Let  $\theta > 0$ . Using (78), we have for every s > t that:

$$\mathbb{N}^{-\theta} \left[ \int_{\mathcal{T}} \Lambda_t(\mathrm{d}v) \, \mathrm{e}^{\langle \Phi, \, \mathcal{M}r_s(\mathcal{T}, v) \rangle \rangle} \right] = \mathbb{N}^{\theta} \left[ \int_{\mathcal{T}} \Lambda_t(\mathrm{d}v) \, \mathrm{e}^{2\theta Z_s - \langle \Phi, \, \mathcal{M}(r_s(\mathcal{T}, v)) \rangle} \right].$$

We apply then (79) to get:

$$\begin{split} \mathbb{N}^{-\theta} \left[ \int_{\mathcal{T}} \Lambda_t(\mathrm{d}v) \, \mathrm{e}^{-\langle \Phi, \mathcal{M}(r_s(\mathcal{T}, v)) \rangle} \right] \\ &= \mathrm{e}^{-2\beta\theta t} \, \mathbb{E} \left[ \mathrm{e}^{2\theta Z_s} \, F(r_s(\mathrm{Graft}_1([0, t], \mathcal{T}^*))) \right] \\ &= \exp \left\{ -2\beta\theta t - 2\beta \int_0^t \mathrm{d}a \, \mathbb{N}^{\theta} \left[ 1 - \mathrm{e}^{-\Phi(a, r_{s-a}(\mathcal{T})) + 2\theta Z_{s-a}} \right] \right\} \\ &= \exp \left\{ -2\beta\theta t - 2\beta \int_0^t \mathrm{d}a \, \left( \mathbb{N}^{-\theta} \left[ 1 - \mathrm{e}^{-\Phi(a, r_{s-a}(\mathcal{T}))} \right] + \mathbb{N}^{\theta} \left[ 1 - \mathrm{e}^{2\theta Z_{s-a}} \right] \right) \right\} \\ &= \exp \left\{ 2\beta\theta t - 2\beta \int_0^t \mathrm{d}a \, \mathbb{N}^{-\theta} \left[ 1 - \mathrm{e}^{-\Phi(a, r_{s-a}(\mathcal{T}))} \right] \right\}, \end{split}$$

where we used the definition of the Kesten tree for the second equality, (78) again for the third one, and that  $\mathbb{N}^{\theta} \left[ 1 - e^{2\theta Z_a} \right] = u(-2\theta, a) = -2\theta$ , see (4) and (7), for the last one. As  $\Phi$  has bounded support, we get taking *s* large enough:

$$\mathbb{N}^{-\theta} \left[ \int_{\mathcal{T}} \Lambda_t(\mathrm{d}v) \,\mathrm{e}^{-\langle \Phi, \,\mathcal{M}(\mathcal{T}, v) \rangle} \right] = \exp\left\{ 2\beta\theta t - 2\beta \int_0^t \mathrm{d}a \,\mathbb{N}^{-\theta} \left[ 1 - \mathrm{e}^{-\Phi(a, \mathcal{T})} \right] \right\}.$$

Then the result follows from the definition of the Kesten tree, that is (79) holds for F given by (80).

Recall  $\mathbb{T}_{loc-K}^{(n),0,*}$  defined in (63) is the Borel subset of  $\mathbb{T}_{loc-K}^{(1)}$  of the trees such that the root is not a branching vertex and the pointed vertex is distinct from the root. The map:

$$(T, \mathbf{v}) \mapsto (d(\varrho, v), \mathcal{M}(T, \mathbf{v}))$$

with  $\mathbf{v} = (\varrho, v)$ , defined on  $\mathbb{T}^{(n),0,*}_{loc-K}$  is injective and bi-measurable, see Corollary 5.33. Furthermore the set  $\mathbb{T}^{(n),0,*}_{loc-K}$  is of full measure with respect to the distribution of  $(\mathcal{T}, \mathbf{v})$  under  $\mathbb{N}^{\theta}[d\mathcal{T}] \Lambda_t(dv)$ , with  $\mathbf{v} = (\varrho, v)$ , as  $\mathbb{N}^{\theta}$ -a.e. the root of  $\mathcal{T}$  is not a branching vertex and  $d(\varrho, v) = t > 0$ . Thus, as t > 0 is fixed, we get that  $(\mathcal{T}, v)$  is a measurable function of  $\mathcal{M}(\mathcal{T}, v)$ . We then conclude by the monotone class theorem that Equation (79) holds for any non-negative measurable function F defined on  $\mathbb{T}^{(n)}_{loc-K}$ .

Let  $\beta > 0$ ,  $\theta \in \mathbb{R}$  and t > 0. Recall  $\tilde{c}_t^{\theta} = (2\theta)/(1 - e^{-2\beta\theta t})$  defined in (3). We consider the probability measure on [0, t]:

(81) 
$$\nu(ds) = \frac{2\beta\theta \,\mathrm{e}^{2\beta\theta s}}{\mathrm{e}^{2\beta\theta t} - 1} \,\mathbf{1}_{[0,t]}(s) \,\mathrm{d}s = \beta \,\tilde{c}_t(\theta) \,\mathrm{e}^{-2\beta\theta(t-s)} \,\mathbf{1}_{[0,t]}(s) \,\mathrm{d}s.$$

Let  $(\mathbf{T}_n, \mathbf{v}_n)$  be, under  $\mathbb{P}^{\theta, t}$ , the tree of  $\mathbb{T}_{\text{dis}}^{(n)}$  defined in Section 6.1.1 associated with the measure  $\nu$  and t > 0 (recall that all the distinguished vertices from  $\mathbf{v}_n$  but the root are at distance t from the root). The following theorem is a generalization of Theorem 6.8 when picking n leaves uniformly at random at level t.

**Theorem 6.10.** Let  $\beta > 0$ ,  $\theta \in \mathbb{R}$ , t > 0 and  $n \in \mathbb{N}^*$ . For every non-negative measurable function F defined on  $\mathbb{T}^{(n)}_{\text{loc}-\text{K}}$ , we have:

(82) 
$$\mathbb{N}^{\theta}\left[\int_{\mathcal{T}^n} \Lambda_t^{\otimes n}(\mathrm{d}\mathbf{v}^*) F(\mathcal{T}, \mathbf{v})\right] = n! \left(\tilde{c}_t^{\theta}\right)^{1-n} \mathrm{e}^{-2\beta\theta t} \mathbb{E}^{\theta, t}\left[\left[F\left(\mathrm{Graft}_n\left((\mathbf{T}_n, \mathbf{v}_n), \mathcal{T}^*\right)\right)\right],$$

where  $\mathbf{v} = (\varrho, \mathbf{v}^*) \in \mathcal{T}^{n+1}$ , with  $\varrho$  the root of  $\mathcal{T}$ , and  $\mathcal{T}^* = (\mathcal{T}^*_A, A \in \mathcal{P}^+_n)$  is under  $\mathbb{P}^{\theta, t}$  a family of independent Kesten trees with parameter  $(\beta, \theta)$ , independent of  $\mathbf{T}_n$ .

*Proof.* We prove Formula (82) by induction. For n = 1, as  $\mathbf{T}_1 = [0, t]$  (with root  $\rho = 0$  and distinguished vertex  $v_1 = t$ ), this is Corollary 6.9.

Let  $k \in \mathbb{N}^*$ . Recall the maps  $\mathbf{L}_k$ , from (52) in Section 5.7, and  $\operatorname{Split}_k$  from (55) in Section 5.8. For  $(T, \mathbf{v}) \in \mathbb{T}_{\operatorname{loc}-K}^{(k)}$  and  $A \in \mathcal{P}_k^+$ , we write  $\mathcal{M}_A[T, \mathbf{v}](\mathrm{d}h, \mathrm{d}\mathbf{t})$  for the measure  $\mathcal{M}(\hat{T}_A(T, \mathbf{v}))$  on  $E = \mathbb{R}_+ \times \mathbb{T}_{\operatorname{loc}-K}^*$ , where  $(\hat{T}_A(T, \mathbf{v}), A \in \mathcal{P}_k) = \operatorname{Split}_k(T, \mathbf{v})$  and the measure  $\mathcal{M}(T, \mathbf{v})$  is defined at the end of Section 5.10. We also recall the notation  $(\ell_A(T, \mathbf{v}), A \in \mathcal{P}_k^+) = \mathbf{L}_k(T, \mathbf{v})$ , and notice that  $\ell_A(T, \mathbf{v}) = 0$  implies that  $\mathcal{M}_A[T, \mathbf{v}] = 0$ . Let  $n \in \mathbb{N}^*$  and  $(\Phi_A, A \in \mathcal{P}_n^+)$  be a family of non-negative measurable functions defined on E. Let f be a bounded non-negative measurable functions defined on  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  (or more simply on  $\mathbb{T}_{\operatorname{dis}}^{(n)}$ ). We shall first prove (82) for a non-negative function F defined on  $\mathbb{T}_{\operatorname{loc}-K}^{(n)}$  of the form:

$$F(T, \mathbf{v}) = f(\operatorname{Span}(T, \mathbf{v})) \exp\left\{-\sum_{A \in \mathcal{P}_n^+} \langle \Phi_A, \mathcal{M}_A[T, \mathbf{v}] \rangle\right\}.$$

Let  $n \geq 2$  and suppose that (82) holds for n-1. For  $k \in \{1, \ldots, n\}$ , we denote by  $\mathcal{T}^{[k]} =$ Span $(\mathcal{T}, \mathbf{v}_k) \in \mathbb{T}^{(k)}_{\text{loc}-\text{K}}$ , where  $\mathbf{v}_k = (v_0 = \varrho, \mathbf{v}_k^*)$  and  $\mathbf{v}_k^* = (v_1, \ldots, v_k)$ ; and we simply write  $\mathcal{M}_A^{[k]}$  for  $\mathcal{M}_A[\mathcal{T}, \mathbf{v}_k]$  and  $\ell_A^{[k]}$  for  $\ell_A(\mathcal{T}, \mathbf{v}_k)$ , so that under  $\mathbb{N}^{\theta}[\mathrm{d}\mathcal{T}] \ell_t^{\otimes n}(\mathrm{d}\mathbf{v}^*)$ :

$$F(\mathcal{T}, \mathbf{v}_n) = f(\mathcal{T}^{[n]}) \exp\left\{-\sum_{A \in \mathcal{P}_n^+} \langle \Phi_A, \mathcal{M}_A^{[n]} \rangle\right\}.$$

We also write  $v_A^{[k]}$  and  $w_A^{[k]}$  for  $v_A$  and  $w_A$  from (50) and (51) with  $(T, \mathbf{v})$  replaced by  $(\mathcal{T}^{[k]}, \mathbf{v}_k)$ ; and thus we have  $\ell_A^{[k]} = d(w_A^{[k]}, v_A^{[k]})$ .

Similarly, under  $\mathbb{E}^{\theta,t}$ , for  $k \geq 2$ , we write also  $\hat{\mathcal{M}}_{A}^{[k]}$  for the measure  $\mathcal{M}(T_{A}^{*})$  restricted to  $[0, \ell_{A}(\mathbf{T}_{k})] \times \mathbb{T}_{\text{loc}-K}^{*}, \hat{v}_{A}^{[k]}$  and  $\hat{w}_{A}^{[k]}$  for  $v_{A}$  and  $w_{A}$  from (50) and (51) with  $(T, \mathbf{v})$  replaced by  $(\mathbf{T}_{k}, \mathbf{v}_{k})$ , and  $\hat{\ell}_{A}^{[k]} = d(\hat{w}_{A}^{[k]}, \hat{v}_{A}^{[k]}) = \ell_{A}(\mathbf{T}_{k})$ . For  $n \geq 2$ , simply writing  $\mathbf{T}_{n}$  for  $(\mathbf{T}_{n}, \mathbf{v}_{n})$ , we have:

$$F\left(\operatorname{Graft}_{n}(\mathbf{T}_{n},\mathcal{T}^{*})\right) = f(\mathbf{T}_{n}) \exp\left\{-\sum_{A \in \mathcal{P}_{n}^{+}} \left\langle \Phi_{A}, \hat{\mathcal{M}}_{A}^{[n]} \right\rangle\right\}.$$

Using the definition of Kesten tree via Poisson point measures and the definition of the function  $\operatorname{Graft}_n$ , we obtain in particular that:

(83) 
$$\mathbb{E}^{\theta,t} \Big[ F \big( \operatorname{Graft}_n(\mathbf{T}_n, \mathcal{T}^*) \big) \Big] = \mathbb{E}^{\theta,t} \left[ F'(\mathbf{T}_n) \right],$$

where

(84) 
$$F'(\mathbf{T}_n) = f(\mathbf{T}_n) \exp\left\{-2\beta \sum_{A \in \mathcal{P}_n^+} \int_0^{\hat{\ell}_A^{[n]}} \mathrm{d}a \,\mathbb{N}^\theta \left[1 - \mathrm{e}^{-\Phi_A(a,\mathcal{T})}\right]\right\}.$$

Recall (27). Set  $p_n = p_{\mathbf{v}_{n-1}}(v_n)$  for the projection of  $v_n$  on  $\mathcal{T}^{[n-1]}$ . Since  $\mathbb{N}^{\theta}$ -a.e.  $p_n \neq \varrho$ , we deduce that there exists  $\mathbb{N}^{\theta}$ -a.e. a unique (random)  $B \in \mathcal{P}_{n-1}^+$  such that  $p_n \in ]\!]w_B^{[n-1]}, v_B^{[n-1]}]\!] \subset \mathcal{T}^{[n-1]}$ , and write  $h_n = d(p_n, w_B^{[n-1]})$ . Recall the function Tree, defined in Section 5.10 just before Lemma 5.31, from  $\mathbb{M}(E)$  into  $\mathbb{T}_{loc-K}^{[2]}$  and the projection  $\tilde{\Pi}$  from  $\mathbb{T}_{loc-K}^{[2]}$  to  $\mathbb{T}_{loc-K}$ , defined just

before Lemma 5.23, which forgets about the marked sub-tree defined in Section 5.6. We simply write Tree' =  $\Pi \circ$  Tree. On the one hand, we have:

(85) 
$$\mathcal{T}^{[n]} = \mathcal{T}^{[n-1]} \circledast_{\min B, H(p_n)} [0, t - H(p_n)], \\ \ell_B^{[n-1]} = \ell_B^{[n]} + \ell_{B \cup \{n\}}^{[n]}, \\ \mathcal{M}_B^{[n-1]} = \mathcal{M}_{B \cup \{n\}}^{[n]} + \mathcal{M}_B^{[n]}(\cdot + h_n, \cdot) + \delta_{\left(h_n, \operatorname{Tree'}\left(\mathcal{M}_{\{n\}}^{[n]}\right)\right)}$$

and, to fix notation, we shall write:

$$\mathcal{M}_B^{[n-1]} = \mathcal{M}_B[\mathcal{T}, \mathbf{v}_{n-1}] = \sum_{i \in I_{n-1}^B} \delta_{\mathbf{h}_i^{[n-1], B}, \mathcal{T}_i^{[n-1], B}}.$$

;

On the other hand, for  $A \in \mathcal{P}_{n-1}^+$  and  $A \neq B$ , we have:

$$\begin{array}{ll} (86) \qquad B \subset A \implies \mathcal{M}_{A}^{[n-1]} = \mathcal{M}_{A\cup\{n\}}^{[n]}, \quad \mathcal{M}_{A}^{[n]} = 0, \quad \ell_{A}^{[n-1]} = \ell_{A\cup\{n\}}^{[n]} \quad \text{and} \quad \ell_{A}^{[n]} = 0, \\ (87) \qquad A \cap B \in \{\emptyset, A\} \implies \mathcal{M}_{A}^{[n-1]} = \mathcal{M}_{A}^{[n]}, \quad \mathcal{M}_{A\cup\{n\}}^{[n]} = 0, \quad \ell_{A}^{[n-1]} = \ell_{A}^{[n]} \quad \text{and} \quad \ell_{A\cup\{n\}}^{[n]} = 0, \\ (88) \quad A \cap B \notin \{\emptyset, B, A\} \implies \mathcal{M}_{A}^{[n-1]} = \mathcal{M}_{A}^{[n]} = \mathcal{M}_{A\cup\{n\}}^{[n]} = 0 \quad \text{and} \quad \ell_{A}^{[n-1]} = \ell_{A}^{[n]} = \ell_{A\cup\{n\}}^{[n]} = 0. \\ \text{It is also easy to rebuild} (\mathcal{M}_{A}^{[n]}, A \in \mathcal{P}_{n}^{+}) \text{ from} (\mathcal{M}_{A}^{[n-1]}, A \in \mathcal{P}_{n-1}^{+}) \text{ and } v_{n}. \end{array}$$

Set

$$F_n = \mathbb{N}^{\theta} \left[ \int_{\mathcal{T}^n} \Lambda_t^{\otimes n}(\mathrm{d}\mathbf{v}_n^*) F(\mathcal{T}, \mathbf{v}_n) \right].$$

Considering that  $\mathcal{T}_i^{[n-1],B}$  is a subset of  $\mathcal{T}$ , we have:

$$\begin{split} F_{n} &= \mathbb{N}^{\theta} \Bigg[ \int_{\mathcal{T}^{n-1}} \Lambda_{t}^{\otimes (n-1)} (\mathrm{d}\mathbf{v}_{n-1}^{*}) \sum_{B \in \mathcal{P}_{n-1}^{+}} \sum_{i \in I_{n-1}^{B}} \int_{\mathcal{T}_{i}^{[n-1],B}} \ell_{t}(dv_{n}) F(\mathcal{T}, \mathbf{v}_{n}) \Bigg] \\ &= \mathbb{N}^{\theta} \Bigg[ \int_{\mathcal{T}^{n-1}} \Lambda_{t}^{\otimes (n-1)} (\mathrm{d}\mathbf{v}_{n-1}^{*}) \\ &\sum_{B \in \mathcal{P}_{n-1}^{+}} \sum_{i \in I_{n-1}^{B}} \Gamma_{B} \left( \mathcal{T}^{[n-1]}, H(w_{B}^{[n-1]}), \mathcal{M}_{B,i}^{[n-1]}, H(w_{B}^{[n-1]}) + \mathbf{h}_{i}^{[n-1],B}, \mathcal{T}_{i}^{[n-1],B} \right) \\ &\times \exp \Bigg\{ - \sum_{A \in \mathcal{P}_{n-1}^{+} \setminus \{B\}} \Bigl\langle \mathbf{1}_{\{B \subset A\}} \, \Phi_{A \cup \{n\}} + \mathbf{1}_{\{A \cap B = \emptyset \text{ or } A\}} \, \Phi_{A}, \mathcal{M}_{A}^{[n-1]} \Bigr\rangle \Bigg\} \Bigg], \end{split}$$

where the measure  $\mathcal{M}_{B,i}^{[n-1]}$  is the measure  $\mathcal{M}_{B}^{[n-1]}$  but for the atom at  $(\mathbf{h}_{i}^{[n-1],B}, \mathcal{T}_{i}^{[n-1],B})$ :

$$\mathcal{M}_{B,i}^{[n-1]} = \mathcal{M}_{B}^{[n-1]} - \delta_{(\mathbf{h}_{i}^{[n-1],B},\mathcal{T}_{i}^{[n-1],B})},$$

and, for  $(T, \mathbf{w}) \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n-1)}, (T', \varrho') \in \mathbb{T}_{\mathrm{loc}-\mathrm{K}}, \nu \in \mathbb{M}(E)$  and  $\mathrm{h}' \geq h \geq 0$ :  $\Gamma_B((T, \mathbf{w}), h, \nu, \mathrm{h}', T') = f(T \circledast_{\min B, \mathrm{h}'} [0, t - \mathrm{h}']) \exp\left\{-\langle \Phi_{B, \mathrm{h}'-h}, \nu \rangle\right\}$   $\times \int_{T'} \Lambda_{t-\mathrm{h}'}(dv) \exp\left\{-\langle \Phi_{\{n\}}, \mathcal{M}(T', (\varrho', v))\rangle\right\},$  with:

(89) 
$$\Phi_{B,h''}(s,\mathbf{t}) = \mathbf{1}_{\{s \le h''\}} \Phi_{B \cup \{n\}}(s,\mathbf{t}) + \mathbf{1}_{\{s > h''\}} \Phi_B(s-h'',\mathbf{t})$$

For 
$$B \in \mathcal{P}_{n-1}^+$$
, using the notation  $\hat{\mathcal{M}}_B^{[n]} = \sum_{i \in \hat{I}_{n-1}^B} \delta_{(\hat{\mathbf{h}}_i^{[n-1],B}, \hat{\mathcal{T}}_i^{[n-1],B})}$ , we set for  $i \in \hat{I}_{n-1}^B$ :  
 $\hat{\mathcal{M}}_{B,i}^{[n-1]} = \hat{\mathcal{M}}_B^{[n-1]} - \delta_{(\hat{\mathbf{h}}_i^{[n-1],B}, \hat{\mathcal{T}}_i^{[n-1],B})}$ .

We deduce from the induction assumption (*i.e.* Equation (82) with n-1 instead of n) and the definition of Kesten tree, with  $F_n = (n-1)! (\tilde{c}_t^{\theta})^{2-n} e^{-2\beta\theta t} G_n$  that:

$$G_{n} = \mathbb{E}^{\theta} \Biggl[ \sum_{B \in \mathcal{P}_{n-1}^{+}} \sum_{i \in \hat{I}_{n-1}^{B}} \Gamma_{B} \left( \mathbf{T}_{n-1}, H(\hat{w}_{[n-1],B}), \hat{\mathcal{M}}_{B,i}^{[n-1]}, H(\hat{w}_{[n-1],B}) + \hat{\mathbf{h}}_{i}^{[n-1],B}, \hat{\mathcal{T}}_{i}^{[n-1],B} \right) \\ \times \exp \Biggl\{ - \sum_{A \in \mathcal{P}_{n-1}^{+} \setminus \{B\}} \Bigl\langle \mathbf{1}_{\{B \subset A\}} \, \Phi_{A \cup \{n\}} + \mathbf{1}_{\{A \cap B = \emptyset \text{ or } A\}} \, \Phi_{A}, \hat{\mathcal{M}}_{A}^{[n-1]} \Bigr\rangle \Biggr\} \Biggr].$$

Since for  $A \in \mathcal{P}_{n-1}^+$ , the random measure  $\mathcal{M}(T_A^*, \hat{\ell}_A^{[n-1]})(\mathrm{d}h', \mathrm{d}\mathcal{T}')$  is conditionally given  $\hat{\ell}_A^{[n-1]}$  a Poisson point measure on  $[0, \hat{\ell}_A^{[n-1]}] \times \mathbb{T}_{\mathrm{loc}-\mathrm{K}}$  with intensity  $2\beta \mathrm{d}h' \mathbb{N}^{\theta}[\mathrm{d}\mathcal{T}']$ , we deduce from Palm formula that:

$$\begin{split} G_{n} &= \mathbb{E}^{\theta} \Biggl[ \sum_{B \in \mathcal{P}_{n-1}^{+}} 2\beta \int_{0}^{\hat{\ell}_{B}^{[n-1]}} \mathrm{d}r \int \mathbb{N}^{\theta} [\mathrm{d}\mathcal{T}] \Gamma_{B} \left( \mathbf{T}_{n-1}, H(\hat{w}_{B}^{[n-1]}), \hat{\mathcal{M}}_{B}^{[n-1]}, H(\hat{w}_{B}^{[n-1]}) + r, \mathcal{T} \right) \\ & \times \exp \Biggl\{ - \sum_{A \in \mathcal{P}_{n-1}^{+} \setminus \{B_{x}\}} \left\langle \mathbf{1}_{\{B_{x} \subset A\}} \Phi_{A \cup \{n\}} + \mathbf{1}_{\{A \cap B_{x} = \emptyset \text{ or } A\}} \Phi_{A}, \hat{\mathcal{M}}_{A}^{[n-1]} \right\rangle \Biggr\} \Biggr] \\ &= \mathbb{E}^{\theta} \Biggl[ 2\beta \int_{\mathbf{T}_{n-1,t}} \mathscr{L}(\mathrm{d}x) \int \mathbb{N}^{\theta} [\mathrm{d}\mathcal{T}] \Gamma_{B_{x}} \left( \mathbf{T}_{n-1}, H(\hat{w}_{B_{x}}^{[n-1]}), \hat{\mathcal{M}}_{B_{x}}^{[n-1]}, H(x), \mathcal{T} \right) \\ & \times \exp \Biggl\{ - \sum_{A \in \mathcal{P}_{n-1}^{+} \setminus \{B_{x}\}} \left\langle \mathbf{1}_{\{B_{x} \subset A\}} \Phi_{A \cup \{n\}} + \mathbf{1}_{\{A \cap B_{x} = \emptyset \text{ or } A\}} \Phi_{A}, \hat{\mathcal{M}}_{A}^{[n-1]} \right\rangle \Biggr\} \Biggr], \end{split}$$

where  $B_x$  is the only element B of  $\mathcal{P}_{n-1}^+$  such that x belongs to the branch B of  $\mathbf{T}_{n-1}$ :  $x \in [\!] \hat{w}_B^{[n-1]}, \hat{v}_B^{[n-1]}]\!]$ , where, as  $\mathbf{T}_{n-1}$  is discrete, we recall that  $\operatorname{Split}_{n-1}(\mathbf{T}_{n-1}) = ([\![ \hat{w}_A^{[n-1]}, \hat{v}_A^{[n-1]}]\!], A \in \mathcal{P}_{n-1})$  with  $\mathcal{P}_{n-1} = \mathcal{P}_{n-1}^+ \cup \{\{0\}\}$ . Using (82) again for n = 1 (or Corollary 6.9) gives:

$$\int \mathbb{N}^{\theta}[\mathrm{d}\mathcal{T}] \Gamma_{B}(\mathbf{T}_{n-1,t},h,\nu,\mathbf{h}',\mathcal{T}) = f\left(\mathbf{T}_{n-1} \circledast_{\min B,\mathbf{h}'} [0,t-\mathbf{h}']\right) \mathrm{e}^{-\langle \Phi_{\mathbf{h}'-h},\nu\rangle} \\ \times \exp\left\{-2\beta\theta(t-\mathbf{h}') - 2\beta \int_{0}^{t-\mathbf{h}'} \mathrm{d}a \,\mathbb{N}^{\theta} \left[1 - \mathrm{e}^{-\Phi_{\{n\}}(a,\mathcal{T})}\right]\right\}.$$

With x chosen according to the length measure  $\mathscr{L}(dx)$  on  $\mathbf{T}_{n-1}$ , the tree  $\mathbf{T}_{n-1} \circledast_{\min B_x, H(x)} [0, t - H(x)]$  is obtained by grafting a branch of length t - H(x) at x on  $\mathbf{T}_{n-1}$  and thus will simply be denoted as  $\mathbf{T}_{n-1} \circledast_x [0, t - H(x)]$  (see also Remark 6.2 for similar notation). Therefore, we

obtain:

We deduce from Lemma 6.1 with the density:

$$f_{\text{dens}}(s) = \frac{2\beta\theta \,\mathrm{e}^{2\beta\theta s}}{\mathrm{e}^{2\beta\theta t} - 1} \,\mathbf{1}_{[0,t]}(s) = \tilde{c}_t^\theta \,\beta \,\mathrm{e}^{-2\beta\theta(t-s)} \,\mathbf{1}_{[0,t]}(s)$$

that for a non-negative measurable function F'' defined on  $\mathbb{T}^{(n)}_{\text{loc}-\text{K}}$  (or  $\mathbb{T}^{(n)}_{\text{dis}}$ ):

$$\mathbb{E}^{\theta,t}\left[2\beta \int_{\mathbf{T}_{n-1}} \mathscr{L}(\mathrm{d}x) F''\left(\mathbf{T}_{n-1} \circledast_x \left[0, t - H(x)\right]\right) \,\mathrm{e}^{-2\beta\theta(t-H(x))}\right] = (\tilde{c}_t^\theta)^{-1} \, n \, \mathbb{E}^{\theta,t}\left[F''(\mathbf{T}_n)\right].$$

Using similar equations as (85), (86), (87) and (88) stated with  $\mathbf{T}_n$  instead of  $(\mathcal{T}, \mathbf{v}_n)$  as well as an obvious choice of F'', we obtain that:

$$G_n = (\tilde{c}_t^{\theta})^{-1} n \mathbb{E}^{\theta, t} \left[ F'(\mathbf{T}_n) \right]$$

where  $F'(\mathbf{T}_n)$  is given by (84). Then, we deduce from (83) that:

$$G_n = (\tilde{c}_t^{\theta})^{-1} n \mathbb{E}^{\theta, t} \Big[ F \big( \operatorname{Graft}_n(\mathbf{T}_n, \mathcal{T}^*) \big) \Big]$$

This gives:

$$\mathbb{N}^{\theta} \left[ \int_{\mathcal{T}^n} \Lambda_t^{\otimes n} (\mathrm{d} \mathbf{v}_n^*) F(\mathcal{T}, \mathbf{v}) \right] = F_n = (n-1)! (\tilde{c}_t^{\theta})^{2-n} \mathrm{e}^{-2\beta\theta t} G_n$$
$$= n! \left( \tilde{c}_t^{\theta} \right)^{1-n} \mathrm{e}^{-2\beta\theta t} \mathbb{E}^{\theta, t} \Big[ F \big( \mathrm{Graft}_n(\mathbf{T}_n, \mathcal{T}^*) \big) \Big].$$

Thus, Equation (82) holds for the functionals F we considered.

Recall that  $\mathbb{T}_{loc-K}^{(n),0,*}$  is the Borel subset of  $\mathbb{T}_{loc-K}^{(n)}$  of the trees such that the root is not a branching vertex and the point vertices (but the root) are distinct from the root. The map:

$$(T, \mathbf{v}) \mapsto \left( \operatorname{Span}(T, \mathbf{v}), \left( \mathcal{M}_A[T, \mathbf{v}], A \in \mathcal{P}_n^+ \right) \right)$$

50

defined on  $\mathbb{T}_{\text{loc}-K}^{(n),0,*}$  is one-to-one onto its image and bi-measurable, see Corollary 5.34. Furthermore the set  $\mathbb{T}_{\text{loc}-K}^{(n),0,*}$  is of full measure with respect to the distribution of  $(\mathcal{T},\mathbf{v})$  under  $\mathbb{N}^{\theta}[\mathrm{d}\mathcal{T}] \Lambda_t^{\otimes n}(d\mathbf{v}^*)$ , with  $\mathbf{v} = (\varrho, \mathbf{v}^*)$ , as  $\mathbb{N}^{\theta}$ -a.e. the root of  $\mathcal{T}$  is not a branching vertex. Thus,  $(\mathcal{T},\mathbf{v})$  is a measurable function of  $(\mathcal{T}^{[n]}, (\hat{\mathcal{M}}_A^{[n]}, A \in \mathcal{P}_n^+))$ . We then conclude by the monotone class theorem that Equation (82) holds for any non-negative measurable function F defined on  $\mathbb{T}_{\text{loc}-K}^{(n)}$ .

6.4. Main result. Let  $\beta > 0$ ,  $\theta, \alpha \in \mathbb{R}_+$  and let  $S^{\alpha,\theta}$  be a Poisson point measure on  $[0,\infty)$  with intensity measure  $f_{\text{int}}(t) \, dt$ , where:

(90) 
$$f_{\rm int}(t) = \alpha \beta \, {\rm e}^{2\beta \theta t}, \quad t \ge 0.$$

We first consider the case  $\alpha > 0$ . Denote by  $(\xi_i, i \in \mathbb{N}^*)$  the increasing sequence of jumping times of the inhomogeneous Poisson process  $(N_t^{\alpha,\theta} = S^{\alpha,\theta}([0,t]), t \ge 0)$ . We consider the  $\mathbb{T}_{dis}^{(n)}$ -valued random variable  $\mathfrak{T}_{\xi_n}$  of Section 6.1.3 for  $n \ge 1$  associated to  $f_{int}$ . In particular, recall that, for every  $n \ge 1$ ,  $\mathfrak{T}_{\xi_n}$  is a discrete tree with n pointed leaves, where all of them are at height  $\xi_n$ .

For every  $n \geq 1$ , let  $\mathcal{T}^{n,*} = (\mathcal{T}_A, A \in \mathcal{P}_n^+)$  be a family of independent Kesten trees with parameter  $(\beta, \alpha)$ , independent of the tree  $\mathfrak{T}_{\xi_n}$ . We defined the random marked tree:

$$\mathcal{T}^{(n)} = \left( \Pi_n^{\circ}(\tilde{\mathcal{T}}^{(n)}), \operatorname{Span}^{\circ}(\tilde{\mathcal{T}}^{(n)}) \right) \quad \text{with} \quad \tilde{\mathcal{T}}^{(n)} = \operatorname{Graft}_n\left(\mathfrak{T}_{\xi_n}, \mathcal{T}^{n,*}\right)$$

Thanks to Lemma 5.17 and Lemma 5.28 on the measurability of the grafting function, we deduce that  $\mathcal{T}^{(n)}$  is a  $\mathbb{T}^{[2]}_{\mathrm{loc}-\mathrm{K}}$ -valued random variable. The family of the distributions of the  $\mathbb{T}^{[2]}_{\mathrm{loc}-\mathrm{K}}$ -valued random trees  $(\mathcal{T}^{(n)}, n \geq 1)$  is consistent in the sense that, for every  $n \geq 1$  and every  $t \leq \xi_n$ ,  $r^{[2]}_t(\mathcal{T}^{(n)}) \stackrel{(d)}{=} r^{[2]}_t(\mathcal{T}^{(n+1)})$ . It is in particular a Cauchy sequence in  $\mathbb{T}^{[2]}_{\mathrm{loc}-\mathrm{K}}$ , and we denote by  $(\mathcal{T}^{\alpha,\theta},\mathfrak{T}^{\alpha,\theta})$  its limit which is thus a  $\mathbb{T}^{[2]}_{\mathrm{loc}-\mathrm{K}}$ -valued random variable. By construction,  $\mathfrak{T}^{\alpha,\theta}$  and  $\mathfrak{T}^{\mathrm{ske}}$  have the same distribution. This construction is a formal way to define the tree obtained by grafting on the infinite discrete tree  $\mathfrak{T}^{\mathrm{ske}}$  (which serves as a backbone) at  $x_i$  a tree  $\mathcal{T}_i$  where  $((x_i, \mathcal{T}_i), i \in I)$  are the atoms of a Poisson point measure of intensity  $\mathscr{L}(\mathrm{d}x)\mathbb{N}^{\theta}(\mathrm{d}\mathcal{T})$ , where  $\mathscr{L}$  is the length measure on  $\mathfrak{T}^{\mathrm{ske}}$ .

For  $\alpha = 0$ , we simply define  $(\mathcal{T}^{\emptyset,\theta}, \mathfrak{T}^{0,\theta})$  as the Kesten tree with parameter  $(\beta, \alpha)$ .

We then define the 
$$\mathbb{T}_{loc-K}^{[2]}$$
-valued random process  $\left((\mathcal{T}_t^{\alpha,\theta},\mathfrak{T}_t^{\alpha,\theta}), t \ge 0\right)$  by setting:  
 $\mathcal{T}_t^{\alpha,\theta} = r_t(\mathcal{T}^{\alpha,\theta}) \quad \text{and} \quad \mathfrak{T}_t^{\alpha,\theta} = r_t(\mathfrak{T}_t^{\alpha,\theta}),$ 

that is  $(\mathcal{T}_t^{\alpha,\theta},\mathfrak{T}_t^{\alpha,\theta}) = r_t^{[2]}((\mathcal{T}^{\alpha,\theta},\mathfrak{T}^{\alpha,\theta}))$ . Recall the  $\mathbb{T}_{\text{loc}-K}$ -valued random variable  $\mathcal{T}$  is under  $\mathbb{N}^{\theta}$  a Lévy tree; and we write  $\mathcal{T}_t = r_t(\mathcal{T})$ . We now give the main result of this section.

**Proposition 6.11.** Let  $\beta \in \mathbb{R}^*_+$ ,  $\theta, \alpha \in \mathbb{R}_+$  and t > 0. For every non-negative measurable functional F on  $\mathbb{T}_{loc-K}$  (or  $\mathbb{T}_K$ ), we have:

$$\mathbb{E}\left[F\left(\mathcal{T}_{t}^{\alpha,\theta}\right)\right] = \mathbb{N}^{\theta}\left[F\left(\mathcal{T}_{t}\right) M_{t}^{\alpha,\theta}\right]$$

*Proof.* We first consider the case  $\alpha > 0$ . Let us fix t > 0, and write  $N_t$  for  $N_t^{\alpha,\theta}$ . Recall that  $\tilde{N}_t(\mathfrak{T}^{\alpha,\theta})$ , the number of vertices of the tree  $\mathfrak{T}^{\alpha,\theta}$  at level t is distributed as  $N_t + 1$ . Using Lemma 5.12 on the measurability of  $\tilde{N}_t$ , we get that  $F\left(\mathcal{T}_t^{\alpha,\theta}\right) \mathbf{1}_{\{\tilde{N}_t(\mathfrak{T}^{\alpha,\theta})=n\}}$  is a well defined non-negative random variable.

Let  $(\mathbf{T}_n, n \ge 0)$  be the sequence of trees defined in Section 6.1.1 associated with the function:

(91) 
$$f_{\text{dens}}(s) = \beta \, \tilde{c}_t(\theta) \, \mathrm{e}^{-2\beta\theta(t-s)} \, \mathbf{1}_{[0,t]}(s)$$

We have, with  $\operatorname{Graft}_k^\circ = \Pi_k^\circ \circ \operatorname{Graft}_k$ :

$$\mathbb{E}\left[F\left(\mathcal{T}_{t}^{\alpha,\theta}\right)\right] = \sum_{n\in\mathbb{N}} \mathbb{E}\left[F\left(\mathcal{T}_{t}^{\alpha,\theta}\right) \mid \tilde{N}_{t}(\mathfrak{T}^{\alpha,\theta}) = n+1\right] \mathbb{P}(N_{t}^{\alpha,\theta} = n)$$
$$= \sum_{n\in\mathbb{N}} \mathbb{E}\left[F\left(r_{t}\left(\operatorname{Graft}_{n+1}^{\circ}(\mathfrak{T}_{t},\mathcal{T}^{n+1,*})\right)\right) \mid N_{t}^{\alpha,\theta} = n\right] \frac{(\alpha/c_{t}^{\theta})^{n} \operatorname{e}^{-\alpha/c_{t}^{\theta}}}{n!}$$
$$= \sum_{n\in\mathbb{N}} \mathbb{E}\left[F\left(r_{t}\left(\operatorname{Graft}_{n+1}^{\circ}(\mathfrak{T}_{n+1},\mathcal{T}^{n+1,*})\right)\right)\right] \frac{(\alpha/c_{t}^{\theta})^{n} \operatorname{e}^{-\alpha/c_{t}^{\theta}}}{n!},$$

where we used that  $\tilde{N}_t(\mathfrak{T}^{\alpha,\theta})$  is distributed as  $N_t^{\alpha,\theta} + 1$  for the first equality, that conditionally on  $\tilde{N}_t(\mathfrak{T}^{\alpha,\theta}) = n + 1$ , the random variable  $\mathcal{T}_t^{\alpha,\theta}$  is distributed as  $r_t \left( \text{Graft}_{n+1}^{\circ}(\mathfrak{T}_t, \mathcal{T}^{n+1,*}) \right)$  conditionally on  $N_t^{\alpha,\theta} = n$  and that  $N_t^{\alpha,\theta}$  is distributed as a Poisson process with intensity  $\alpha$  at time  $1/c_t^{\theta}$  (see Lemma 4.2) for the second one, that  $\mathfrak{T}_t$  conditionally on  $N_t^{\alpha,\theta} = n$  is distributed as  $\mathbf{T}_{n+1}$  with  $f_{\text{int}}$  and  $f_{\text{dens}}$  in (73) given by (90) and (91) (see Lemma 6.6) for the last one. Recall that  $r_t$  and  $\Pi_{n+1}^{\circ}$  is measurable. Using Theorem 6.10 and that  $\nu(ds)$  in (81) is exactly  $f_{\text{dens}}(s) ds$ with  $f_{\text{dens}}$  given by (91), we have:

$$\mathbb{E}\left[F\left(r_t\left(\operatorname{Graft}_{n+1}^{\circ}(\mathbf{T}_{n+1},\mathcal{T}^{n+1,*})\right)\right)\right]$$

$$=\frac{\left(\tilde{c}_t^{\theta}\right)^n e^{2\beta\theta t}}{(n+1)!} \mathbb{N}^{\theta}\left[\int_{\mathcal{T}^{n+1}} \Lambda_t^{\otimes(n+1)}(\mathrm{d}\mathbf{v}^*) F(r_t \circ \Pi_{n+1}^{\circ}(\mathcal{T},\mathbf{v}))\right]$$

$$=\frac{\left(\tilde{c}_t^{\theta}\right)^n e^{2\beta\theta t}}{(n+1)!} \mathbb{N}^{\theta}\left[\int_{\mathcal{T}^{n+1}} \Lambda_t^{\otimes(n+1)}(\mathrm{d}\mathbf{v}^*) F(\mathcal{T}_t)\right]$$

$$=\frac{\left(\tilde{c}_t^{\theta}\right)^n e^{2\beta\theta t}}{(n+1)!} \mathbb{N}^{\theta}\left[Z_t^{n+1}F(\mathcal{T}_t)\right],$$

as  $Z_t = \Lambda_t(\mathbf{1})$  is the total local time of  $\mathcal{T}$  at level t. Thus, using the definition of  $M_t^{\alpha,\theta}$  in (12), we obtain:

$$\mathbb{E}\left[F\left(\mathcal{T}_{t}^{\alpha,\theta}\right)\right] = \sum_{n \in \mathbb{N}} \frac{\left(\tilde{c}_{t}^{\theta}\right)^{n} e^{2\beta\theta t}}{(n+1)!} \mathbb{N}^{\theta}\left[Z_{t}^{n+1}F(\mathcal{T}_{t})\right] \frac{(\alpha/c_{t}^{\theta})^{n} e^{-\alpha/c_{t}^{\theta}}}{n!} = \mathbb{N}^{\theta}\left[F\left(\mathcal{T}_{t}\right) M_{t}^{\alpha,\theta}\right]$$

The simpler case  $\alpha = 0$ , which is left to the reader, can also be handled in a similar way.  $\Box$ 

As a conclusion, we deduce the following result for  $\alpha > 0$ .

**Theorem 6.12.** Let  $\alpha, \beta > 0, \theta \in \mathbb{R}$ . Assume that the function a is such that as  $t \to \infty$ :

$$a_t \sim \begin{cases} \alpha \beta^2 t^2, & \text{if } \theta = 0; \\ \alpha (2\theta)^{-2} e^{2\beta |\theta| t}, & \text{if } \theta \neq 0. \end{cases}$$

For every non-negative measurable function F on  $\mathbb{T}_K$  and s > 0, we have:

$$\lim_{t \to \infty} \mathbb{N}^{\theta} \left[ F\left(\mathcal{T}_{s}\right) \mid Z_{t} = a_{t} \right] = \mathbb{E} \left[ F\left(\mathcal{T}_{s}^{\alpha, |\theta|}\right) \right].$$

*Proof.* Clearly, Proposition 3.5 still holds if  $H_s$  is  $\mathcal{G}_s = \sigma(r_s(\mathcal{T}))$  measurable, that is  $H_s = F(\mathcal{T}_s)$  with F non-negative defined on  $\mathbb{T}_{loc-K}$ , and  $Z_t$  is the total local time of  $\mathcal{T}$  at level t, see Section 6.2. We deduce that:

$$\lim_{t \to \infty} \mathbb{N}^{\theta}[F(\mathcal{T}_s) | Z_t = a_t] = \mathbb{N}^{|\theta|} \left[ F(\mathcal{T}_s) M_s^{\alpha, |\theta|} \right] = \mathbb{E} \left[ F(\mathcal{T}_s^{\alpha, |\theta|}) \right],$$

where we used Proposition 6.11 for the last equality.

Similarly, we also get the following result for  $\alpha = 0$ . Recall that  $\mathcal{T}^{0,\theta}$  is a Kesten tree with parameter  $(\beta, \theta)$ .

**Theorem 6.13.** Let  $\beta > 0$ ,  $\theta \in \mathbb{R}$ . Assume that the function *a* is positive such that as  $t \to \infty$ :

$$a_t = \begin{cases} o(t^2), & \text{if } \theta = 0; \\ o(e^{2\beta|\theta|t}) & \text{if } \theta \neq 0. \end{cases}$$

For every non-negative measurable function F on  $\mathbb{T}_{K}$  and s > 0, we have:

$$\lim_{t \to \infty} \mathbb{N}^{\theta} \left[ F\left(\mathcal{T}_{s}\right) \mid Z_{t} = a_{t} \right] = \mathbb{E} \left[ F\left(\mathcal{T}_{s}^{0, |\theta|}\right) \right].$$

### Index of notation

	Spanning and truncation			
Trees and pointed trees $T$ to $T$ , reparis potations for these (or	- Span <sup>°</sup> $(T, \mathbf{v})$ : the discrete rooted sub-tree of $T$ spanned by the pointed vertices $\mathbf{v}$ .			
class of equiv. trees).	<ul> <li>Span(T, v): the rooted tree (Span°(T, v), v) with the pointed vertices v.</li> <li>The map Π<sub>n</sub>° removes the pointed vertices (but the root) from an n-pointed tree: Π<sub>n</sub>°(T, v) = (T, ρ). Thus:</li> </ul>			
d: generic distance on a tree.				
$\varrho{:}$ generic notation for the root of trees.				
$H(x) = d(\varrho, x)$ : height of the vertex $x$ .				
H(T): height of the tree $T$ .	$\Pi_n^{\circ}(\operatorname{Span}(T, \mathbf{v})) = \operatorname{Span}^{\circ}(T, \mathbf{v}).$			
$T_x$ : subtree of T above the vertex $x \in T$ .	- $r_t(T, \mathbf{v})$ : the tree T truncated at level t with the spanned tree Span <sup>°</sup> (T, $\mathbf{v}$ ), and the pointed vertices $\mathbf{v}$			
$\llbracket x, y \rrbracket$ : the branch joining the vertices $x$ to $y$ .				
$T_0$ : the rooted tree reduced to its root.	$= r^{[2]} r^{[2],+} r^{[2],-} r^{[2]} \tilde{r}^{[2],+} $ various trunca-			
$T_1$ : the rooted infinite branch.	tion on marked trees (see Sect. 5.5 and 5.6).			
$\mathscr{L}$ or $\mathscr{L}^T$ : length measure on the tree $T$ .				
$\mathbf{v} = (v_0 = \varrho, v_1, \dots, v_n)$ : generic notation for pointed vertices of a tree.	Splitting and grafting - $\mathbf{L}_n(T, \mathbf{v})$ record the lengths of all the branches			
$(T, \mathbf{v})$ a (or a class of equiv. of) rooted <i>n</i> -pointed tree.	of the subtree $\text{Span}(T, \mathbf{v})$ spanned by the <i>n</i> pointed vertices:			
$(T, S) = (T, S, d, \rho)$ a (or a class of equiv. of)	$\mathbf{L}_n(T, \mathbf{v}) = (\ell_A(T, \mathbf{v}), A \in \mathcal{P}_n^+),$			
marked tree with $\rho \in S \subset T$ .	with $\mathcal{P}_n^+$ the set of all subsets $A \subset \{1, \ldots, n\}$ such that $A \neq \emptyset$ .			
Grafting a tree on a tree	- $\operatorname{Split}_n(T, \mathbf{v})$ record the subtrees of $T$ associated to all the branches of $\operatorname{Span}(T, bv)$ :			
$(T \otimes_i T, \mathbf{v} \otimes \mathbf{v})$ , also denoted by $T \otimes_i T'$ , is the tree obtained by grafting $T'$ on $T$ at the	(92) $\operatorname{Split}_{n}(T, \mathbf{v}) = \left(\hat{T}_{A}(T, \mathbf{v}), A \in \mathcal{P}_{n}\right)$			
pointed vertex $v_i \in T$ and identifying the root	$= \frac{1}{2} \left( \left( 0 \right) \right)$			

 $\rho'$  of T' with  $v_i$ . The pointed vertices  $\mathbf{v} \circledast \mathbf{v}'$ 

are the concatenation of the pointed vertices

 $\mathbf{v}$  of T and the pointed vertices  $\mathbf{v}'$  (but for the

-  $T \circledast_{i,h} T'$ , is the tree obtained by grafting T'

-  $T \circledast_{i,h}^{\epsilon} T'$ , with  $\epsilon \in \{g, d\}$ , same as above but

for the pointed vertices of T' which are in-

serted on the left (if  $\epsilon = g$ ) or on the right of

on T at level h on the branch  $\llbracket \varrho, v_i \rrbracket$ .

root) of T'.

 $v_i$  (if  $\epsilon = d$ ).

with  $\mathcal{P}_n = \mathcal{P}_n^+ \cup \{\{0\}\}.$ 

- Graft<sub>n</sub>( $T', (T_A^*, A \in \mathcal{P}_n^+)$ ): replace the branches labeled by A, of the discrete *n*pointed tree T' by the trees  $T_A^*$  with a marked infinite branch cut at the length  $\ell_A(T, \mathbf{v})$ . (The discrete tree ( $T', \mathbf{v}'$ ) can be coded/replaced by  $\mathbf{L}_n(T', \mathbf{v}')$ .)
- Intuitively, we have for  $(T, \mathbf{v})$  a *n*-pointed tree whose root is not a branching vertex (see (60)):

$$(T, \mathbf{v}) = \operatorname{Graft}_n \left( \operatorname{Span}_n(T, \mathbf{v}), \operatorname{Split}_n(T, \mathbf{v}) \right).$$

Set of (equiv. classes of) trees $\mathbb{T}_{V}$ set of (equiv. classes of) rooted compact	Trees with a marked branch and point measures
trees.	- $\mathbb{M}(E)$ set of point measures on $E = \mathbb{R}_+ \times$
- $\mathbb{T}_{\mathrm{K}}^{(n)}$ set of (equiv. classes of) rooted <i>n</i> -	$\mathbb{T}^*_{\text{loc}-K}$ which are bounded on bounded sets
pointed compact trees; $\mathbb{T}_{\mathrm{K}}^{(0)} = \mathbb{T}_{\mathrm{K}}.$	OI $E$ .
- $d_{\text{GH}}^{(n)}$ the distance on $\mathbb{T}_{\text{K}}^{(n)}$ ; $d_{\text{GH}}^{(0)} \equiv d_{\text{GH}}$ .	- Tree : $\mathbb{M}(E) \to \mathbb{I}_{loc-K}$ maps the measure $\mathcal{M} = \sum_{i \in I} \delta_{h_i, T_i}$ to the marked tree $(T, T_1)$ ,
- $\mathbb{T}_{loc-K}$ set of (equiv. classes of) rooted loc. compact trees.	with the rooted tree $T$ obtained by grafting the trees $T_i$ on the rooted infinite branch $T_1$
- $\mathbb{T}^*_{\text{loc}-K} = \mathbb{T}_{\text{loc}-K} \setminus \{T_0\}.$	at level $h_i$ .
- $\mathbb{T}^0_{loc-K}$ subset of $\mathbb{T}_{loc-K}$ of trees whose root is not a branching vertex.	- $\mathcal{M}$ : $\mathbb{T}_{loc-K}^{spine} \to \mathbb{M}(E)$ maps the marked tree $(T, T_1)$ to the measure $\sum_{i \in I} \delta_{h_i, T_i}$ where $T_i \setminus \{\rho_i\}$ are the connected component of $T \setminus T_1$
- $\mathbb{T}^{0,*}_{\mathrm{loc}-\mathrm{K}} = \mathbb{T}^0_{\mathrm{loc}-\mathrm{K}} \cap \mathbb{T}^*_{\mathrm{loc}-\mathrm{K}}.$	with root $\varrho_i \in T_1$ and $h_i = d(\varrho, \varrho_i)$ , where $\varrho_i$ is the common root of $T$ and $T_1$ .
- $\mathbb{T}_{loc-K}^{(n)}$ set of (equiv. classes of) rooted <i>n</i> - pointed loc. compact trees; $\mathbb{T}_{loc-K}^{(0)} = \mathbb{T}_{loc-K}$ .	- $\mathcal{M}$ is also defined on $\mathbb{T}^{(1)}_{loc-K}$ .
- $d_{\text{LGH}}^{(n)}$ the distance on $\mathbb{T}_{\text{loc}-\text{K}}^{(n)}$ ; $d_{\text{LGH}}^{(0)} \equiv d_{\text{LGH}}$ .	Reconstruction results
- $\mathbb{T}_{loc-K}^{(n),0}$ subset of $\mathbb{T}_{loc-K}^{(n)}$ of trees whose root is not a branching vertex.	- With Id the identity map: $Tree \circ \mathcal{M} = Id  on \ \mathbb{T}^{spine}_{loc-K},$
- $\mathbb{T}_{loc}^{(n),*}$ subset of $\mathbb{T}_{loc}^{(n)}$ of trees whose all	$\mathcal{M} \circ \text{Tree} = \text{Id} \text{ on } \tilde{\mathbb{M}}(E) = \text{Im}(\mathcal{M}).$
pointed vertices (but the root) are distinct from the root.	- $(T, \mathbf{v}) \in \mathbb{T}^{(1),0,*}_{\text{loc}-\text{K}}$ can be recovered in a measurable way from $(d(o, v), \mathcal{M}(T, \mathbf{v}))$ .
- $\mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n),0,*} = \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n),0} \cap \mathbb{T}_{\mathrm{loc}-\mathrm{K}}^{(n),*}.$	$- (T \mathbf{y}) \in \mathbb{T}^{(n),0,*} \text{ can be recov-}$
- $\mathbb{T}_{\text{dis}}^{(n)}$ subset of $\mathbb{T}_{K}^{(n)} \subset \mathbb{T}_{\text{loc}-K}^{(n)}$ of discrete trees.	$(\mathbf{r}, \mathbf{v})$ $(r$
- $\mathbb{T}^{[2]}_{loc-K}$ set of (equiv. classes of) rooted loc. compact marked trees.	$(\operatorname{Span}_{n}(T, \mathbf{v}), (\mathcal{M}_{A}[T, \mathbf{v}], A \in \mathcal{P}_{n}^{+})), \text{ where } \mathcal{M}_{A}[T, \mathbf{v}] = \mathcal{M}(\hat{T}_{A}(T, \mathbf{v})), \text{ with } \hat{T}_{A}(T, \mathbf{v}) \in \mathbb{T}^{(1)}$
- $\mathbb{T}_{\text{loc}-K}^{\text{spine}}$ subset of $\mathbb{T}_{\text{loc}-K}^{[2]}$ of marked trees $(T, S)$ such that $S = T_1$ , with $T_1$ the infinite branch.	$\lim_{b \to -K}$ defined by the splitting operation in (92).

#### References

≻.....

- R. Abraham, A. Bouaziz, and J.-F. Delmas. Very fat geometric Galton-Watson trees. ESAIM Probab. Stat., 24:294–314, 2020.
- [2] R. Abraham and J.-F. Delmas. Williams' decomposition of the Lévy continuum random tree and simultaneous extinction probability for populations with neutral mutations. *Stochastic Processes and their Applications*, 119(4):1124–1143, 2009.
- [3] R. Abraham and J.-F. Delmas. A continuum-tree-valued Markov process. Ann. Probab., 40(3):1167–1211, 2012.

- [4] R. Abraham and J.-F. Delmas. Local limits of conditioned Galton-Watson trees: the condensation case. Electron. J. Probab., 19:Article 56, 1–29, 2014.
- [5] R. Abraham and J.-F. Delmas. Asymptotic properties of expansive Galton-Watson trees. *Electron. J. Probab.*, 24(15):1–51, 2019.
- [6] R. Abraham, J.-F. Delmas, and P. Hoscheit. A note on the Gromov-Hausdorff-Prokhorov distance between (locally) compact metric measure spaces. *Electronic Journal of Probability*, 18, 2013.
- [7] R. Abraham, J.-F. Delmas, and P. Hoscheit. Exit times for an increasing Lévy tree-valued process. Probab. Theory Related Fields, 159(1-2):357–403, 2014.
- [8] R. Abraham, J.-F. Delmas, and M. Nassif. Global regime for general additive functionals of conditioned Bienaymé-Galton-Watson trees. Probability Theory and Related Fields, pages 1–75, 2021.
- [9] D. Aldous. The continuum random tree III. Annals of Probability, 21:248–289, 1993.
- [10] J. Berestycki, A. E. Kyprianou, and A. Murillo-Salas. The prolific backbone for supercritical superprocesses. Stochastic processes and their applications, 121(6):1315–1331, 2011.
- [11] V. I. Bogachev. Measure theory. Vol. I, II. Springer-Verlag, Berlin, 2007.
- [12] Z.-Q. Chen, Y.-X. Ren, and T. Yang. Skeleton decomposition and law of large numbers for supercritical superprocesses. Acta Applicandae Mathematicae, 159(1):225–285, 2019.
- [13] D. J. Daley and D. Vere-Jones. An introduction to the theory of point processes. Vol. I. Probability and its Applications (New York). Springer-Verlag, New York, second edition, 2003. Elementary theory and methods.
- [14] D. J. Daley and D. Vere-Jones. An introduction to the theory of point processes. Vol. II. Probability and its Applications (New York). Springer, New York, second edition, 2008. General theory and structure.
- [15] D. A. Dawson and Z. Li. Stochastic equations, flows and measure-valued processes. The Annals of Probability, 40(2):813–857, 2012.
- [16] J.-F. Delmas. Height process for super-critical continuous state branching process. Markov Process. Related Fields, 14(2):309–326, 2008.
- [17] T. Duquesne. Continuum random trees and branching processes with immigration. Stochastic Process. Appl., 119(1):99–129, 2009.
- [18] T. Duquesne and J.-F. Le Gall. Random trees, Lévy processes and spatial branching processes. Astérisque, (281):vi+147, 2002.
- [19] T. Duquesne and J.-F. Le Gall. Probabilistic and fractal aspects of Lévy trees. Probab. Theory Related Fields, 131(4):553–603, 2005.
- [20] T. Duquesne and M. Winkel. Growth of Lévy trees. Probab. Theory Related Fields, 139(3-4):313–371, 2007.
- [21] S. N. Evans. Probability and real trees, volume 1920 of Lecture Notes in Mathematics. Springer, Berlin, 2008. Lectures from the 35th Summer School on Probability Theory held in Saint-Flour, July 6–23, 2005.
- [22] S. N. Evans, J. Pitman, and A. Winter. Rayleigh processes, real trees, and root growth with re-grafting. Probab. Theory Related Fields, 134(1):81–126, 2006.
- [23] D. Fekete, J. Fontbona, and A. E. Kyprianou. Skeletal stochastic differential equations for continuous-state branching processes. *Journal of Applied Probability*, 56(4):1122–1150, 2019.
- [24] D. Fekete, J. Fontbona, and A. E. Kyprianou. Skeletal stochastic differential equations for superprocesses. Journal of Applied Probability, 57(4):1111–1134, 2020.
- [25] S. Janson. Simply generated trees, conditioned Galton-Watson trees, random allocations and condensation. Probab. Surv., 9:103–252, 2012.
- [26] T. Jonsson and S. Stefánsson. Condensation in nongeneric trees. J. Stat. Phys., 142:277–313, 2011.
- [27] A. E. Kyprianou and Y.-X. Ren. Backbone decomposition for continuous-state branching processes with immigration. Statistics & probability letters, 82(1):139–144, 2012.
- [28] A. Lambert. Quasi-stationary distributions and the continuous-state branching process conditioned to be never extinct. *Electron. J. Probab.*, 12:no. 14, 420–446, 2007.
- [29] Z.-H. Li. Asymptotic behaviour of continuous time and state branching processes. J. Austral. Math. Soc. Ser. A, 68(1):68–84, 2000.
- [30] G. Miermont. Tessellations of random maps of arbitrary genus. Annales scientifiques de l'École Normale Supérieure, Ser. 4, 42(5):725–781, 2009.
- [31] L. Overbeck. Martin boundaries of some branching processes. Ann. Inst. H. Poincaré Probab. Statist., 30(2):181–195, 1994.
- [32] R. Purves. Bimeasurable functions. Fund. Math., 58:149–157, 1966.
- [33] L. C. G. Rogers and J. W. Pitman. Markov functions. Ann. Probab., 9(4):573–582, 1981.

Romain Abraham, Institut Denis Poisson, Université d'Orléans, Université de Tours, CNRS, France

Email address: romain.abraham@univ-orleans.fr

JEAN-FRANÇOIS DELMAS, CERMICS, ECOLE DES PONTS, FRANCE *Email address*: delmas@cermics.enpc.fr

HUI HE, LABORATORY OF MATHEMATICS AND COMPLEX SYSTEMS, SCHOOL OF MATHEMATICAL SCIENCES, BEIJING NORMAL UNIVERSITY, P.R. CHINA

Email address: hehui@bnu.edu.cn